

# **Economics**

**Discussion Paper** 

Series

EDP-2402

# **Evolutionary Finance: Models with Short-Lived Assets**

Zerong Chen

October 2024

Economics

School of Social Sciences The University of Manchester Manchester M13 9PL

## Evolutionary Finance: Models with Short-Lived Assets

#### Zerong Chen\*

Abstract. Evolutionary Finance explores the "survival and extinction" questions of investment strategies (portfolio rules) in the market selection process. It studies the stochastic dynamics of financial markets, where asset prices are determined endogenously by short-run equilibrium between supply and demand, which is formed each period as a result of the interaction of strategies employed by competing market participants. This paper focuses on "short-lived" risky securities that are traded at the beginning of each period and yield payoffs at the end of it (which live only one period), with the cycle then repeating. We review some key models developed in this area, which address the following problems in order: 1) introducing the central results that we are primarily interested in under substantially more general assumptions; 2) exploring the Nash equilibrium properties of survival strategies and the single survivor problem within the framework of independent and identically distributed states of the world and fixed-mix portfolio rules; 3) extending the discussion on the single survivor problem to a considerably broader scope, emphasizing its Markovian nature; 4) including a risk-free asset into the market; 5) allowing for short-selling in the market. The two main results of these studies are: i) the existence of survival strategies that can be expressed by explicit formulas, i.e., Kelly's rule of "betting one's beliefs"; and, ii) the asymptotic uniqueness (within a specific class of strategies called basic) of such survival strategies.

**Key words:** Evolutionary finance, Survival portfolio rules, Short-lived assets, Stochastic dynamic games, Evolutionary game theory.

JEL Classification: C73, D53, G11, C73, D58.

\*Economics Department, University of Manchester, Oxford Road, Manchester, M13 9PL, UK. E-mail: zerong.chen@postgrad.manchester.ac.uk.

I am grateful to Igor Evstigneev, who introduced me into the fascinating area of Evolutionary Finance and suggested to write this paper, which grew out of my MSc Economics thesis.

## Chapter 1

# Introduction

"...the perception of potential threats to survival may be much more important in determining behavior than the perceptions of potential profits, so that profit maximization is not really the driving force. It is fear of loss rather than hope of gain that limits our behavior."

> Kenneth E. Boulding Evolutionary Economics, 1981, p.108.

Conventional models of equilibrium and dynamics of asset markets are based on the classical Walrasian general equilibrium theory<sup>1</sup>, which assumes that investors behave fully rationally and maximize utilities of consumption subject to budget constraints (i.e., a well-defined and precisely stated constrained optimization problem). In contrast, the present work in Evolutionary Finance  $(EF)^2$ develops an alternative equilibrium concept – behavioral equilibrium – admitting that market participants may be boundedly rational and have a whole variety of patterns of behavior determined by their individual psychology. Strategies may involve, for instance, mimicking, satisficing, rules of thumb (based on experience), etc.; and, may be interactive, depending on the behavior of the others. This approach, inspired by the insights of behavioral economics and finance<sup>3</sup>, overcomes several drawbacks of traditional theory, particularly by eliminating the need for the "perfect foresight" assumption to establish equilibrium and the reliance on knowledge of unobservable individual utilities and beliefs. In this sense, it opens new possibilities for the modern financial markets modeling, especially on the global level, where the main objectives are: domination in a market segment, fastest capital growth, or simply survival (especially in crisis environments).

We examine game-theoretic models of financial markets with endogenous asset prices determined by short-run equilibrium of supply and demand. Equi-

<sup>&</sup>lt;sup>1</sup>For a textbook treatment of this subject, see Magill and Quinzii [111].

 $<sup>^{2}</sup>$  For surveys describing the state of the art in this filed, see Evstigneev et al. [61, 64]; for an elementary textbook treatment of this subject, see Evstigneev et al. [63], Chapter 20.

<sup>&</sup>lt;sup>3</sup>See, e.g., Tversky and Kahneman [158], Shleifer [149], Shiller [148], and Thaler [154].

librium is formed consecutively in each time period by the interaction of general, adaptive portfolio rules adopted by competing investors, depending on the exogenous random factors and observed history of the game. Uncertainty in asset payoffs at each period is modeled via an exogenous discrete-time stochastic process that governs the evolution of the states of the world, which aims to capture various macroeconomic and business cycle variables that may affect investors' behavior. There are two fundamental types of models in EF: with short-lived (one-period) assets and long-lived dividend-paying assets, see, e.g., Amir et al. [4] and [3], respectively. This paper focuses on models of the former type. Short-lived assets refer to standardized contracts for various real assets (e.g., energy or natural resources, agricultural products, real estate, etc.) that are traded (at the market prices) at the beginning of each time period [t, t + 1]and yield payoffs at the end of it, contingent on random events that might occur by date t + 1.

The primary aim of the study is to identify investment strategies that guarantee "long-run survival", i.e., keeping a strictly positive, bounded away from zero, share of market wealth over an infinite time horizon, irrespective of what strategies used by others. The main result demonstrates that Kelly's [95] well-known portfolio rule of "betting one's beliefs" guarantees the property of unconditional survival<sup>4</sup>. Moreover, the strategy satisfying this property is essentially unique: any other strategy of this kind (within a specific class of strategies called *basic*) converges asymptotically to the Kelly rule. This result on asymptotic uniqueness, can be regarded as an analogue of turnpike theorems (see Remark 2.4).

The models we examine in this paper, through a game-theoretic lens, bridge two fundamental paradigms: stochastic dynamic games (strategic frameworks, as pioneered by Shapley [145])<sup>5</sup> and evolutionary game theory (solution concepts).

The modeling framework postulates a dynamic non-cooperative market game, in which the mechanisms of short-term price formation and market clearing follow those of one-shot strategic market games (see Shapley and Shubik [146], Amir et al. [7], Erickson and Pakes [57]). On the conceptual level, this approach relates to the games of survival<sup>6</sup> first explored by Milnor and Shapley [120] as a constant-sum stochastic game, which can be treated as a natural game-theoretic analogue of the famous gambler's ruin decision problem (Dubins and Savage [53]). A zero-sum matrix game is played by two players at each stage over an infinite time horizon, the outcome of which determines their wealth dynamics (as a state variable), and the final outcome shows either bankruptcy of one player or a draw. Likewise, Shubik and Whitt [151] analyzed a N-player

 $<sup>^4\,{\</sup>rm The}$  state of the art in this area related to the Kelly investment criterion is surveyed by MacLean et al. [109].

 $<sup>^{5}</sup>$  For more recent developments, see Neyman and Sorin [125]. In view of this paper's focus on long-run survival only, the most closely relevant class of stochastic games would be the one with undiscounted rewards, as discussed by Vieille [160, 161, 162].

<sup>&</sup>lt;sup>6</sup>For textbook treatments of this class of games, see Luce and Raiffa ([107], Section A8.4), Maitra and Sudderth ([113], Section 7.16). For more recent studies on similar classes of games see Secchi and Sudderth [143] and references therein. Related questions are discussed in Borch [24], Shubik and Thompson [150], Karni and Schmeidler [94].

dynamic market game with one unit of a durable good available per period and a fixed total wealth distributed across the players in exogenous fixed shares. Finally, Giraud and Stahn [77] extended the basic Shapley-Shubik model to a two-period financial economy with uncertainty, allowing short selling, akin to the work presented in Chapter 6 in our setting.

The closest game-theoretic models to our setting are those in capital growth theory considered by Bell and Cover [15, 16], which demonstrate the Kelly portfolio rule as "competitively optimal," established via an appropriate zerosum game.<sup>7</sup> However, there are two crucial differences among them: 1) their models assume exogenous asset prices within a standard framework of capital growth theory, while we extend that theory to a setting with endogenous price formation mechanism; 2) the fundamental game solution concept we adopt is defined in terms of a property holding almost surely, rather than the traditional notion of a Nash equilibrium involving payoff maximization (defined in terms of expectations).

This study's focus on survival aligns it with the evolutionary approach in the social sciences, a tradition that traces back to Malthus, who inspired Darwin (on the history of these ideas, see Hodgeson [90]). Significant contributions to the field in the 20th century were made by Schumpeter [139], Nelson and Winter [123], and others.

For decades, these ideas were mainly theoretical, until the recent crisis turned them from abstract theoretical considerations into the realm of vital practical importance. The emphasis on safeguarding the survival of financial institutions became a policy priority for governments and regulators.

The focus on the "survival and extinction" of investment strategies in the market selection process connects our research to evolutionary game theory (Weibull [165], Vega-Redondo [159], Samuelson [132], Hofbauer and Sigmund [91], Sandholm [134]), which was initially designed for modeling biological systems and later found fruitful applications in economics. The concept of a survival portfolio rule, stable within the market selection process, is akin to the concepts of evolutionary stable strategies (ESS) introduced by Maynard Smith and Price [118], and Schaffer [136, 137]. However, the market selection mechanism in our research is radically different from the typical schemes in evolutionary game theory, where species or agents undergo repeated random matchings in large populations, leading to their long-run survival or extinction.<sup>8</sup> Standard frameworks in that field deal with models based on a given static game that defines the process of evolutionary dynamics, where players follow relatively simple, predefined algorithms that completely characterize their behavior.

Our model differs in its essence. Although the solution concept we deal with

<sup>&</sup>lt;sup>7</sup>For related research in capital growth theory, see Kelly [95], Latané [99], Breiman [34], Algoet and Cover [6], Hakansson and Ziemba [82], MacLean et al. [109], Kuhn and Luenberger [98], Ziemba and Vickson [169], MacLean and Ziemba [110], etc. For the history of these ideas initially expressed by Claude Shannon in his lectures on investment problems [144], and for relevant discussions, see Cover [42]. For textbook treatments towards capital growth theory, see Cover and Thomas [43], Chapter 16; Evstigneev et al. [63], Chapter 17.

<sup>&</sup>lt;sup>8</sup> The evolutionary process may involve random noise (Foster and Young [70], Fudenberg and Harris [71], and Cabrales [39]) and the underlying game may be random (Germano [75]).

is of an evolutionary nature, the notion of the strategy we use is characteristic of the conventional setting in dynamic stochastic games, where a general rule prescribes what action to take based on the history of random states of the world and the observation of all previous play. Players are allowed to employ any such rule, possess all information needed for this purpose, and pursue a clear goal on guaranteed survival. Moreover, it has been shown that the objective of survival in this context aligns with the objective of winning a specific game linked to the original one, i.e., in the sense that "in order to survive you have to win" (for a detailed explanation, see Section 2.4.2).

This paper reviews the study of the following models with short-lived assets in EF.

Firstly, we introduce a model with substantially more general assumptions, serving as the basic framework for the subsequent, more specialized models. A comprehensive exposition of the model structure — e.g., the key concepts, the evolutionary dynamics in the asset market, etc. — is presented. The main focus of this study is to identify the portfolio rules that guarantee an investor to "survive" in the long-run market selection process. It turns out that the existence and asymptotic uniqueness (in a specific class of basic strategies) of survival strategies have been revealed. However, an interesting and more delicate question that we do not consider in this chapter is the analysis of the *single* survivor problem, which will be explored in the subsequent two chapters, under the corresponding, more restrictive assumptions, respectively.

Then, we examine a model with much stronger assumptions: (i) independent and identically distributed (i.i.d.) states of the world, and (ii) a restriction to fixed-mix portfolio rules (prescribing to select investment proportions initially and remain them fixed throughout the entire duration). Under this setting, we obtain two main results: (i) by viewing the model from a different, gametheoretic, perspective and analyzing the Nash equilibrium properties of survival strategies, a unique symmetric almost sure Nash equilibrium can be formed; and as a consequence, (ii) the survival strategy of "betting one's beliefs" leads an investor employing it to be the single survivor with global stability.

Next, we present a model under Markovian nature, which removes the two strong restrictions mentioned above, and instead, uses a homogeneous discretetime Markov process to describe the states of the world. The central result shows that an investor who distributes wealth across available assets according to their relative conditional expected payoffs (a direct analogue of the Kelly rule above) is a single survivor, provided this strategy is asymptotically distinct from the CAPM rule.

Further, we extend the model to a market where risky securities and a riskfree asset (which can be regarded as cash and serves as a numeraire) are traded. A key feature of the model is that the payoffs of the risky assets are assumed to depend linearly on the total cash in the previous period invested by all the market participants. And the main discussion is on the questions of existence and asymptotic uniqueness of the survival strategies.

Finally, we consider a model to incorporate short selling (or endogenous asset supply). In this model, investors' decisions are specified by a pair of

vectors: the vector of long positions defined by investment proportions and the vector of short positions defined by physical units. Our primary interest is also in the fundamental questions of the existence and asymptotic uniqueness of a survival strategy. The result shows, the strategy (without short selling) guarantees survival in a market where the rivals of this investor can sell short, and this strategy is asymptotically unique.

The paper is organized as follows. Chapter 1 delivers an introduction for this research direction – EF with short-lived assets. Chapter 2 studies the results of a basic model under general assumptions. Chapter 3 discusses a model with i.i.d. states of the world and fixed-mix strategies setting. Chapter 4 examines a model with Markov nature. Chapter 5 presents a model including a risk-free asset. Chapter 6 deals with a model with short selling. And Chapter 7 concludes.

## Chapter 2

# Model with General Assumptions

In this chapter, we introduce a model under the most general assumptions, which serves as the basic framework for each special model analyzed in the following chapters. The primary aim is to identify the portfolio rules that allow an investor to "survive" in the long-run market selection process, i.e., to keep a positive, bounded away from zero, share of market wealth over an infinite time horizon. The key results presented in this chapter demonstrate the existence and asymptotic uniqueness of a survival strategy using Kelly portfolio rule.<sup>1</sup>

### 2.1 The Basic Model

The Model Settings. We consider a market where  $K \ge 2$  risky assets (securities) are traded among  $N \ge 2$  investors (traders). A portfolio of investor i at date  $t \ge 0$  is represented by a vector  $x_t^i := (x_{t,1}^i, ..., x_{t,K}^i) \in \mathbb{R}_+^K$ , where  $x_{t,k}^i$  stands for the quantity ("physical units") of asset k held in the portfolio  $x_t^i$ . All the coordinates of  $x_t^i$  are assumed to be non-negative, indicating that short selling is ruled out in this model. The vector  $p_t := (p_{t,1}, ..., p_{t,K}) \in \mathbb{R}_+^K$  denotes the market prices of all assets. For each asset k = 1, ..., K, the coordinate  $p_{t,k}$  in vector  $p_t$  expresses the price of one unit of asset k at date t. The market value of investor i's portfolio  $x_t^i$  at date t is given by the scalar product  $\langle p_t, x_t^i \rangle := \sum_{k=1}^K p_{t,k} x_{t,k}^i$ .

The sequence  $s_1, s_2, ...$  is an exogenous discrete-time stochastic process with values in a measurable space S, where  $s_t$  indicates the state of the world at date t and  $s^t := (s_1, ..., s_t)$  represents the history of the process  $(s_t)$  up to date t. Suppose that we have  $V_{0,k} > 0$  and  $V_{t,k}(s^t) > 0$  total units of each asset k available in the market at date 0 and each subsequent period t = 1, 2, ...,

<sup>&</sup>lt;sup>1</sup>This chapter reviews the study by Amir, Evstigneev, and Schenk-Hoppé [4], a fundamental evolutionary finance model for short-lived assets with substantially more general assumptions.

respectively. Assets live for only one period (short-lived assets): they are traded at the start of each period [t, t+1] and yield payoffs  $A_{t+1,k}$  at the end of it; then the cycle repeats. The payoff  $A_{t,k}(s^t) \geq 0$  for each asset k at date t = 1, 2, ...depends, generally, on t and  $s^t$ , with the property that these measurable payoff functions satisfy

$$\sum_{k=1}^{K} A_{t,k}(s^{t}) > 0 \text{ for all } t, s^{t}.$$
(2.1)

Inequality (2.1) implies that in each random scenario, at least one asset yields a strictly positive payoff.

Investors begin with initial endowments  $w_0^i > 0$  (i = 1, 2, ..., N), which serve as their budgets at date t = 0. For  $t \ge 1$ , investor *i*'s budget is given by  $w_t^i = w_t^i(s^t) := \langle A_t(s^t), x_{t-1}^i(s^{t-1}) \rangle$ , where  $A_t(s^t) := (A_{t,1}(s^t), ..., A_{t,K}(s^t))$ . This budget is formed by the payoffs  $A_t(s^t)$  of the assets in investor *i*'s previous portfolio  $x_{t-1}^i$ , and is re-invested into the assets available at date *t*, which will generate payoffs  $A_{t+1,k}(s^{t+1}), k = 1, ..., K$ , at date t + 1.

For each  $t \ge 0$ , each investor i = 1, 2, ..., N selects a vector of *investment* proportions  $\lambda_t^i := (\lambda_{t,1}^i, ..., \lambda_{t,K}^i)$ , indicating how the available budget  $w_t^i$  is allocated across assets. Vectors  $\lambda_t^i$  belong to the unit simplex

$$\Delta^K := \{ (a_1, ..., a_K) \in \mathbb{R}^K_+ : a_1 + ... + a_K = 1 \}.$$

The investment proportions at each  $t \ge 0$  are chosen by N investors simultaneously and independently (i.e., a simultaneously move N-person dynamic game). In this game under consideration, they represent the players' actions or decisions. For  $t \ge 1$ , the decisions may depend, generally, on the history  $s^t := (s_1, ..., s_t)$  of the process of states of the world and the history of the game

$$\lambda^{t-1} := (\lambda_l^i), \ i = 1, ..., N, \ l = 0, ..., t-1,$$

which includes information about all the previous actions taken by all the players. A portfolio rule, or an investment (trading) strategy  $\Lambda^i$  of investor i, is formed by a vector  $\Lambda_0^i \in \Delta^K$  and a sequence of measurable functions  $\Lambda_t^i(s^t, \lambda^{t-1}), t \geq 1$  with values in  $\Delta^K$ , prescribing his/her choice of investment proportions at each time period  $t \geq 0$ . This framework provides a general game-theoretic definition of a pure strategy, assuming full information about the game history (containing all players' previous actions) and the knowledge of all the past and present states of the world. Within such broad class of general portfolio rules, we will distinguish those basic portfolio rules:  $\Lambda_t^i$  depends only on  $s^t$ , and not on the market history  $\lambda^{t-1}$ .

#### A Description for the Asset Market Dynamics.

1. For date t = 0:

Suppose each investor *i* has chosen his/her investment proportions  $\lambda_0^i = (\lambda_{0,1}^i, ..., \lambda_{0,K}^i) \in \Delta^K$ . Then the amount that trader *i* invests in asset *k* 

is  $\lambda_{0,k}^i w_0^i$ , and the total amount invested in asset k by all traders equals  $\sum_{i=1}^N \lambda_{0,k}^i w_0^i$ . The equilibrium price  $p_{0,k}$  of each asset k can be determined from

$$p_{0,k}V_{0,k} = \sum_{i=1}^{N} \lambda_{0,k}^{i} w_{0}^{i}, \ k = 1, 2, ..., K.$$
(2.2)

The left-hand side of (2.2) shows the total value, expressed in terms of the prevailing price  $p_{0,k}$ , of the *k*th asset purchased by all the market participants at t = 0 (recall that the quantity of each asset *k* at date 0 is  $V_{0,k}$ ). On the right-hand side, we have the total money invested in asset *k* by all the traders.

Then, the investors' portfolios  $x_0^i = (x_{0,1}^i, ..., x_{0,K}^i)$ , i = 1, 2, ..., N, can be determined from the following equations:

$$x_{0,k}^{i} = \frac{\lambda_{0,k}^{i} w_{0}^{i}}{p_{0,k}}, \ k = 1, 2, ..., K, \ i = 1, ..., N, \tag{2.3}$$

indicating that the current market value  $p_{0,k}x_{0,k}^i$  of the *k*th position in portfolio  $x_0^i$  is equal to the proportion  $\lambda_{0,k}^i$  of trader *i*'s budget  $w_0^i$ .

2. For date  $t \geq 1$ :

Suppose now that each investor *i* has selected  $\lambda_t^i = (\lambda_{t,1}^i, ..., \lambda_{t,K}^i) \in \Delta^K$  at date *t*. Then the balance between aggregate monetary asset supply and demand leads to the formula that determines the equilibrium prices  $p_{t,k}$ :

$$p_{t,k}V_{t,k} = \sum_{i=1}^{N} \lambda_{t,k}^{i} \langle A_{t}, x_{t-1}^{i} \rangle, \ k = 1, ..., K,$$
(2.4)

which, in turn, gives the investors' portfolios  $x_t^i = (x_{t,1}^i, ..., x_{t,K}^i)$ :

$$x_{t,k}^{i} = \frac{\lambda_{t,k}^{i} \langle A_{t}, x_{t-1}^{i} \rangle}{p_{t,k}}, \ k = 1, ..., K, \ i = 1, ..., N.$$
(2.5)

Here, in contrast with the case t = 0, the investors' budgets  $w_t^i = \langle A_t, x_{t-1}^i \rangle$ at date  $t \ge 1$  are not given exogenously as initial endowments, rather they are generated by the payoffs of the previous date's portfolios  $x_{t-1}^i$ .

Given a strategy profile  $(\Lambda^1, ..., \Lambda^N)$ ,  $\Lambda^i = \{\lambda_t^i(s^t)\}_{t=0}^{\infty}$ , of the investors, a path of the market game can be generated by setting  $\lambda_0^i = \Lambda_0^i$ , i = 1, ..., N,

$$\lambda_t^i = \Lambda_t^i(s^t, \lambda^{t-1}), \ t = 1, 2, ..., \ i = 1, ..., N,$$
(2.6)

and by defining  $p_t$  and  $x_t^i$  recursively according to Eqs. (2.2)–(2.5).<sup>2</sup> The random dynamical system described determines the state of the market at each

<sup>&</sup>lt;sup>2</sup>Note that one might propose a seemingly more general definition of a strategy, assuming that the traders can use information not about their own and their rivals' decisions, but also about the past prices  $p_0, ..., p_{t-1}$  and portfolios  $x_l^i$ , i = 1, ..., N, l = 0, ..., t-1. However, all these data are determined by the history  $s^{t-1}$  of the states of the world and the traders' decisions  $(\lambda_l^i)$ , i = 1, ..., N, l = 0, ..., t-1 up to date t - 1. Thus this broader definition does not lead, in fact, to a more general notion of a strategy.

date  $t \ge 1$  as a measurable vector function of  $s^t$ :

$$(p_t(s^t); x_t^1(s^t), ..., x_t^N(s^t); \lambda_t^1(s^t), ..., \lambda_t^N(s^t)),$$
(2.7)

where  $p_t(s^t)$ ,  $x_t^i(s^t)$  and  $\lambda_t^i(s^t)$  are the vectors of prices, investors' portfolios and their investment proportions, respectively. (For t = 0, these vectors are constant.)

**Remark 2.1** Portfolio positions  $x_{t,k}^i$  are well-defined by Eqs. (2.3) and (2.5) only if the prices  $p_{t,k}$  are strictly positive for all  $t \ge 0$ , or equivalently, if the aggregate value of demand  $\sum_{i=1}^{N} \lambda_{t,k}^i w_t^i$  for each asset k is strictly positive at each  $t \ge 0$ :

$$\sum_{i=1}^{N} \lambda_{0,k}^{i} w_{0}^{i} > 0 \text{ and } \sum_{i=1}^{N} \lambda_{t,k}^{i} \langle A_{t}, x_{t-1}^{i} \rangle > 0.$$
(2.8)

Strategy profiles that satisfy condition (2.8) for all assets throughout the entire recursive procedure defined above are termed *admissible*, and only such strategy profiles will be examined in the following discussion. This assumption guarantees that the random dynamical system under consideration is well-defined, with  $p_{t,k} > 0$  for all t and k. By summing up Eqs. (2.3) and (2.5) over i = 1, ..., N, we obtain:

$$\sum_{i=1}^{N} x_{0,k}^{i} = \frac{\sum_{i=1}^{N} \lambda_{0,k}^{i} w_{0}^{i}}{p_{0,k}} = \frac{p_{0,k} V_{0,k}}{p_{0,k}} = V_{0,k},$$
(2.9)

and

$$\sum_{i=1}^{N} x_{t,k}^{i} = \frac{\sum_{i=1}^{N} \lambda_{t,k}^{i} \langle A_{t}, x_{t-1}^{i} \rangle}{p_{t,k}} = \frac{p_{t,k} V_{t,k}}{p_{t,k}} = V_{t,k}, \ t \ge 1,$$
(2.10)

indicating that the market clears for each asset k at each date  $t \ge 0$ .

A sufficient condition for a strategy profile to be admissible is provided in hypothesis (2.14) below, which applies to all the strategy profiles we consider. Suppose that one of the traders, e.g., trader 1, follows a *fully diversified* portfolio rule, prescribing to allocate strictly positive proportions  $\lambda_{t,k}^1$  to each asset k = 1, ..., K. Then, any strategy profile that includes such a portfolio rule is admissible. Indeed, for t = 0, we obtain  $p_{0,k} \ge V_{0,k}^{-1}\lambda_{0,k}^1w_0^1 > 0$  from (2.2) and  $x_0^1 = (x_{0,1}^1, ..., x_{t,K}^1) > 0$  (coordinatewise) from (2.3). For  $t \ge 1$ , assuming  $x_{t-1}^1 > 0$  and arguing by induction, it gives  $\langle A_t, x_{t-1}^1 \rangle > 0$  by virtue of (2.1), which in turn yields  $p_t > 0$  from (2.4) and  $x_t^1 > 0$  from (2.5), as long as  $\lambda_{t,k}^1 > 0$ .

**Remark 2.2** Some comments on the stochastic control framework underlying the game at hand. There are two general modeling approaches in discretetime stochastic control theory—both in its conventional, single-agent version, and its game-theoretic setting, where decisions are made by multiple players with different objectives. Models of the first kind are characterized in terms of transition functions (stochastic kernels) specifying the distribution of the state of the system at time t + 1 for each given state and control at time t; see e.g. Shapley [145], Bertsekas and Shreve [18], Dynkin and Yushkevich [52]. In models of the second kind (such as the one in this paper), random factors influencing the system are described in terms of an exogenous random process of states of the world, the distribution of which does not depend on the actions of the players. This approach is often associated with the term "stochastic programming" (e.g., Dynkin [51], Rockafellar and Wets [129], Wallace and Ziemba [164], Birge and Louveaux [19]). Although theoretically both approaches are equivalent in many cases (see Dynkin and Yushkevich [52], Section 2.2), in varying contexts one is more natural and convenient than the other. Generally, the latter is preferable when the model possesses properties of convexity, which is characteristic for economic and financial applications. In a stochastic game setting, the second approach has been pursued by Haurie and coauthors (see, e.g., Haurie et al. [86]), with a primary focus on strategies that depend only on the exogenous states of the world. Haurie et al. [86] call them *S-adapted* (adapted to the given filtration S); we call them *basic* in this paper.

Remark 2.3 Some comments on the modeling of short-run equilibrium in this work. The dynamics of the asset market described above are similar to the dynamics of the commodity market as outlined in the classical treatise by Alfred Marshall [115] (Book V, Chapter II "Temporary Equilibrium of Demand and Supply"). Samuelson ([133], pp. 321–323) introduced Marshall's ideas into formal economics, who noticed that in order to study the process of market dynamics by using the Marshallian "temporary equilibrium method," one needs to distinguish between at least two sets of economic variables changing with different speeds. Then the set of variables changing slower (in our context, investors' portfolios) can be temporarily fixed, while the other set (in our context, the asset prices) can be assumed to rapidly reach the unique state of partial equilibrium.

The above notion of temporary, or moving, equilibrium was first introduced in economics by Marshall. However, in the last decades the term "temporary equilibrium" has been by and large understood differently. For the most part, it was associated with a different concept suggested by Lindahl [100] and Hicks [88], which was further developed in formal settings by Hildenbrand, Grandmont, and others (see, e.g., Grandmont and Hildenbrand [80], Grandmont [78], and Magill and Quinzii [112]). The characteristic feature of the Lindahl-Hicks temporary equilibrium is its formulation in terms of forecasts or beliefs of market participants about the future states of the world. Mathematically, beliefs of economic agents are represented by transition functions (stochastic kernels) that condition the distributions of future states of the world upon the agents' private information.

The models studied in EF do not use information about individual utilities, beliefs and other unobservable agents' characteristics. What matters is the investment strategy as such, rather than the data and the logic on which its choice is based. The results obtained are stated in the form of investment recommendations that use only some fundamental information about the market, in the same spirit as, for example, in the well-known principles of derivative securities pricing (Black-Scholes, Merton, etc.; see, e.g., Evstigneev, et al. [63], Part II).

## 2.2 The Main Results: Existence and Asymptotic Uniqueness of Survival Portfolio Rules

The Notion of Survival Strategies. Consider an admissible strategy profile  $(\Lambda^1, ..., \Lambda^N)$  of the traders and the corresponding path (2.7) of the random dynamical system generated by it. For  $t \ge 1$ , trader *i*'s wealth is

$$w_t^i = w_t^i(s^t) := \langle A_t(s^t), x_{t-1}^i(s^{t-1}) \rangle, \qquad (2.11)$$

and the total market wealth equals

$$W_t := \sum_{i=1}^N w_t^i = \sum_{k=1}^N A_{t,k}(s^t) V_{t-1,k}(s^{t-1}) \ (>0).$$

Our main focus is on the long-run behavior of the *relative wealth*, or the *market* shares,

$$r_t^i := \frac{w_t^i}{W_t}$$

of the investors, i.e., on the asymptotic properties of the sequence of vectors  $r_t = (r_t^1, ..., r_t^N)$  as  $t \to \infty$ .

Given an admissible strategy profile  $(\Lambda^1, ..., \Lambda^N)$ , we say that the strategy  $\Lambda^i$  (or investor *i* using it) survives with probability one if  $\inf_{t\geq 0} r_t^i > 0$  almost surely (a.s.), i.e., for almost all realizations of the process of states of the world  $(s_t)$ , the relative wealth of investor *i* is bounded away from zero by a strictly positive random constant. A portfolio rule  $\Lambda$  is called a survival strategy if investor *i* using it survives with probability one regardless of what portfolio rules  $\Lambda^j$ ,  $j \neq i$ , are employed by the other investors.

**Explicit Expression for the Survival Strategy**  $\Lambda^*$ . To formulate the central result on survival strategies, define the *relative payoffs* of assets k = 1, ..., K at time t as:

$$R_{t,k}(s^{t}) := \frac{A_{t,k}(s^{t})V_{t-1,k}(s^{t-1})}{\sum_{m=1}^{K} A_{t,m}(s^{t})V_{t-1,m}(s^{t-1})},$$
(2.12)

which collectively form the vector  $R_t(s^t) = (R_{t,1}(s^t), ..., R_{t,K}(s^t))$ . Consider the investment strategy  $\Lambda^* = (\lambda_t^*)$  defined by

$$\lambda_t^*(s^t) := E_t R_{t+1}(s^{t+1}), \tag{2.13}$$

where  $E_t(\cdot) = E(\cdot|s^t)$  represents the conditional expectation given  $s^t$  (if t = 0, then  $E_0(\cdot) = E(\cdot)$ , i.e., unconditional expectation). The portfolio rule (2.13), depending only on the history  $s^t$  of the process  $(s_t)$ , prescribes to allocate wealth according to the proportions of the conditional expectations of the relative payoffs. It is a generalization of the Kelly portfolio rule of "betting your beliefs" well-known in capital growth theory — see Kelly [95], Breiman [34], Algoet and Cover [6], Hakansson and Ziemba [82], and Thorp [156]. Suppose that for each asset k = 1, 2, ..., K,

$$E\ln E_t R_{t+1,k}(s^{t+1}) > -\infty.$$
(2.14)

This assumption implies that the conditional expectation  $E_t R_{t+1,k} = E(R_{t+1,k}|s^t)$  is strictly positive a.s., and so we can choose a version of this conditional expectation that is strictly positive for all  $s^t$ . This version,  $\lambda_t^*(s^t)$ , will be used in the definition of the portfolio rule (2.13).

# Two Theorems on the Existence and Asymptotic Uniqueness of the Survival Strategy $\Lambda^*$ .

#### **Theorem 2.1** The portfolio rule $\Lambda^*$ is a survival strategy.

Note that  $\Lambda^*$  belongs to the class of basic portfolio rules, where  $\lambda_t^*(s^t)$  depends only on the history  $s^t$  of the process of states of the world and not on the market history  $\lambda^{t-1}$ . The following theorem demonstrates that within this class, the survival strategy  $\Lambda^* = (\lambda_t^*)$  is essentially unique: any other basic survival strategy is asymptotically similar to  $\Lambda^*$ .

**Theorem 2.2** If  $\Lambda = (\lambda_t)$  is a basic survival strategy, then

$$\sum_{t=0}^{\infty} ||\lambda_t^* - \lambda_t||^2 < \infty \ (a.s.).$$

$$(2.15)$$

Here,  $\|\cdot\|$  denotes the Euclidean norm in a finite-dimensional space.

**Remark 2.4** Theorem 2.2 is akin to various *turnpike* results in the theory of economic growth, suggesting that all optimal or "quasi-optimal" optimal paths of an economic system incline in the long run to essentially the same route, see, e.g., Dorfman et al. [49], Nikaido [126], McKenzie [119], Arkin and Evstigneev [8]. Survival strategies  $\Lambda$  can be characterized by the property that the wealth  $w_i^t$  of any other investors j cannot grow asymptotically faster, with strictly positive probability, than the wealth of investor i using  $\Lambda$  — see Section 2.4.2. The class of such investment strategies shown in Theorem 2.2 is a direct counterpart of Gale's turnpike theorem (Gale [72], Theorem 8) for "good" paths of economic dynamics — paths that cannot be "infinitely worse" than the turnpike; a stochastic version of this result is established in Arkin and Evstigneev [8], Chapter 4, Theorem 6.

The class of basic strategies is *sufficient* in the following sense. Any sequence of market share vectors  $r_t = (r_t^1, ..., r_t^N)$  (where  $r_t = r_t(s^t)$ ) generated by strategy profile  $(\Lambda^1, ..., \Lambda^N)$  can also be generated by a strategy profile  $(\lambda_t^1(s^t), ..., \lambda_t^N(s^t))$  composed of basic portfolio rules, with the corresponding vector functions  $\lambda_t^i(s^t)$  defined recursively by (2.6). Thus it is sufficient to prove Theorem 2.1 only for basic portfolio rules; this will imply that the portfolio rule (2.13) survives in competition with any, not necessarily basic, strategies. However, this reasoning cannot be applied to the problem of asymptotic uniqueness of  $\Lambda^*$  in the class of general survival strategies, and Theorem 2.2 fails to hold in this class — a counterexample is provided in Section 2.4.1.

### 2.3 Proofs of the Main Results

*Proof of Theorem 2.1.* 1st step. We start by deriving a system of equations that describe the dynamics of the market shares  $r_t^i$ . From Eqs. (2.2)–(2.5) and (2.11), we have

$$p_{t,k}V_{t,k} = \langle \lambda_{t,k}, w_t \rangle, \ x_{t,k}^i = \frac{\lambda_{t,k}^i w_t^i V_{t,k}}{\langle \lambda_{t,k}, w_t \rangle},$$
(2.16)

where  $\lambda_{t,k} := (\lambda_{t,k}^1, ..., \lambda_{t,k}^N)$  and  $w_t := (w_t^1, ..., w_t^N)$ . Consequently,

$$w_{t+1}^{i} = \sum_{k=1}^{K} A_{t+1,k} x_{t,k}^{i} = \sum_{k=1}^{K} A_{t+1,k} V_{t,k} \frac{\lambda_{t,k}^{i} w_{t}^{i}}{\langle \lambda_{t,k}, w_{t} \rangle}.$$
 (2.17)

By summing up (2.17) over investors i = 1, ..., N, we get

$$W_{t+1} = \sum_{k=1}^{K} A_{t+1,k} V_{t,k} \frac{\sum_{i=1}^{N} \lambda_{t,k}^{i} w_{t}^{i}}{\langle \lambda_{t,k}, w_{t} \rangle} = \sum_{k=1}^{K} A_{t+1,k} V_{t,k}.$$
 (2.18)

Dividing the left-hand side of (2.17) by  $W_{t+1}$ , the right-hand side of (2.17) by  $\sum_{m=1}^{K} A_{t+1,m} V_{t,m}$ , and applying (2.18) and (2.12), we arrive at the system of equations

$$r_{t+1}^{i} = \sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_{t,k}^{i} r_{t}^{i}}{\langle \lambda_{t,k}, r_{t} \rangle}, \ i = 1, ..., N.$$
(2.19)

2nd step. Observe that it is sufficient to prove Theorem 2.1 in the case of N = 2 investors. Consider the random dynamical system (2.19) and define

$$\tilde{\lambda}_{t,k}^2(s^t) = \begin{cases} (\lambda_{t,k}^2 r_t^2 + \dots + \lambda_{t,k}^N r_t^N) / (1 - r_t^1) & \text{if } r_t^1 < 1, \\ 1/K & \text{if } r_t^1 = 1. \end{cases}$$
(2.20)

Then it gives

$$\begin{split} \lambda_{t,k}^2 r_t^2 + \ldots + \lambda_{t,k}^N r_t^N &= (1 - r_t^1) \tilde{\lambda}_{t,k}^2, \\ \langle \lambda_{t,k}, r_t \rangle &= r_t^1 \lambda_{t,k}^1 + (1 - r_t^1) \tilde{\lambda}_{t,k}^2, \end{split}$$

and so

$$r_{t+1}^{1} = \sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_{t,k}^{1} r_{t}^{1}}{r_{t}^{1} \lambda_{t,k}^{1} + (1 - r_{t}^{1}) \tilde{\lambda}_{t,k}^{2}}.$$
(2.21)

By summing up formula (2.19) over i = 2, ..., N, we have

$$1 - r_{t+1}^{1} = \sum_{k=1}^{K} R_{t+1,k} \frac{\tilde{\lambda}_{t,k}^{2} (1 - r_{t}^{1})}{r_{t}^{1} \lambda_{t,k}^{1} + (1 - r_{t}^{1}) \tilde{\lambda}_{t,k}^{2}}.$$
 (2.22)

Therefore, the sequence of market share vectors  $(r_t^1(s^t))$  generated by the original N-dimensional system (2.19) is identical to the analogous sequence generated by the two-dimensional system (2.21)–(2.22), which corresponds to the game involving two investors i = 1, 2 with investment proportions  $\lambda_{t,k}^1(s^t)$  and  $\tilde{\lambda}_{t,k}^2(s^t)$ , respectively.

3rd step. Assume N = 2 and  $\lambda_{t,k}^1 = \lambda_{t,k}^*$ . Since  $\lambda_{t,k}^* > 0$ , our standing hypothesis on the strict positivity of investor 1's investment proportions holds. Defining  $\kappa_t = \kappa_t(s^t) := r_t^1(s^t)$ , we obtain from (2.19) with N = 2:

$$\kappa_{t+1} = \sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_{t,k}^1 \kappa_t}{\lambda_{t,k}^1 \kappa_t + \lambda_{t,k}^2 (1-\kappa_t)}$$

Observe that the process  $\ln \kappa_t$  is a submartingale. Indeed, we have

$$E_{t} \ln \kappa_{t+1} - \ln \kappa_{t} = E_{t} \ln \sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_{t,k}^{1}}{\lambda_{t,k}^{1} \kappa_{t} + \lambda_{t,k}^{2} (1 - \kappa_{t})}$$

$$\geq E_{t} \sum_{k=1}^{K} R_{t+1,k} \ln \frac{\lambda_{t,k}^{1}}{\lambda_{t,k}^{1} \kappa_{t} + \lambda_{t,k}^{2} (1 - \kappa_{t})}$$

$$= \sum_{k=1}^{K} \lambda_{t,k}^{1} \ln \frac{\lambda_{t,k}^{1}}{\lambda_{t,k}^{1} \kappa_{t} + \lambda_{t,k}^{2} (1 - \kappa_{t})}$$

$$= \sum_{k=1}^{K} \lambda_{t,k}^{1} \ln \lambda_{t,k}^{1} - \sum_{k=1}^{K} \lambda_{t,k}^{1} \ln [\lambda_{t,k}^{1} \kappa_{t} + \lambda_{t,k}^{2} (1 - \kappa_{t})] \ge 0 \text{ (a.s.)}.$$

Here, we applied Jensen's inequality for the concave function  $\ln x$  along with the elementary inequality

$$\sum_{k=1}^{K} a_k \ln a_k \ge \sum_{k=1}^{K} a_k \ln b_k \quad [\ln 0 := -\infty]$$
(2.23)

holding for any vectors  $(a_1, ..., a_K) > 0$  and  $(b_1, ..., b_K) \ge 0$  with  $\sum a_k = \sum b_k = 1$  (see Lemma 2.2 below).

Further, we have

$$\kappa_{t+1} = \kappa_t \sum_{k=1}^K R_{t+1,k} \frac{\lambda_{t,k}^1}{\lambda_{t,k}^1 \kappa_t + \lambda_{t,k}^2 (1 - \kappa_t)}$$
$$\geq \kappa_t \sum_{k=1}^K R_{t+1,k} (\min_m \lambda_{t,m}^1) = \kappa_t (\min_m \lambda_{t,m}^1).$$

Since  $E \min_m \ln \lambda_{t,m}^1 > -\infty$  from assumption (2.14) and  $\kappa_0$  is a strictly positive non-random number, each of the random variables  $0 < \kappa_t \leq 1$  satisfies  $E |\ln \kappa_t| < \infty$ .

The non-positive submartingale  $\ln \kappa_t$  has a finite limit a.s., which implies  $\kappa_t \to \kappa_\infty$  (a.s.), where  $\kappa_\infty$  is a strictly positive random variable. Therefore,

the sequence  $\kappa_t > 0$  is bounded away from zero with probability one, indicating that investor 1 survives almost surely.

The proof of Theorem 2.2 is based on the following two lemmas.

**Lemma 2.1** Suppose  $\xi_t$  is a submartingale such that  $\sup_t E\xi_t < \infty$ . Then, the sequence of non-negative random variables  $\sum_{t=0}^{\infty} (E_t\xi_{t+1} - \xi_t)$  converges a.s.

*Proof.* From the definition of submartingale, we get  $\zeta_t := E_t \xi_{t+1} - \xi_t \ge 0$ . Further, we obtain

$$\sum_{t=0}^{T-1} E\zeta_t = \sum_{t=0}^{T-1} (E\xi_{t+1} - E\xi_t) = E\xi_T - E\xi_0,$$

and so the series  $\sum_{t=0}^{T-1} E\zeta_t$  is bounded because  $\sup_T E\xi_T < \infty$ . Consequently, the series of expectations  $\sum_{t=0}^{\infty} E\zeta_t$  of the non-negative random variables  $\zeta_t$  converges, which in turn implies  $\sum_{t=0}^{\infty} \zeta_t < \infty$  a.s., since  $E\sum_{t=0}^{\infty} \zeta_t = \sum_{t=0}^{\infty} E\zeta_t$  (see, e.g., Saks [131], Theorem I.12.3).

**Lemma 2.2** <sup>3</sup>For any vectors  $(a_1, ..., a_K) > 0$  and  $(b_1, ..., b_K) \ge 0$  that satisfy  $\sum a_k = \sum b_k = 1$ , the following inequality holds:

$$\sum_{k=1}^{K} a_k \ln a_k - \sum_{k=1}^{K} a_k \ln b_k \ge \frac{1}{4} \sum_{k=1}^{K} (a_k - b_k)^2.$$
(2.24)

*Proof.* We have  $\ln x \le x - 1$ , which gives  $(\ln x)/2 = \ln \sqrt{x} \le \sqrt{x} - 1$ , and thus  $-\ln x \ge 2 - 2\sqrt{x}$ . From this inequality, we obtain

$$\sum_{k=1}^{K} a_k (\ln a_k - \ln b_k) = -\sum_{k=1}^{K} a_k \ln \frac{b_k}{a_k} \ge \sum_{k=1}^{K} a_k (2 - 2\frac{\sqrt{b_k}}{\sqrt{a_k}})$$
$$= 2 - 2\sum_{k=1}^{K} \sqrt{a_k b_k} = \sum_{k=1}^{K} (a_k - 2\sqrt{a_k b_k} + b_k)$$
$$= \sum_{k=1}^{K} (\sqrt{a_k} - \sqrt{b_k})^2.$$

This leads to (2.24) because  $(\sqrt{a} - \sqrt{b})^2 \ge (a - b)^2/4$  for  $0 \le a, b \le 1$ .

Proof of Theorem 2.2. Let  $\Lambda = (\lambda_t)$  be a basic survival strategy. Assume that investors i = 1, 2, ..., N - 1 follow the strategy  $\Lambda^* = (\lambda_t^*)$  and investor N uses  $\Lambda$ . By summing up (2.19) with  $\lambda_t^i = \lambda_t^*$  over i = 1, ..., N - 1, it gives

$$\hat{r}_{t+1}^{1} = \sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_{t,k}^{*} \hat{r}_{t}^{1}}{\lambda_{t,k}^{*} \hat{r}_{t}^{1} + \lambda_{t,k} (1 - \hat{r}_{t}^{1})},$$

<sup>&</sup>lt;sup>3</sup>Lemma 2.2 follows from an inequality between the *Kullback-Leibler divergence* (generalizing the left-hand side of (2.24)) and the *Hellinger distance* (which reduces to  $[\sum(\sqrt{a_k} - \sqrt{b_k})^2]^{1/2}$ ) — see, e.g., Borovkov [26], Section II.31.

where  $\hat{r}_t^1 := r_t^1 + \ldots + r_t^{N-1}$  is the market share of the group of investors  $i = 1, 2, \ldots, N-1$  and  $1 - \hat{r}_t^1 = r_t^N$  is that of investor N. Further, we get

$$1 - \hat{r}_{t+1}^1 = \sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_{t,k} (1 - \hat{r}_t^1)}{\lambda_{t,k}^* \hat{r}_t^1 + \lambda_{t,k} (1 - \hat{r}_t^1)}$$

Hence, the dynamics of  $\hat{r}_t^1 = r_t^1 + \ldots + r_t^{N-1}$ ,  $1 - \hat{r}_t^1 = r_t^N$  is exactly the same as those of  $\hat{r}_t^1, \hat{r}_t^2 = 1 - \hat{r}_t^1$  of two investors i = 1, 2 (N = 2) with the strategies  $(\lambda_t^1) = (\lambda_t^*)$  and  $(\lambda_t^2) = (\lambda_t)$ , respectively. Since  $(\lambda_t)$  is a survival strategy, the random series  $r_t^N = 1 - \hat{r}_t^1 = \hat{r}_t^2$  is bounded away from zero a.s.

As we establish in the proof of Theorem 2.1 (step 3), the random sequence  $\ln \kappa_t := \ln \hat{r}_t^1$  is a non-positive submartingale satisfying

$$E_t \ln \kappa_{t+1} - \ln \kappa_t$$

$$\geq \sum_{k=1}^K \lambda_{t,k}^* \ln \lambda_{t,k}^* - \sum_{k=1}^K \lambda_{t,k}^* \ln[\lambda_{t,k}^* \kappa_t + \lambda_{t,k} (1 - \kappa_t)] \text{ (a.s.)}.$$
(2.25)

According to Lemma 2.1, the sequence  $\sum (E_t \ln \kappa_{t+1} - \ln \kappa_t)$  of non-negative random variables converges almost surely. Based on inequalities (2.24) and (2.25), this suggests that the sum

$$\sum_{t=0}^{\infty} \sum_{k=1}^{K} [\lambda_{t,k}^* - \lambda_{t,k}^* \kappa_t - \lambda_{t,k} (1 - \kappa_t)]^2 = \sum_{t=0}^{\infty} (1 - \kappa_t)^2 ||\lambda_t^* - \lambda_t||^2$$
(2.26)

is finite with probability one. Since  $\inf(1 - \kappa_t) = \inf \hat{r}_t^2 > 0$  a.s., the fact that the sequence in (2.26) converges a.s. implies (2.15).

### 2.4 Discussions on the Main Results

#### 2.4.1 A Counterexample

In this section, an example is provided to illustrate that Theorem 2.2 cannot be naturally extended to the class of general, not necessarily basic, portfolio rules. One might conjecture that the following extension of Theorem 2.2 is valid: if  $\Lambda^1_t(\cdot)$  is a general survival strategy of player 1, then for any set of strategies  $\Lambda^2_t(\cdot),...,\Lambda^N_t(\cdot)$  employed by players 2, 3, ..., N, the investment proportion vectors  $\lambda^1_t(s^t)$  of investor 1 generated by the strategy profile  $(\Lambda^1_t(\cdot),\Lambda^2_t(\cdot),...,\Lambda^N_t(\cdot))$  will converge to  $\lambda^*_t(s^t)$  (in one sense or another) as  $t \to \infty$ . Below is a counterexample to this conjecture.<sup>4</sup>

A Constructed Model Serving as Counterexample. There are two traders (N = 2). The random states of the world  $s_1, s_2, ...$  are independent and identically distributed in the state space S consisting of two elements:  $S = \{1, 2\}$ ,

<sup>&</sup>lt;sup>4</sup>The strategy in this counterexample is constructed in a way that is akin to the *trigger* strategies described in the "folk theorems" of game theory (see, e.g., Myerson [122]).

with probabilities  $P\{s_t = 1\} = 1/3$  and  $P\{s_t = 2\} = 2/3$ . There are two assets k = 1, 2 with  $V_{t,k}(s^t) = 1$  for all  $t, s^t, k = 1, 2$ , and  $A_{t,k}(s^t) = A_k(s_t)$ , where  $A(1) = (A_1(1), A_2(1)) = (1, 0), A(2) = (A_1(2), A_2(2)) = (0, 1)$ . In this model,  $R_{t,k}(s^t) = R_k(s_t)$ , where

$$R(1) = (R_1(1), R_2(1)) = (1, 0), \ R(2) = (R_1(2), R_2(2)) = (0, 1),$$
(2.27)

and

$$\lambda^* = E_t R(s_{t+1}) = ER(s_{t+1}) = (1/3, 2/3).$$

Consider some vector  $0 < b \neq \lambda^*$  in the unit simplex  $\Delta^2$  and a set  $\Gamma \subseteq \Delta^2$  such that

$$E\frac{\langle a, R(s)\rangle}{\langle b, R(s)\rangle} \le 1, \ a \in \Gamma.$$
(2.28)

As an example, let  $b = (b_1, b_2) = (1/4, 3/4)$  and

$$\Gamma = \{ a = (a_1, 1 - a_1) : 0 \le a_1 \le 1/4 \}.$$

Then (2.28) holds, since

$$E\frac{\langle a, R(s)\rangle}{\langle b, R(s)\rangle} = \frac{1}{3}\frac{a_1}{b_1} + \frac{2}{3}\frac{a_2}{b_2} = \frac{4}{3}a_1 + \frac{8}{9}(1-a_1) = \frac{4a_1+8}{9} \le 1,$$

as long as  $a_1 \leq 1/4$ .

Define the strategy  $\Lambda_t^1(s^t, \lambda^{t-1}), t = 0, 1, ...,$  of player 1 as follows. For t = 0 put  $\Lambda_0^1 = b$ . For  $t \ge 1$ , define

$$\Lambda_t^1(\lambda_0^2, ..., \lambda_{t-1}^2) = \begin{cases} b, & \text{if } \lambda_l^2 \in \Gamma, \ l = 0, ..., t-1; \\ \lambda^*, & \text{otherwise.} \end{cases}$$
(2.29)

The action prescribed by this strategy is independent of  $s^t$  and player 1's previous actions  $\lambda_l^1$ , l = 0, ..., t - 1, depending only on the previous actions  $\lambda_l^2$ , l = 0, ..., t - 1 of player 2, the rival of player 1. This strategy says, if the rival at least once selects a vector of investment proportions outside the set  $\Gamma$ , then 1 immediately adopts  $\lambda^*$  and sticks with it forever. Otherwise, 1 always plays b.

**Proposition 2.1** The portfolio rule  $\Lambda_t^1(\lambda_0^2, ..., \lambda_{t-1}^2)$  is a survival strategy. There exists a whole class  $\mathcal{L}$  of strategies  $\Lambda_t^2(\cdot)$  of player 2 such that the vectors  $\lambda_t^1(s^t)$  of investment proportions of player 1 generated by the strategy profile  $(\Lambda_t^1(\cdot), \Lambda_t^2(\cdot))$  coincide for all t with the constant vector  $b \neq \lambda^*$  (and hence do not converge to  $\lambda^*$  in any sense). The class  $\mathcal{L}$  includes all the basic strategies  $(\lambda_t^2(s^t))$  such that  $\lambda_t^2(s^t) \in \Gamma$  for all t,  $s^t$ .

*Proof.* Consider any strategy  $\Lambda_t^2(\cdot)$  of player 2. The strategy profile  $(\Lambda_t^1(\cdot), \Lambda_t^2(\cdot))$  is admissible because  $\Lambda_t^1(\cdot) > 0$ . Therefore, the path of the game is well-defined and the market shares  $r_t^1, r_t^2$  of players i = 1, 2 satisfy

$$r_{t+1}^{i} = \sum_{k=1}^{2} R_{k}(s_{t+1}) \frac{\lambda_{t,k}^{i} r_{t}^{i}}{\lambda_{t,k}^{1} r_{t}^{1} + \lambda_{t,k}^{2} r_{t}^{2}}, \ i = 1, 2$$
(2.30)

(see (2.19)), where  $(\lambda_{t,1}^i, \lambda_{t,2}^i) = (\lambda_{t,1}^i(s^t), \lambda_{t,2}^i(s^t))$  are the sequences of vectors of investment proportions of players i = 1, 2 generated by the strategy profile  $(\Lambda^1_t(\cdot), \Lambda^2_t(\cdot))$ . We need to show that  $r^1_t$  is bounded away from zero (a.s.), or equivalently, that the random sequence  $\gamma_t := r_t^2/r_t^1$  is bounded above (a.s.).

By virtue of (2.27), we obtain

$$\gamma_{t+1} = \frac{\lambda_{t,s_{t+1}}^2 r_t^2 / (\lambda_{t,s_{t+1}}^1 r_t^1 + \lambda_{t,s_{t+1}}^2 r_t^2)}{\lambda_{t,s_{t+1}}^1 r_t^1 / (\lambda_{t,s_{t+1}}^1 r_t^1 + \lambda_{t,s_{t+1}}^2 r_t^2)} = \frac{\lambda_{t,s_{t+1}}^2 r_t^2}{\lambda_{t,s_{t+1}}^1 r_t^1} = \frac{\langle \lambda_t^2, R(s_{t+1}) \rangle}{\langle \lambda_t^1, R(s_{t+1}) \rangle} \gamma_t. \quad (2.31)$$

Denote by  $\tau = \tau(s_1, s_2, ...)$  the moment of time such that  $\lambda_{\tau}^2 \notin \Gamma, \lambda_{\tau-1}^2 \in \Gamma$ , ..., $\lambda_0^2 \in \Gamma$ , if such a moment of time exists; otherwise, put  $\tau = \infty$ . Hence,  $\tau < \infty$  is the first moment of time when the random sequence  $\lambda_t^2$  leaves the set  $\Gamma$ , and  $\tau = \infty$  if this sequence always stays in  $\Gamma$ .

If  $\tau = \infty$ , then  $\gamma_{t+1} = \beta_{t+1}\gamma_0$ , where

$$\beta_{t+1} = \frac{\langle \lambda_t^2, R(s_{t+1}) \rangle}{\langle b, R(s_{t+1}) \rangle} \dots \frac{\langle \lambda_0^2, R(s_1) \rangle}{\langle b, R(s_1) \rangle}$$

(see (2.29) and (2.31)). If  $\tau < \infty$ , then for each  $t > \tau$  we have

$$\gamma_{t+1} = \frac{\langle \lambda_t^2, R(s_{t+1}) \rangle}{\langle \lambda^*, R(s_{t+1}) \rangle} \dots \frac{\langle \lambda_{\tau+1}^2, R(s_{\tau+2}) \rangle}{\langle \lambda^*, R(s_{\tau+2}) \rangle} \cdot \frac{\langle \lambda_{\tau}^2, R(s_{\tau+1}) \rangle}{\langle b, R(s_{\tau+1}) \rangle} \dots \frac{\langle \lambda_0^2, R(s_1) \rangle}{\langle b, R(s_1) \rangle} \gamma_0 = \alpha_{t+1} \delta,$$

where

$$\alpha_{t+1} := \frac{\langle \lambda_t^2, R(s_{t+1}) \rangle}{\langle \lambda^*, R(s_{t+1}) \rangle} \dots \frac{\langle \lambda_0^2, R(s_1) \rangle}{\langle \lambda^*, R(s_1) \rangle}, \ \delta := \frac{\langle \lambda^*, R(s_{\tau+1}) \rangle}{\langle b, R(s_{\tau+1}) \rangle} \dots \frac{\langle \lambda^*, R(s_1) \rangle}{\langle b, R(s_1) \rangle} \gamma_0.$$

Fix some element  $\bar{a} \in \Gamma$ , and define  $\bar{\lambda}_t^2 := \bar{a}$  if  $\lambda_t^2 \notin \Gamma$  and  $\bar{\lambda}_t^2 := \lambda_t^2$  if  $\lambda_t^2 \in \Gamma$ . Then if  $\tau = \infty$ , we obtain  $\beta_{t+1} = \bar{\beta}_{t+1}$  (t = 0, 1, ...), where

$$\bar{\boldsymbol{\beta}}_{t+1} := \frac{\langle \bar{\lambda}_t^2, \boldsymbol{R}(s_{t+1}) \rangle}{\langle \boldsymbol{b}, \boldsymbol{R}(s_{t+1}) \rangle} ... \frac{\langle \bar{\lambda}_0^2, \boldsymbol{R}(s_1) \rangle}{\langle \boldsymbol{b}, \boldsymbol{R}(s_1) \rangle}$$

because in this case  $\lambda_t^2 \in \Gamma$ , and so  $\bar{\lambda}_t^2 = \lambda_t^2$ , for all t. Observe that the process  $\bar{\beta}_t$  is a non-negative supermartingale (with respect to the system of  $\sigma$ -algebras generated by  $s^t$ ). Indeed, in view of (2.28),

$$E(\bar{\boldsymbol{\beta}}_{t+1}|s^t) = \bar{\boldsymbol{\beta}}_t E \frac{\langle a, R(s) \rangle}{\langle b, R(s) \rangle} \leq \bar{\boldsymbol{\beta}}_t$$

where  $a := \overline{\lambda}_t^2(s^t) \in \Gamma$ . Thus the process  $\overline{\beta}_t$  converges a.s., and hence is bounded a.s.. Consequently, if  $\tau = \infty$ , the process  $\gamma_t = \beta_t \gamma_0 = \bar{\beta}_t \gamma_0$  is bounded a.s.. Further, observe that the sequence  $\alpha_t$  is a non-negative martingale. Indeed,

if we fix  $s^t$  and put  $a := (a_1, a_2) = \lambda_t^2(s^t)$ , then

$$E(\alpha_{t+1}|s^t) = \alpha_t E \frac{\langle a, R(s) \rangle}{\langle \lambda^*, R(s) \rangle} = \alpha_t$$

because

$$E\frac{\langle a, R(s)\rangle}{\langle \lambda^*, R(s)\rangle} = \frac{1}{3}\frac{\langle a, R(1)\rangle}{\langle \lambda^*, R(1)\rangle} + \frac{2}{3}\frac{\langle a, R(2)\rangle}{\langle \lambda^*, R(2)\rangle} = \frac{1}{3}\frac{a_1}{1/3} + \frac{2}{3}\frac{a_2}{2/3} = 1.$$

Therefore  $\alpha_t$  is bounded a.s., and so if  $\tau < \infty$ , then

$$\sup_{t>\tau} \gamma_{t+1} = \delta \sup_{t>\tau} \alpha_{t+1} < \infty \text{ (a.s.)}.$$

Thus we have obtained that the process  $\gamma_t$  is bounded a.s. both when  $\tau = \infty$  and when  $\tau < \infty$ . This proves that  $\Lambda^1_t(\cdot)$  is a survival strategy.

It remains to observe that if player 2 uses a strategy (e.g., a basic one) for which  $\lambda_t^2(s^t) \in \Gamma$  for all  $t, s^t$ , then we have  $\lambda_t^1 = b \neq \lambda^*$  for all  $t, s^t$ .

#### 2.4.2 Strategies of Survival as Unbeatable Strategies

The Notion of Unbeatable Strategies. Consider an abstract game of N players i = 1, ..., N choosing strategies  $\Lambda^i$  from some given sets. Suppose  $w^i = w^i(\Lambda^1, ..., \Lambda^N)$  is the outcome of the game for player i corresponding to the strategy profile  $(\Lambda^1, ..., \Lambda^N)$ , where possible outcomes  $w^i$  are elements of a set  $W^i$ . Assume that a preference relation

$$w^i \succcurlyeq_{ij} w^j, w^i \in \mathcal{W}^i, \ w^j \in \mathcal{W}^j, \ i \neq j,$$

is given, allowing for the comparison of the relative performance of players i and j. A strategy  $\Lambda$  of player i is called *unbeatable* if for any admissible strategy profile  $(\Lambda^1, \Lambda^2, ..., \Lambda^N)$  in which  $\Lambda^i = \Lambda$ , we have

$$w^{i}(\Lambda^{1}, \Lambda^{2}, ..., \Lambda^{N}) \succcurlyeq_{ij} w^{j}(\Lambda^{1}, \Lambda^{2}, ..., \Lambda^{N}) \text{ for all } j \neq i.$$
(2.32)

Thus, if player *i* employs the strategy  $\Lambda$ , then this investor cannot be outperformed by any of his/her rivals  $j \neq i$ , irrespective of what strategies they use.

Let us return to the game we consider. Denote by  $\mathcal{W}$  the set of sequences of positive random variables, and define the following relation  $\succeq$  between two sequences  $\alpha = (\alpha_t)$  and  $\beta = (\beta_t)$  in  $\mathcal{W}$ . We write  $\beta \succeq \alpha$  (or  $\alpha \preccurlyeq \beta$ ) if there exists a random variable H > 0 such that  $\alpha_t \leq H\beta_t$  (a.s.), meaning that almost surely, the sequence  $\alpha$  does not grow asymptotically faster than  $\beta$ . Consider the asset market game with N players, each having initial endowments  $w_0^1 > 0, ..., w_0^N > 0$ . Given an admissible strategy profile  $(\Lambda^1, ..., \Lambda^N)$  of Ninvestors, we obtain a sequence  $w^i := (w_0^i, w_1^i, ...)$  of positive random variables for each i = 1, 2, ..., N, where  $w_t^i$  stands for investor i's wealth at date t. As we have seen, the evolution of the vectors  $w_t := (w_t^1, ..., w_t^N)$  is governed by the random dynamical system (2.17). We can view the sequence  $w^i = w^i(\Lambda^1, ..., \Lambda^N)$ as an outcome of the game for player i corresponding to the strategy profile  $(\Lambda^1, ..., \Lambda^N)$ , and apply the above definition with  $\mathcal{W}^i = \mathcal{W}$  and the preference relation  $\succeq$  (independent of i and j) to this specific game at hand. Its meaning in the present context is as follows: a portfolio rule  $\Lambda$  is an unbeatable strategy of investor i if it guarantees the asymptotically fastest growth rate of investor *i*'s wealth, irrespective of what strategies used by the other investors.

#### The Relation between "Survival" and "Unbeatable".

Proposition 2.2 A portfolio rule is a survival strategy if and only if it is unbeatable.

*Proof.* Since the market shares are expressed as  $r_t^i = w_t^i/W_t$ , relations (2.32) hold if and only if

$$r^i \succcurlyeq r^j \text{ for all } j \neq i.$$
 (2.33)

If the market share  $r_t^i$  of investor *i* satisfies  $r_t^i \ge c$  (a.s.), where *c* is a strictly positive random variable, then  $w_t^i \ge cW_t \ge cw_t^j$  (a.s.) for all *j*. Thus  $w_t^j \le c^{-1}w_t^i$  (a.s.), and so (2.32) holds. Conversely, if  $w_t^j \le Hw_t^i$  (a.s.) for some random variable H > 0, then  $W_t \leq [(N-1)H + 1]w_t^i$  (a.s.), which yields  $r_t^i \ge [(N-1)H+1]^{-1}$  (a.s.).

**Remark 2.5** The idea of an unbeatable (or *winning*) strategy was central in the pre-Nash game theory. At those times, solving a game meant primarily finding a winning strategy that yields the player who uses it a superiority over any other players who do not use it, see e.g., Bouton [33], Borel [25]. Extensive work has been done around the notion of *determinacy* of a game, which is formulated in terms of winning strategies: a game is determined if one of the players has a winning strategy.<sup>5</sup> The problem of finding a winning strategy in two-person zero-sum games essentially reduces to finding an equilibrium (minimax) strategy. For this reason, questions of this kind were typically formulated and examined in terms of minimax strategies in zero-sum games, especially after the seminal paper by von Neumann [163]. In the 1950s, when game theory started developing primarily as a mathematical framework of economic modeling, non-zero sum N-player games came to the fore, and the notion of Nash equilibrium became central to the field.

The concept of an unbeatable strategy as such reemerged in theoretical biology, serving as the starting point for the development of evolutionary game theory. Hamilton [84] used this notion, and the term "unbeatable strategy" though without rigorous formalization — in his paper on the analysis of sex ratios in populations of certain species. Later, Maynard Smith and Price [118] formalized Hamilton's idea, but at the same time somewhat modified its content. The notion commonly known as the Maynard Smith's evolutionary stable strategy (ESS) can be described as a *conditionally unbeatable* strategy. It cannot be beaten as long as the rival is "weak enough." In the context of evolutionary biology, Maynard Smith's ESS refers to a strategy that cannot be beaten if the fraction of the rivals (mutants) in the population is sufficiently small.<sup>6</sup> This definition assumes an infinite population, as it involves considering arbitrarily

<sup>&</sup>lt;sup>5</sup>For relavant studies, see, e.g., Zermelo [166], Schwalbe and Walker [140], Gale and Stewart [73], Martin [116], Telgársky [153].
 <sup>6</sup> An unconditional variant of the Maynard Smith's ESS was considered by Kojima [97].

small fractions of the population. Versions of Maynard Smith's ESS applicable to finite populations were considered by Schaffer [136, 137]. Schaffer's notions of ESS, the weaker and the stronger ones, are also conditionally unbeatable strategies. The former assumes that the population contains only *one* mutant, while the latter considers the presence of several identical mutants. Having in mind this, evolutionary, branch of the history of unbeatable strategies, it is not surprising that they have appeared in our EF context.

## Chapter 3

# Model with I.I.D. States of the World and Fixed-Mix Strategies

An interesting and more delicate question, which we did not consider in Chapter 2, is the analysis of conditions on the strategy profile of investors under which the Kelly rule  $\Lambda^*$  is the *single* survivor in the market selection process. More precisely, suppose investor 1 uses  $\Lambda^*$ , while all the others use some other strategies  $\Lambda^2, ..., \Lambda^N$  distinct from  $\Lambda^*$ . Under what conditions on  $(\Lambda^2, ..., \Lambda^N)$ , the market share of investor 1 tends to one almost surely (so that he/she not only survives, but drives all the others out of the market) in the long run?

This question was examined by Blume and Easley [21] in the case of Arrow securities, fixed-mix strategies and i.i.d. states of the world  $s_t$ . In this case, when the market is complete, the analysis essentially reduces to the classical capital growth theory with exogenous asset returns (Algoet and Cover [6]), from which the results on the single survivor follow. In models of incomplete markets, the analogous question was studied by Evstigneev et al. [58] (a finite space S with i.i.d.  $s_t$ , fixed-mix strategies only) and Amir et al. [2] (a finitestate stationary Markov  $s_t$ , basic strategies only), which we will present in this chapter and the next chapter, respectively. However, the analysis of the single survivor problem in the most general framework given in Chapter 2, is still open for further research.

In this chapter, a special case of the model (as mentioned above, i.i.d. states of the world and fixed-mix strategies) is given in Section 3.1. Then, we analyze it from a different, game-theoretic, perspective in Section 3.2. We consider a game associated with this model in which the payoffs of the players are defined in terms of the growth rates of their relative wealth. It turns out that in the game under consideration the survival strategy  $\Lambda^*$  forms with probability one a unique symmetric Nash equilibrium.

This study establishes a link between the EF models and the classical ones,

e.g., game-theoretic models of asset markets dealing with relative wealth of investors considered by Bell and Cover [15, 16]). In those models, the notion of Nash equilibrium was defined in terms of the expectations of random payoffs. Whereas, we consider a different (stronger) solution concept: almost sure Nash equilibrium. According to this definition of an equilibrium strategy, any unilateral deviation from it leads to a decrease in the random payoff with probability one, not just in the expected payoff. For formulating this evolutionary solution concept, a strategy strictly dominating the market at an exponential rate, we introduce the Lyapunov exponent of the relative growth of wealth of an investor as the objective function. Here, "relative" refers to the fraction of wealth of the investor and the total wealth of the group of his/her rivals.

And so, the result of a single survivor with global stability becomes a direct consequence of the conclusion of exponential market domination in the almost sure Nash equilibrium that we proved. The relevant introduction for this part is outlined in Section 3.3.  $^1$ 

### 3.1 The Model

A Summary on Changes in Model Settings. The model framework is substantially consistent with the general model given in Section 2.1, only with the addition of two stronger assumptions as follows:

• uncertainty is modeled by a sequence  $s_t$ , t = 1, 2, ..., of independent and identically distributed (i.i.d.) random variables (states of the world), taking values in a finite space S. In each period the states are drawn in accordance with the probability distribution  $p = (p_s)_{s \in S}$ ,  $p_s > 0$ . We use P to denote the product measure of p constructed on the space of sample paths  $(s_t)_{t=1,2,...}$ , and E denotes the expectation with respect to the measure P.

Accordingly, the model has the following changes:

- the total amount of security k in the market is  $V_k > 0$ .
- The payoff  $A_{t,k} = A_k(s_t) \ge 0$  of asset k = 1, 2, ..., K at date t = 1, 2, ..., K at date t = 1, 2, ... depends only on the state of the world  $s_t$  at that date. The functions  $A_k(s), s \in S$ , are measurable and satisfy

$$\sum_{k=1}^{K} A_k(s) > 0 \text{ for all } s.$$

$$(3.1)$$

- Assume the no redundant assets assumption holds: the functions  $A_1(s), ..., A_K(s)$  (equivalently, the functions  $R_1(s), ..., R_K(s)$ ) are linearly independent with respect to the probability distribution of  $s_t$ ,

 $<sup>^{1}</sup>$  The analysis in Section 3.2 is based on the work of Belkov, Evstigneev, Hens, and Xu [14], and the analysis in Section 3.3 is based on the paper by Evstigneev, Hens, and Schenk-Hoppé [58].

i.e. the equality  $\sum \gamma_k A_k(s_t) = 0$  holding a.s. for some constants  $\gamma_k$  implies that  $\gamma_1 = \ldots = \gamma_K = 0$ .

• we focus exclusively on fixed-mix strategies, and a "strategy" ("portfolio rule") will always mean a "fixed-mix strategy" ("fixed-mix portfolio rule").

For the asset market dynamics, the equilibrium prices  $p_{t,k}$  of each asset k and the investors' portfolios  $x_t^i = (x_{t,1}^i, ..., x_{t,K}^i)$ , i = 1, 2, ..., N for  $t \ge 0$  can be determined from the recursive process in terms of Eqs. (2.2)–(2.5) as we stated in Chapter 2. Note that here  $V_k = V_{t,k}$ ,  $\lambda_k^i = \lambda_{t,k}^i$ , k = 1, 2, ..., K, i = 1, 2, ..., N for  $t \ge 0$ .

It follows from (2.10) that the total market wealth,  $W_t := w_t^1 + \ldots + w_t^N$ , is expressed as follows:

$$W_t = \sum_{i=1}^N \langle A_t, x_{t-1}^i \rangle = \sum_{i=1}^N \sum_{k=1}^K A_{t,k} x_{t-1,k}^i = \sum_{k=1}^K A_{t,k} \sum_{i=1}^N x_{t-1,k}^i = \sum_{k=1}^K A_{t,k} V_k,$$

which implies in view of (3.1) that  $W_t > 0$ . Note that in view of  $A_{t,k} = A_k(s_t)$ , so we have  $W_t(s^t) = W(s_t)$ , similarly  $R_k(s^t) = R_k(s_t)$ , as what we gives below.

Since we are now considering i.i.d. states of the world, there will be a minor change (i.e., from the conditional expectation form to the unconditional one) to the sufficient condition for a strategy profile to be admissible, which follows from the assumption that

$$ER_k(s_t) > 0$$

for each k = 1, 2, ..., K. Assume that one of the investors, e.g., investor 1, adopts a fully diversified portfolio rule, prescribing to invest into all the assets in strictly positive proportions  $\lambda_{t,k}^1$ , k = 1, 2, ..., K, then any strategy profile containing this portfolio rule is admissible. Indeed, for t = 0, we have  $p_{0,k} \ge V_k^{-1} \lambda_{0,k}^1 w_0^1 > 0$ and  $x_0^1 = (x_{0,1}^1, ..., x_{0,K}^1) > 0$  (coordinatewise). Assuming that  $x_{t-1}^1 > 0$  and arguing by induction, we obtain

$$w_t^1 = \langle A_t, x_{t-1}^1 \rangle > 0,$$

which in turn yields  $p_t > 0$  and  $x_t^1 > 0$ , as long as  $\lambda_{t,k}^1 > 0$ .

As previously defined, a strategy  $\Lambda^i$  of trader *i* is called a survival strategy (portfolio rule) if for any strategies  $\Lambda^j$  of investors  $j \neq i$ , the market share  $r_t^i := \frac{w_t^i}{W_t}$  of trader *i* is strictly positive and bounded away from zero almost surely:

$$\inf r_t^i > 0 \quad (a.s.).$$

**Explicit Expression for the Survival Strategy**  $\Lambda^*$ . Define the *relative payoffs* by

$$R_k(s) := \frac{A_k(s)V_k}{\sum_{m=1}^{K} A_m(s)V_m}$$
(3.2)

and put

$$\lambda_k^* = ER_k(s_t) \Big[ = \sum_{s \in S} p_s R_k(s) > 0 \Big], \ k = 1, .., K.$$
(3.3)

Consider the constant (independent of t and  $s^t$ ) strategy  $\lambda_t^* = \lambda^* = (\lambda_1^*, ..., \lambda_K^*)$ , where the numbers  $\lambda_k^*$ , k = 1, ..., K, are given by formula (3.3). The portfolio rule specified by (3.3) prescribes to allocate wealth across assets according to the proportions of the expected relative payoffs, which do not depend on tbecause the random elements  $s_t$  are i.i.d.. Strategies of this type are called *fixed*mix (or constant proportions) portfolio rules, prescribing to choose investment proportions at time 0 and keep them fixed over the whole infinite time horizon.

## Same Results for the Existence and Asymptotic Uniqueness of the Survival Strategy $\Lambda^*$ .

#### **Theorem 3.1** The portfolio rule $\Lambda^*$ is a survival strategy.

Note that although  $\Lambda^*$  is a fixed-mix strategy, it guarantees unconditional survival in competition with all, not necessarily fixed-mix strategies. And the following theorem shows that the survival strategy  $\Lambda^* = (\lambda_t^*)$  is essentially unique: any other survival portfolio rule is in a sense asymptotically similar to  $\Lambda^*$ .

**Theorem 3.2** If  $\Lambda = (\lambda_t)$  is a survival strategy, then

$$\sum_{t=0}^{\infty} ||\lambda_t^* - \lambda_t||^2 < \infty \quad (a.s.).$$
(3.4)

For proofs of these results, see Theorem 2.1 and Theorem 2.2 in Section 2.2.

## 3.2 A Classical Game-theoretic Perspective: Nash Equilibrium Properties of Survival Portfolio Rules

#### 3.2.1 The Main Result: Almost Sure Nash Equilibrium

The Notion of Symmetric Almost Sure Nash Equilibrium. Given an admissible strategy profile  $(\lambda^1, ..., \lambda^N)$ , the performance of a strategy  $\lambda^i$  employed by investor *i* will be characterized by the following random variable

$$\xi^{i} := \limsup_{t \to \infty} \frac{1}{t} \ln \frac{w_{t}^{i}}{\sum_{i \neq i} w_{t}^{j}}, \qquad (3.5)$$

generally, taking values in  $[-\infty, +\infty]$ . The expression  $w_t^i / \sum_{j \neq i} w_t^j$  in (3.5) is the relative wealth of player *i* against the group  $\{j : j \neq i\}$  of *i*'s rivals. The random variable  $\xi^i = \xi^i(s^{\infty}; \lambda^1, ..., \lambda^N)$  depends on the strategy profile  $(\lambda^1, ..., \lambda^N)$  and

on the whole history  $s^{\infty} := (s_1, s_2, ...)$  of states of the world from time 1 to  $\infty$ . In the game under consideration,  $\xi^i$  plays the role of the (random) payoff function of player *i*.

We shall say that a strategy  $\overline{\lambda}$  forms a symmetric Nash equilibrium almost surely (a.s.) if

$$\xi^{i}(s^{\infty};\bar{\lambda},...,\bar{\lambda}) \ge \xi^{i}(s^{\infty};\bar{\lambda},...,\lambda,...,\bar{\lambda}) \quad (a.s.)$$
(3.6)

for each *i*, each strategy  $\lambda$  of investor *i* and each set of initial endowments  $w_0^1 > 0, ..., w_0^N > 0$ . The Nash equilibrium is termed *strict* if the inequality in (3.6) is strict for any  $\overline{\lambda}$ .

Recall that we consider only those admissible strategy profiles. Clearly, if all the players use the same strategy  $\bar{\lambda}$ , then the strategy profile  $(\bar{\lambda}, ..., \bar{\lambda})$  is admissible if and only if the vector  $\bar{\lambda}$  is strictly positive (see defining formula (2.8) in Section 2.1).

#### A Unique Symmetric Almost Sure Nash Equilibrium Formed by $\Lambda^*$ .

**Theorem 3.3** The portfolio rule  $\lambda^*$  forms a unique symmetric Nash equilibrium almost surely. If an investor *i* uses any strategy  $\lambda$  distinct from  $\lambda^*$ , then

$$\xi^{i}(s^{\infty};\lambda^{*},...,\lambda,...,\lambda^{*}) < \xi^{i}(s^{\infty};\lambda^{*},...,\lambda^{*}) = 0 \quad (a.s.),$$

$$(3.7)$$

and so the Nash equilibrium formed by strategy  $\lambda^*$  is strict.

The uniqueness of  $\lambda^*$  is understood in the following sense: if  $\lambda$  is a portfolio rule that forms a symmetric Nash equilibrium a.s., then  $\lambda = \lambda^*$ .

**Remark 3.1** We conclude this section with a comment on the definition of the Lyapunov exponent (3.5) in terms of the variables  $w_t^i / \sum_{j \neq i} w_t^j$ . In the Evolutionary Finance literature, relative wealth is typically defined as  $r_t^i = w_t^i / \sum_{j=1}^N w_t^j$  (the market share of investor *i*), which measures the relative wealth of player *i* compared to all investors, not just *i*'s rivals. In many cases, results can be equivalently formulated both in terms of relative wealth—as it is defined in this paper—and market shares. However, this is *not* the case in the context of the present work. The consideration of the Lyapunov exponent

$$\eta^{i} := \limsup_{t \to \infty} \frac{1}{t} \ln \frac{w_{t}^{i}}{\sum_{j=1}^{N} w_{t}^{j}}$$
(3.8)

leads to a trivial notion of a Nash equilibrium. If the payoff functions of the players i = 1, ..., N in the market game are characterized by (3.8), then any completely mixed strategy  $\lambda$  forms a symmetric Nash equilibrium. Indeed,  $\eta^i(\lambda, ..., \lambda) = 0$  because if all the investors adopt the same strategy, their market shares remain constant. On the other hand,  $\eta^i$  is always non-positive, and so  $\eta^i(\lambda^1, ..., \lambda^K) \leq 0 = \eta^i(\lambda, ..., \lambda)$ , which implies that the strategy profile  $(\lambda, ..., \lambda)$  is a Nash equilibrium.

#### 3.2.2 Proof of the Main Result

The proof of Theorem 3.3 is based on an auxiliary result, Lemma 3.1 below. For any  $\lambda = (\lambda_1, ..., \lambda_K) \in \Delta^K$  and  $\kappa \in (0, 1]$  put

$$F(\lambda,\kappa,s) := \frac{\sum_{k=1}^{K} R_k(s) \frac{\lambda_k^*}{\lambda_k^* \kappa + \lambda_k (1-\kappa)}}{\sum_{k=1}^{K} R_k(s) \frac{\lambda_k}{\lambda_k^* \kappa + \lambda_k (1-\kappa)}},$$
(3.9)

where  $\lambda_k^* > 0$  are defined by (3.3). Observe that the nominator A of the fraction  $F(\lambda, \kappa, s) =: A/B$  is not less than

$$c := \min\{\lambda_1^*, \dots, \lambda_K^*\},$$

while the denominator B is not greater than  $c^{-1}$ . Consequently,

$$c^2 \le F(\lambda, \kappa, s) \le +\infty. \tag{3.10}$$

**Lemma 3.1** For any  $\lambda \in \Delta^K$  distinct from  $\lambda^*$ , there exist numbers H > 0 and  $\delta > 0$  such that

$$E\min\{H, \ln F(\lambda, \kappa, s)\} \ge \delta \tag{3.11}$$

for all  $\kappa \in (0, 1]$ .

The proof of Lemma 3.1 is routine but rather lengthy, thus it is relegated to the Appendix A.

**Remark 3.2** A comment about the function  $F(\lambda, \kappa, s)$  is in order. An expression involving F appears in the random dynamical system (3.12) analyzed in the course of the proof of Theorem 3.3. A key step in the proof lies in the application of the law of large numbers to the sequence of martingale differences  $G_t^H - E_{t-1}G_t^H$ , where  $G_t^H$  is defined in terms of the truncation of F by some constant H (see (3.15)). As commonly understood, merely the finiteness of expectations is not enough for the validity of this version of the law of large numbers, but the boundedness of the random variables is fully sufficient – hence the truncation of F by H.

Proof of Theorem 3.3. To prove Theorem 3.3, it suffices to consider the case of two investors (N = 2) because the group of N - 1 investors who all use  $\lambda^*$  is equivalent in terms of wealth dynamics to one, using the strategy  $\lambda^*$  and possessing the aggregate wealth of this group (see Section 2.3, Proof of Theorem 2.1,  $2^{nd}$  step). Now suppose investors 1 and 2 use the strategies  $\lambda^*$  and  $\lambda \neq \lambda^*$ , respectively. Put

$$z_t := w_t^1 / w_t^2 = r_t^1 / r_t^2.$$

Since  $\lambda^* > 0$ , the variable  $z_t$  takes on its values in  $(0, +\infty]$ . The sequence  $(z_t)$  satisfies

$$z_{t+1} = z_t \frac{\sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_k^*}{\lambda_k^* r_t^1 + \lambda_k (1 - r_t^1)}}{\sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_k}{\lambda_k^* r_t^1 + \lambda_k (1 - r_t^1)}}$$
(3.12)

(see Eqs. (2.21)(2.22)). To verify (3.7), it is sufficient to show

$$\liminf_{t \to \infty} t^{-1} \ln z_t > 0 \text{ (a.s.)}. \tag{3.13}$$

Indeed, if inequality (3.13) holds, then

$$\xi^{2}(\lambda^{*},\lambda) = \limsup_{t \to \infty} t^{-1} \ln \frac{w_{t}^{2}}{w_{t}^{1}} = \limsup_{t \to \infty} t^{-1} \ln \frac{r_{t}^{2}}{r_{t}^{1}} = \limsup_{t \to \infty} (-t^{-1} \ln z_{t})$$
$$= -\liminf_{t \to \infty} (t^{-1} \ln z_{t}) < 0 = \xi^{2}(\lambda^{*},\lambda^{*}) \text{ (a.s.)}.$$

We shall now verify (3.13). Put  $G_t = \ln z_t - \ln z_{t-1}$ , then

$$\sum_{t=1}^{T} G_t = \sum_{t=1}^{T} (\ln z_t - \ln z_{t-1}) = \ln z_T - \ln z_0.$$

Thus it is sufficient to show that

$$\liminf_{T \to \infty} T^{-1} \sum_{t=1}^{T} G_t > 0 \text{ (a.s.)}.$$

For any constant H, define  $G_t^H := \min(G_t, H)$ . Since  $G_t^H \leq G_t$  it suffices to prove that

$$\liminf_{T \to \infty} T^{-1} \sum_{t=1}^{T} G_t^H > 0 \text{ (a.s.)}$$
(3.14)

for some H. We have

$$G_{t+1} = \ln z_{t+1} - \ln z_t = \ln \frac{\sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_k^*}{\lambda_k^* r_t^1 + \lambda_k (1 - r_t^1)}}{\sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_k}{\lambda_k^* r_t^1 + \lambda_k (1 - r_t^1)}} = \ln F(\lambda, r_t^1; s_{t+1})$$

and

$$G_{t+1}^{H}(s^{t+1}) = \min\{H, \ln F(\lambda, r_t^1(s^t); s_{t+1})\}.$$
(3.15)

In view of Lemma 3.1, there exist H > 0 and  $\delta > 0$ , such that  $E_t G_{t+1}^H \ge \delta$ , where  $E_t(\cdot) = E(\cdot|s^t)$  denotes the conditional expectation given  $s^t$ . When computing  $E_t G_{t+1}^H$ , we fix  $s^t$  and take the unconditional expectation of  $G_{t+1}^H$  with respect to  $s_{t+1}$ , which is valid since  $s^t$  and  $s_{t+1}$  are independent.

Finally, we have

$$\frac{1}{T}\sum_{t=1}^{T}G_{t}^{H} = \frac{1}{T}\sum_{t=1}^{T}E_{t-1}G_{t}^{H} + \frac{1}{T}\sum_{t=1}^{T}(G_{t}^{H} - E_{t-1}G_{t}^{H}).$$
(3.16)

By virtue of (3.10) and (3.15), for each H the sequence of random variables  $G_t^H(s^t)$  is bounded uniformly in t and  $(s_t)_{t\geq 1}$ . Therefore, we can apply the

strong law of large numbers for martingale differences (see, e.g., [83]), which gives

$$\frac{1}{T} \sum_{t=1}^{T} (G_t^H - E_{t-1} G_t^H) \to 0 \text{ (a.s.)}.$$

Consequently,

$$\liminf_{t \to \infty} T^{-1} \sum_{t=1}^T G_t^H \ge \delta,$$

which proves (3.14).

It remains to show the uniqueness of a symmetric almost sure equilibrium. Assume a strategy  $\lambda \neq \lambda^*$  forms such an equilibrium. Then,

$$0 = \xi^{i}(s^{\infty}; \lambda, ..., \lambda) \ge \xi^{i}(s^{\infty}; \lambda, ..., \lambda, \lambda^{*})$$
(a.s.), (3.17)

where

$$\xi^{i}(s^{\infty};\lambda,...,\lambda,\lambda^{*}) = \limsup_{t \to \infty} t^{-1} \ln \frac{r_{t}^{N}}{1 - r_{t}^{N}}.$$

As we have proved above that

$$\liminf_{t \to \infty} t^{-1} \ln \frac{r_t^N}{1 - r_t^N} > 0 \text{ (a.s.)},$$

which yields the inequality "<" in (3.17). This is a contradiction.  $\blacksquare$ 

## 3.3 Global Stability of Single Survivor

The Main Result: A Theorem for Single Survivor Problem. The market selection process of this model can be described by the evolution of the market shares from the following random dynamical system:

$$r_{t+1}^{i}(s^{t+1}) = \sum_{k=1}^{K} R_{k}(s_{t+1}) \frac{\lambda_{k}^{i} r_{t}^{i}(s^{t})}{\sum_{i=1}^{N} \lambda_{k}^{i} r_{t}^{i}(s^{t})}, \ i = 1, 2, ..., N.$$
(3.18)

As previously defined, an investor i (or the strategy  $\lambda^i$  employed by i) is a single survivor if, for any strictly positive initial vector  $r_0$ , we have  $r_t^i \to 1$  a.s. and  $r_t^j \to 0$  a.s. when  $j \neq i$ . Denote by  $e_i$  the vector in  $\mathbb{R}^N$ , whose coordinates are equal to 0 except for the *i*th coordinate which equals 1. If investor *i* is a single survivor, then all the random paths (3.18) are attracted to  $e_i$  a.s., regardless of the initial state. In this sense,  $e_i$  is a globally stable fixed point of the random dynamical system under consideration. Theorem 3.4 below indicates a strategy  $\lambda^* \in \Delta^K$  such that, under quite general assumptions, the following assertion holds: if investor *i* uses the strategy  $\lambda^*$ , whereas all the other investors adopt distinct strategies  $\lambda^j \neq \lambda^*$ , then investor *i* is a single survivor. And this theorem is valid for any strictly positive vector of initial wealth  $r_0$ , and so yields the global stability of the state  $e_i$ , as mentioned above. **Theorem 3.4** Let investor *i* use the strategy  $\lambda^i = \lambda^*$ , while all the other investors  $j \neq i$  use strategies  $\lambda^j \neq \lambda^*$ . Then investor *i* is the single survivor.

Theorem 3.4 is a direct consequence of Theorem 3.3 (for the proof of Theorem 3.3, see last section). This is valid in the following sense. By virtue of Theorem 3.3, if all the investors except one, say investor *i*, use the strategy  $\lambda^*$  and *i* uses any other strategy  $\lambda$  distinct from  $\lambda^*$ , then the relative wealth  $w_t^i / \sum_{j \neq i} w_t^j$  of *i* tends to zero at the exponential rate  $\xi^i < 0$  (a.s.). In other words, the group of investors using  $\lambda^*$  drives all the others who adopt  $\lambda$  out of the market, which is interpreted as the property of global (holding for all initial states) evolutionary stability of  $\lambda^*$  in Evolutionary Finance. Thus, we arrive at Theorem 3.4: investor *i* uses the strategy  $\lambda^*$  while all the other investors  $j \neq i$ employ strategies  $\lambda^j \neq \lambda^*$ , then investor *i* will eventually drive all the other investors out of the market, accumulate the wealth of the entire market and become the single survivor.

An earlier version of the proof for Theorem 3.4 was provided by Evstigneev et al. [58], Section 3, containing the redundant assumption of completely mixed strategies (strict positivity of all investment proportions). The proof was completed by utilizing an auxiliary result on an inequality (Evstigneev et al. [58], Lemma 1). Importantly, this inequality serves as the foundation for establishing Lemma 3.1 and Lemma 4.1, which are essential for proving Theorem 3.3 and Theorem 4.1, respectively. Consequently, these elementary inequalities, along with their proofs, are collected in Appendix A.

**Remark 3.3** It is important to note that the assumption of non-existence of redundant assets (as we described in Section 3.1) is crucial. This assumption can be fulfilled only if the number of elements in S is not less than K. As a result, it cannot hold in the deterministic case – when S consists of only a single point. In the latter case, it is straightforward to construct an example when the investor i using strategy  $\lambda^*$  is not a single survivor. Indeed, assume i = 1 and the strategy  $\lambda^1 = \lambda^*$  is representable in the form  $\lambda^* = \sum_{j=2}^{I} b_j \lambda^j$ , where  $b = (b_j) \in \Delta^{I-1}, b_j > 0$ . Then, for any  $\kappa \in (0, 1), \lambda^*$  can be represented as  $\sum_{j=1}^{I} a_j \lambda^j$ , where  $a_1 = 1 - \kappa$  and  $a_j = \kappa b_j, j \ge 2$ . In this deterministic case, the values  $R_k(s) = \lambda_k^*$  do not depend on s, and it follows from (3.18) that the vector  $r = (a_j) \in \Delta^I$  is a fixed point of the dynamical system in question for each  $\kappa \in (0, 1)$ . This leads to the failure of (global as well as local) stability of the point  $e_1$ .

## Chapter 4

# Model with Markov Nature

This chapter continues the study shown in previous chapter, for which fixed-mix portfolio rules and i.i.d. distributed states of the world have been considered. The goal for this chapter is to remove these two simplifying assumptions that substantially reduce the scope of the models. That is, (i) basic, rather than fixed-mix strategies, will be considered; (ii) we use a homogeneous discrete-time Markov process, instead of the i.i.d. distributed random variables, to describe the states of the world. With this more general set of assumptions, the model becomes more aligned with real-life behavior, where investors tend to make adjustments in response to unfolding events and newly disclosed information, rather than committing to a single constant strategy throughout the entire duration. Additionally, some serial dependence, at least of a Markov nature, can be postulated to address the limitation that such a complex evolution of many relevant state variables can hardly be captured by a random process with i.i.d. values. Consequently, it is reasonable that this model contributes to a considerably enlarged scope of theory at hand, with enhanced realism and greater potential applicability.

A central result is that in any-complete or incomplete-market for shortlived assets, a trader distributing wealth across available assets in accordance with their relative conditional expected payoffs (the portfolio rule  $\lambda^*$ ) eventually accumulates the entire market wealth, i.e. the single survivor in the market selection process, provided that  $\lambda^*$  is asymptotically distinct from the CAPM rule (prescribing investment in the market portfolio). And the  $\lambda^*$ -trader accumulates the entire market wealth at an exponential rate, if  $\lambda^*$  remains bounded away from the CAPM rule for "sufficiently many" time periods. <sup>1</sup>

### 4.1 The Model

A Summary on Changes in Model Settings. Founded on the basic model framework stated in Section 2.1, a summary of the changes that occur in this

<sup>&</sup>lt;sup>1</sup>This chapter reviews the work by Amir, Evstigneev, Hens, and Schenk-Hoppé [2].

model is as follows:

- Consider a market where  $K \geq 2$  short-lived risky assets (securities) are traded by  $N \geq 2$  investors (traders). Let S be a finite set and  $s_t$ , t = 0, 1, 2, ... (the "state of the world" at time t), a homogeneous Markov chain with transition function  $p(\sigma|s)$ , specifying the conditional probabilities  $P\{s_{t+1} = \sigma | s_t = s\}$ .
- The total amount of units of each asset k = 1, ..., K available in the market is  $V_k(s) > 0$  of  $s \in S$  (exogenously given). Each trader *i* selects a portfolio  $x_t^i = x_t^i(s^t) := (x_{t,1}^i, ..., x_{t,K}^i)$  at each date t = 0, 1, 2, ..., which depends on the history  $s^t = (s_0, ..., s_t)$  of the process  $s_t$  up to date *t* (the argument  $s^t$ is often omitted as long as this does not cause ambiguity). For each date  $t \ge 0$ , each asset k = 1, ..., K, and in every random scenario  $s^t$ , the asset market clears:

$$\sum_{k=1}^{N} x_{t,k}(s^{t}) = V_{k}(s_{t}), \qquad (4.1)$$

where the demand for physical units of each asset k (left-hand side of (4.1)) is equal to its supply (right-hand side of (4.1)). One unit of asset k issued at date t yields payoff  $A_k(s_{t+1}, s_t) \ge 0$  at date t+1, and we assume

$$\sum_{k=1}^{K} A_k(\sigma, s) > 0 \tag{4.2}$$

for all  $\sigma, s \in S$ .

• If trader *i* selects a portfolio  $x_t^i = (x_{t,k}^i)$  at date  $t \ge 0$ , then the market wealth  $w_{t+1}^i$  of the portfolio  $x_t^i$  at date t + 1 can be expressed as

$$w_{t+1}^{i} = \sum_{k=1}^{K} A_k(s_{t+1}, s_t) x_{t,k}^{i}$$

For each trader *i*, the initial wealth is given by a strictly positive number  $w_0^i > 0$ . In view of (4.1), we obtain

$$\sum_{i=1}^{N} w_{t+1}^{i} = \sum_{k=1}^{K} A_{k}(s_{t+1}, s_{t}) V_{k}(s_{t}), \quad t \ge 0.$$
(4.3)

• For each  $t \ge 0$ , each investor i = 1, ..., N selects a vector of investment proportions

$$\lambda_t^i := (\lambda_{t,1}^i, \dots, \lambda_{t,K}^i), \ \lambda_t^i = \lambda_t^i(s^t), \quad t \ge 0,$$
(4.4)

of the budget  $w_t^i$ , such that

$$\lambda_{t,k}^{i} > 0, \ \sum_{k=1}^{K} \lambda_{t,k}^{i} = 1.$$
 (4.5)

Note that (4.5) implies that vectors  $\lambda_t^i$  belong to the relative interior (expressed as  $\Delta_+^K$ ) of the unit simplex  $\Delta^K$ , where

$$\Delta_{+}^{K} := \{ (a_1, ..., a_K) \in \mathbb{R}_{+}^{K} : a_k > 0, \ \sum_{k=1}^{K} a_k = 1 \},\$$

and such strategies are sometimes called *completely mixed*.

A Description for the Asset Market Dynamics. Given each investor *i* has chosen a strategy  $(\lambda_{t,k}^i)$ , the equilibrium price  $p_{t,k} = p_{t,k}(s^t)$  of asset *k* at date  $t \ge 0$  can be determined from the equation

$$p_{t,k} = \frac{1}{V_k(s_t)} \sum_{i=1}^N \lambda_{t,k}^i w_t^i$$

Then, the portfolio  $x_t^i$  of investor *i* at date  $t \ge 0$  will be determined by

$$x_{t,k}^i = \frac{\lambda_{t,k}^i \, w_t^i}{p_{t,k}}.$$

From the last two equations, we find

$$x_{t,k}^{i} = V_{k}(s_{t}) \frac{\lambda_{t,k}^{i} w_{t}^{i}}{\sum_{j=1}^{N} \lambda_{t,k}^{j} w_{t}^{j}}.$$
(4.6)

This leads to the following equation expressing the wealth  $w_{t+1}^i$  of investor *i* at date t+1 through  $w_t^i$ :

$$w_{t+1}^{i} = \sum_{k=1}^{K} A_{k}(s_{t+1}, s_{t}) V_{k}(s_{t}) \frac{\lambda_{t,k}^{i} w_{t}^{i}}{\sum_{j=1}^{N} \lambda_{t,k}^{j} w_{t}^{j}}.$$
(4.7)

Since  $w_0^i > 0$ , by way of induction we obtain  $w_t^i > 0$  for each t (see (4.2) and (4.5)). As a result, we conclude that the evolution of the relative market shares of the investors,  $r_t^i := \frac{w_t^i}{W_t}$  (where  $W_t = \sum_{i=1}^N w_t^i$ ), is governed by the equations

$$r_{t+1}^{i} = \sum_{k=1}^{K} R_{k}(s_{t+1}, s_{t}) \frac{\lambda_{t,k}^{i} r_{t}^{i}}{\sum_{j=1}^{N} \lambda_{t,k}^{j} r_{t}^{j}}, \quad i = 1, ..., N,$$
(4.8)

where

$$R_k(s_{t+1}, s_t) = \frac{A_k(s_{t+1}, s_t) V_k(s_t)}{\sum_{m=1}^{K} A_m(s_{t+1}, s_t) V_m(s_t)}.$$

The numbers  $R_k(s_{t+1}, s_t)$  characterize the relative (normalized) payoff of each asset k, which satisfy  $R_k(s_{t+1}, s_t) \ge 0$  and

$$\sum_{k=1}^{K} R_k(s_{t+1}, s_t) = 1.$$
(4.9)

We are primarily interested in strategies that (i) allow an investor to survive, i.e., to maintain a strictly positive relative market share  $r_t^i$  in the limit, and (ii) enable the investor to dominate the market, i.e., to gather in the limit the entire market wealth. We say that an investor *i* (or the strategy  $\lambda^i = (\lambda_{t,k}^i)$ ) is a single survivor in the market selection process (4.8) if

$$\lim r_t^i = 1 \quad \text{a.s.} \tag{4.10}$$

Condition (4.10) implies  $\lim r_t^j = 0$  a.s. for all  $j \neq i$ , meaning that investor i in the limit accumulates all the market wealth. If the sequence  $r_t^i$  involved in (4.10) converges to 1 at an exponential rate, we shall say that the strategy  $\lambda^i$  dominates the others exponentially.

Remark 4.1 It is an important problem to identify those strategies which lead an investor employing them to be the single survivor. Hens and Schenk-Hoppé [87] and Evstigneev, et al. [58] (the special case shown in Section 3.3) considered this problem within two different settings (local and global, respectively). In the latter model, Theorem 3.4 generalizes the result of Blume and Easley [21], which dealt with the case of Arrow securities  $(S = \{1, 2, ..., K\}, A_k(s) = 0$  if  $s \neq k$  and  $A_k(s) = 1$  if s = k). Furthermore, the strategy (3.3) defined in terms of the expected payoffs in Theorem 3.4 may be regarded as a development of the Kelly rule of "betting one's beliefs" (Kelly [95]). This rule was originally designed in connection with gambling problems, but later on it was successfully applied to portfolio theory (Thorp [155], Aurell et al. [10]). In next section, a version of Theorem 3.4 applicable to the more general model described in current section will be given, i.e., we can define a strategy  $\lambda^*$  of "betting one's beliefs"—a direct analogue of the one considered in Theorem 3.4. As it turns out, we cannot, generally, guarantee  $\lambda^*$  to be the single survivor. Nevertheless, this conclusion does obtain under a natural sufficient condition, having a clear economic meaning. And a necessary and sufficient condition for an investor employing  $\lambda^*$  to be the single survivor dominating the others exponentially will be provided.

## 4.2 The Main Results: Conditions for Single Survivor

Assumptions on the Random Dynamical System. Consider the random dynamical system (4.8) describing the evolution of the relative market shares  $r_t^i(s^t)$  of the investors i = 1, 2, ..., N. Note that if  $r_t = (r_t^i)$  is a strictly positive vector, then, from (4.8), (4.9) and (4.5) we know that  $r_{t+1}$  is a strictly positive vector as well. Thus  $r_t = r_t(s^t)$  is a random process with values in the relative interior  $\Delta^N_+$  of the unit simplex. The initial state  $r_0 = (r_0^1, ..., r_0^N) \in \Delta^N_+$ , from which this process starts, is fixed  $(r_0^i = w_0^i / \sum w_0^j)$ .

The random dynamical system (4.8) will be analyzed under the following assumptions.
(A1). The functions

$$R_k^*(s) := \sum_{\sigma \in S} p(\sigma|s) R_k(\sigma, s), \quad k = 1, 2, ..., K,$$
(4.11)

take on strictly positive values for each  $s \in S$ . (A2). For each  $s \in S$ , the functions  $R_1(\cdot, s), ..., R_K(\cdot, s)$  restricted to the set

$$\Pi(s) = \{ \sigma \in S : \ p(\sigma|s) > 0 \}$$

are linearly independent.

According to (A1), the conditional expectation

$$R_k^*(s) = E[R_k(s_{t+1}, s_t) \mid s_t = s]$$
(4.12)

of the relative payoff  $R_k(s_{t+1}, s_t)$  of each asset k given  $s_t = s$  is strictly positive at each state s. Assumption (A2) indicates the absence of *conditionally* redundant assets. The term "conditionally" refers to the fact that the functions  $R_k(\cdot, s), k = 1, ..., K$ , are linearly independent on the set  $\Pi(s)$ —the support of the conditional distribution  $p(\sigma|s)$ .

In what follows, we will focus exclusively on those investment strategies  $\lambda = (\lambda_{t,k})$  that satisfy the additional assumption outlined below.

(B). The coordinates  $\lambda_{t,k}(s^t)$  of the vectors  $\lambda_t(s^t)$  are bounded away from zero by a strictly positive non-random constant  $\rho$  (which might depend on the strategy  $\lambda$ , but not on k,t and  $s^t$ ), i.e.  $\inf_{i,k,t,s^t} \lambda_{t,k}^i(s^t) > \rho > 0$ .

In (4.5), we included the completely mixed strategies assumption  $\lambda_{t,k} > 0$ . Now that (B) contains the additional requirement of uniform strict positivity of  $\lambda_{t,k}$ .

The Kelly Rule. It is crucial to emphasize the strategy  $\lambda^* = (\lambda^*_{t,k}(s_t))$  defined by the formula

$$\lambda_{t,k}^*(s_t) = R_k^*(s_t), \tag{4.13}$$

where  $R_k^*(s)$  is the conditional expectation of  $R_k(s_{t+1}, s_t)$  given  $s_t = s$  (see (4.11) and (4.12)). As mentioned, this is a direct analog of the Kelly rule of "betting one's beliefs", which takes on the form (3.3) in the i.i.d. distributed  $s_t$  case. Note that  $\lambda_k^*(s_t) = \lambda_{t,k}^*(s_t)$  depends only on the current state  $s_t$  and not on t (explicitly) or the whole history  $s^t$ , which implies, by virtue of (A1) and in view of finiteness of S, that the strategy  $\lambda^*$  satisfies (B).

**The CAPM Rule**. To proceed further, a recursive method of constructing strategies based on (Markovian) decision rules will be given. Suppose one of the investors, say investor 1, has a privilege of making the investment decision at time t with full information about the current market structure  $r_t$  and the actions  $\lambda_t^2(s^t)$ ,  $\lambda_t^3(s^t)$ , ...,  $\lambda_t^N(s^t)$  that have just been undertaken by all the

other investors 2, 3, ..., N. Formally, the decision of investor 1 is specified by a function

$$f_t(r, l^2, ..., l^N), \ r \in \Delta^N_+, \ l^j \in \Delta^K_+ \quad (j = 2, 3, ..., N)$$

taking values in  $\Delta_{+}^{K}$ . Suppose that such functions (i.e. decision rules) are given for all t = 0, 1, 2, ..., and that investors 2, ..., N have chosen some strategies  $\lambda_{t}^{2}, ..., \lambda_{t}^{N}$  (t = 0, 1, 2, ...). Then one can construct a strategy  $\lambda_{t}^{1}(s^{t}), t = 0, 1, 2, ...$ , of investor 1 by using

$$\lambda_t^1(s^t) = f_t(r_t, \lambda_t^2, \dots, \lambda_t^N), \qquad (4.14)$$

where  $r_t = r_t(s^t)$  and  $\lambda_t^j = \lambda_t^j(s^t), j = 2, ..., N$ .

Let us consider a particular decision rule  $f = (f_1, ..., f_K)$  (which does not explicitly depend on t) defined by

$$f(r, l^2, ..., l^N) = \sum_{j=2}^N \frac{r^j}{1 - r^1} \, l^j, \qquad (4.15)$$

where  $r = (r^1, ..., r^N) \in \Delta_+^N$ ,  $l^j = (l_1^j, ..., l_K^j) \in \Delta_+^K$ , thus the vector  $f = (f_1, ..., f_K)$  belongs to  $\Delta_+^K$ . Note that the vector f is a convex combination of the vectors  $l^2, ..., l^N$  with weights  $r^j(1-r^1)^{-1}$ , which implies: if the coordinates  $l_k^j$  of the vectors  $l^j$  are bounded away from 0 by a constant  $\rho > 0$ , then the coordinates  $f_k$  of f are bounded away from 0 by the same constant. Consequently, if the strategies  $\lambda_t^2, ..., \lambda_t^N$  satisfy (B), the strategy (4.14) satisfies (B) as well. In what follows, we will use the notation  $f = (f_k)$  for the particular decision rule described in (4.15).

The decision rule (4.15) exhibits several remarkable properties. First of all, observe the following. Suppose investor 1 uses the strategy  $\lambda_t^1(s^t)$  defined by (4.14) in terms of the decision rule (4.15). Then we obtain

$$\lambda_{t,k}^{1} = \sum_{j=1}^{N} \lambda_{t,k}^{j} r_{t}^{j}, \qquad (4.16)$$

which, in view of (4.8) and (4.9), yields

$$r_{t+1}^1 = r_t^1.$$

Hence, if investor 1 employs the strategy generated by the decision rule (4.15), then, regardless of what strategies are adopted by the others, the relative market share of investor 1 remains constant over time. This observation results in the following conclusion. If one of the other traders 2, ..., N employs  $\lambda^*$ , then this trader cannot be a single survivor, as long as trader 1 uses (4.14)–(4.15), and thus maintains a constant strictly positive market share  $r_t^1 = r_0^1$  for all t.

Further, the portfolio of investor 1, who employs the strategy  $\lambda_t^1$  defined in terms of (4.14)–(4.15), is given by

$$x_{t,k}^{1} = V_{k} \frac{\lambda_{t,k}^{1} w_{t}^{1}}{\sum_{j=1}^{N} \lambda_{t,k}^{j} w_{t}^{j}} = V_{k} \frac{\lambda_{t,k}^{1} r_{t}^{1}}{\sum_{j=1}^{N} \lambda_{t,k}^{j} r_{t}^{j}} = V_{k} r_{t}^{1},$$

for each k = 1, ..., K (see (4.6) and (4.16)). Hence, the vector  $x_t^1 = (x_{t,1}^1, ..., x_{t,K}^1)$  turns out to be proportional to the market portfolio, represented by the vector  $(V_1, ..., V_k)$ , whose components indicate the amount of assets k = 1, ..., K currently traded in the market. According to the well-known Tobin mutual fund theorem (Magill and Quinzii [111], Proposition 16.15), portfolios with this structure arise from the mean-variance optimization in the Capital Asset Pricing Model (CAPM). Therefore, it is natural to refer to the decision rule (4.15) as the CAPM decision rule and the strategy generated by it as the CAPM strategy.

Three Theorems on Single Survivor Problem. A sufficient condition for the strategy (4.13) to be a single survivor is given in Theorem 4.1 below, based on the dynamical system (4.8) and assuming that all the investors  $i \in \{1, 2, ..., N\}$ employ some strategies  $\lambda^i = (\lambda_t^i)$  satisfying (B). We define

$$\boldsymbol{\zeta}_t = (\zeta_{t,1},...,\zeta_{t,K}) = f(r_t,\lambda_t^2,...,\lambda_t^N),$$

where f is the CAPM decision rule (4.15). The symbol  $|\cdot|$  denotes the sum of the absolute values of the coordinates of a finite-dimensional vector.

**Theorem 4.1** Suppose investor 1 uses the strategy  $\lambda^1 = \lambda^*$  defined by (4.13). Let the following condition be fulfilled: (C). With probability 1, we have

$$\liminf_{t \to \infty} |\lambda^*(s_t) - \zeta_t| > 0. \tag{4.17}$$

Then investor 1 is a single survivor, and, moreover,

$$\liminf_{t \to \infty} \frac{1}{t} \ln \frac{r_t^1}{1 - r_t^1} > 0 \tag{4.18}$$

almost surely.

Property (4.18) indicates that the relative market share of investor 1 tends to one at an exponential rate, whereas the relative market shares of all the other investors vanish at such rates, and so the strategy  $\lambda^*$  dominates the others exponentially.

Assumption (C) can be restated as follows: there exists a strictly positive random variable  $\kappa$  such that,

$$|\lambda^*(s_t) - \zeta_t(s^t)| \ge \kappa \tag{4.19}$$

a.s. for all t large enough. Inequality (4.19) requires that the actions  $\lambda^*(s_t)$  prescribed by the strategy  $\lambda^*$  should differ by not less than  $\kappa > 0$  from the actions

$$\zeta_t(s^t) = (\zeta_{t,1}(s^t), ..., \zeta_{t,K}(s^t)), \ \ \zeta_{t,k}(s^t) = \sum_{j=2}^N \frac{r_t^j(s^t)}{1 - r_t^1(s^t)} \, \lambda_{t,k}^j(s^t),$$

prescribed by the CAPM decision rule. In this context, we do not assume that any of the market participants indeed employs the CAPM rule; we need it only as an indicator, a proper deviation of which from  $\lambda^*$  guarantees  $\lambda^*$  to be a single survivor.

In concrete instances, verifying (C) directly might not be easy. Thus we introduce another hypothesis, (C1), which is stronger than (C) but can be more conveniently checked in various examples.

(C1). There exists a strictly positive random variable  $\kappa$  such that, with probability 1, the distance between the vector  $\lambda^*(s_t) \in \mathbb{R}^K$  and the convex hull of the vectors  $\lambda_t^2(s^t), ..., \lambda_t^N(s^t) \in \mathbb{R}^K$  is not less than  $\kappa$  for all t large enough.

Clearly (C1) implies (C) because  $\zeta_t = f(r_t, \lambda_t^2, ..., \lambda_t^N)$  is a convex combination of  $\lambda_t^2, ..., \lambda_t^N$ . Assumption (C), which suffices for investor *i* to be a single survivor, turns out to be close to a necessary one. The theorem below provides a version of (C) that is both necessary and sufficient for the conclusion of Theorem 4.1 to hold.

**Theorem 4.2** Investor 1 using the strategy (4.13) is a single survivor in the market selection process, and, moreover, dominates the others exponentially, if and only if the following condition is fulfilled:

(C2). There exists a random variable  $\kappa > 0$  such that

$$\liminf_{T \to \infty} \frac{1}{T} \# \{ t \in \{0, ..., T\} : |\lambda^*(s_t) - \zeta_t(s^t)| \ge \kappa \} > 0$$
(4.20)

with probability 1.

The symbol # represents the number of elements in a finite set. Observe that (C2) follows from (C). Indeed, hypothesis (C) is equivalent to the existence of a random variable  $\kappa$  for which, inequality (4.19) is fulfilled a.s. for all t large enough. In this case, the limit in (4.20) is equal to 1, and may be regarded as a *density* (in the set of natural numbers) of those natural numbers t for which inequality (4.19) holds. Hypothesis (C2) only requires this density to be strictly positive, whereas (C) asserts that (4.19) should hold from some t on.

Recall Theorem 4.1, from which it follows immediately that if the relation

$$\liminf_{t \to \infty} \frac{1}{t} \ln \frac{r_t^1}{1 - r_t^1} \le 0$$
(4.21)

holds with positive probability, then, there exists a (random) sequence  $t_k$  such that

$$|\lambda^*(s_{t_k}) - \zeta_{t_k}(s^{t_k})| \to 0 \tag{4.22}$$

with positive probability. By appropriately strengthening (4.21), we can make a stronger statement about convergence in (4.22), as provided below.

**Theorem 4.3** Let the following condition be fulfilled: (D1). There exists a random variable  $0 < \gamma < 1$  such that  $E \ln \gamma > -\infty$  and

$$r_t^1 < 1 - \gamma$$

a.s. for all t.

Then we have

$$|\lambda^*(s_t) - \zeta_t| \to 0 \quad a.s.$$

We will actually prove Theorem 4.3 under a weaker condition:

(D2). The expectations

$$E[\ln(1-r_t^1)]$$

do not converge to  $-\infty$ .

Clearly (D1) is stronger than both (D2) and (4.21), but (D2) does not necessarily imply (4.21). Assumption (D1) holds, e.g., if one of the investors i = 2, ..., N employs the CAPM strategy (and so this investor's relative market share keeps constant). Then, as Theorem 4.3 shows, the difference between the budget shares of investor 1 prescribed by the strategy  $\lambda^*$  and the budget shares prescribed by the CAPM decision rule converges to 0 almost surely.

### 4.3 Proofs of the Main Results

Theorem 4.1 is a direct consequence of Theorem 4.2.

Proof of Theorem 4.2. By virtue of (4.8), we write

$$\frac{1-r_{t+1}^1}{1-r_t^1} = \frac{\sum_{i=2}^N r_{t+1}^i}{1-r_t^1} = \sum_{k=1}^K R_k(s_{t+1},s_t) \frac{(1-r_t^1)^{-1} \sum_{i=2}^N \lambda_{t,k}^i r_t^i}{q_{t,k}}$$
$$= \sum_{k=1}^K R_k(s_{t+1},s_t) \frac{\zeta_{t,k}}{q_{t,k}},$$

where

$$q_{t,k} = \sum_{m=1}^{N} \lambda_{t,k}^{m} r_{t}^{m} = \lambda_{t,k}^{1} r_{t}^{1} + (1 - r_{t}^{1}) \frac{\sum_{i=2}^{N} \lambda_{t,k}^{i} r_{t}^{i}}{1 - r_{t}^{1}} = \lambda_{t,k}^{1} r_{t}^{1} + \zeta_{t,k} (1 - r_{t}^{1}).$$

Consequently,

$$1 - r_{t+1}^{1} = \sum_{k=1}^{K} R_{k}(s_{t+1}, s_{t}) \frac{\zeta_{t,k}(1 - r_{t}^{1})}{\lambda_{t,k}^{1} r_{t}^{1} + \zeta_{t,k}(1 - r_{t}^{1})},$$
(4.23)

and

$$r_{t+1}^{1} = \sum_{k=1}^{K} R_{k}(s_{t+1}, s_{t}) \frac{\lambda_{t,k}^{1} r_{t}^{1}}{\lambda_{t,k}^{1} r_{t}^{1} + \zeta_{t,k}(1 - r_{t}^{1})}.$$
(4.24)

For each t = 1, 2, ..., consider

$$D_t = \ln \frac{r_t^1 (r_{t-1}^1)^{-1}}{(1 - r_t^1)(1 - r_{t-1}^1)^{-1}}$$

We have

$$D_1 + \dots + D_T = \ln \frac{r_T^1}{(1 - r_T^1)} - \ln \frac{r_0^1}{(1 - r_0^1)}.$$
(4.25)

Therefore, (4.18) holds if and only if

$$\liminf_{T \to \infty} \frac{1}{T} \left( D_1 + \dots + D_T \right) > 0 \quad \text{a.s.}$$

According to hypothesis (B), for each set of strategies  $(\lambda_{t,k}^i)$ , i = 1, ..., N, there exists a constant H > 1, such that  $(\inf_{i,k,t} \lambda_{t,k}^i)^{-1} \leq H$ . For this H, we have

$$H^{-1} \le \frac{r_{t+1}^i}{r_t^i} \le H, \quad i = 1, ..., N,$$

which implies

$$H^{-1} \le \frac{1 - r_{t+1}^1}{1 - r_t^1} \le H$$

because  $1 - r_t^1 = \sum_{m=2}^N r_t^m$ . As a result, the random variables  $D_t$  are uniformly bounded.

We have the following identity

$$\frac{1}{T} \sum_{t=1}^{T} D_t = \frac{1}{T} \sum_{t=1}^{T} E(D_t | s^{t-1}) + \frac{1}{T} \sum_{t=1}^{T} [D_t - E(D_t | s^{t-1})].$$

In view of uniform boundedness of  $D_t$ , we can apply to the process of martingale differences  $B_t := D_t - E(D_t|s^{t-1})$  the strong law of large numbers (Hall and Heyde [83], Theorem 2.19), which yields  $T^{-1}(B_1 + \ldots + B_T) \to 0$  with probability 1. Hence, we obtain

$$\liminf_{T \to \infty} \frac{1}{T} \sum_{t=1}^{T} D_t = \liminf_{T \to \infty} \frac{1}{T} \sum_{t=1}^{T} E(D_t | s^{t-1}), \tag{4.26}$$

and so (4.18) is equivalent to

$$\liminf_{T \to \infty} \frac{1}{T} \sum_{t=1}^{T} E(D_t | s^{t-1}) > 0 \quad \text{a.s.}$$
(4.27)

By virtue of (4.23), (4.24), we have

$$E[D_t|s^{t-1}] = E[\ln\frac{r_t^1(r_{t-1}^1)^{-1}}{(1-r_t^1)(1-r_{t-1}^1)^{-1}}|s^{t-1}]$$
  
$$= \sum_{\sigma \in S} p(\sigma|s_{t-1}) \ln\frac{\sum_k^{R_k(\sigma, s_{t-1})} \frac{\lambda_{t-1,k}^1}{\lambda_{t-1,k}^1r_{t-1}^1 + \zeta_{t-1,k}(1-r_{t-1}^1)}}{\sum_k^{R_k(\sigma, s_{t-1})} \frac{\zeta_{t-1,k}}{\lambda_{t-1,k}^1r_{t-1}^1 + \zeta_{t-1,k}(1-r_{t-1}^1)}},$$
  
(4.28)

where

$$\zeta_{t-1,k} = \zeta_{t-1,k}(s^{t-1}) = \frac{\sum_{i=2}^{N} \lambda_{t-1,k}^{i} r_{t-1}^{i}}{1 - r_{t-1}^{1}},$$
(4.29)

$$\lambda_{t-1,k}^{i} = \lambda_{t-1,k}^{i}(s^{t-1}), \ r_{t-1}^{i} = r_{t-1}^{i}(s^{t-1}), \tag{4.30}$$

$$\Lambda_{t-1,k} - \Lambda_{t-1,k}(s_{t-1}) - \Pi_k(s_{t-1}).$$

Let us use Lemma 4.1 below (for the proof, see Appendix A) to estimate the expression in (4.28).

**Lemma 4.1** There exists a constant  $L_{\rho}$  and a function  $\delta_{\rho}(\gamma) \geq 0$  of  $\gamma \in [0, \infty)$  satisfying the following conditions:

- 1. The function  $\delta(\cdot)$  is non-decreasing, and  $\delta_{\rho}(\gamma) > 0$  for all  $\gamma > 0$ .
- 2. For any  $s \in S$ ,  $\kappa \in [0,1]$  and  $\mu = (\mu_k) \in \Delta_{\rho}^K$ , we have

$$L_{\rho}|R^{*}(s) - \mu| \ge \Phi(s, \kappa, \mu) \ge \delta_{\rho}(|R^{*}(s) - \mu|).$$
(4.31)

In light of this lemma, we get

$$E(D_t|s^{t-1}) \ge \delta_{\rho}(|R^*(s_{t-1}) - \zeta_{t-1}(s^{t-1})|), \tag{4.32}$$

where  $\rho$  is the strictly positive constant that bounds the coordinates of  $\lambda_t^i$  away from 0. We use  $N(T) = N(T, s^T)$  to denote the set of those  $t \in [0, T]$  for which  $|R^*(s_t) - \zeta_t(s^t)| \ge \kappa$ . We have

$$\begin{split} \liminf_{T \to \infty} \frac{1}{T} \sum_{t=1}^{T} E(D_t | s^{t-1}) &\geq \liminf_{T \to \infty} \frac{1}{T} \sum_{t=1}^{T} \delta_{\rho}(|R^*(s_{t-1}) - \zeta_{t-1}(s^{t-1})|) \\ &\geq \liminf_{T \to \infty} \frac{1}{T} \sum_{t=0}^{T-1} \delta_{\rho}(|R^*(s_t) - \zeta_t(s^t)|) \geq \liminf_{T \to \infty} \frac{1}{T} \sum_{t \in N(T-1)} \delta_{\rho}(|R^*(s_t) - \zeta_t(s^t)|) \\ &\geq \delta_{\rho}(\kappa) \cdot \liminf_{T \to \infty} \frac{1}{T} \#\{N(T-1)\} > 0, \end{split}$$

where the last inequality follows from (C2). Therefore, we have established (4.27), which is equivalent to (4.18).

Now, assume (4.18), and so (4.27), hold. According to Lemma 4.1, we find

$$E(D_t|s^{t-1}) \le L_{\rho} \cdot |R^*(s_{t-1}) - \zeta_{t-1}(s^{t-1})|,$$

and hence (4.27) gives

m

$$\liminf_{T \to \infty} \frac{1}{T} \sum_{t=1}^{T} d_t > 0 \quad \text{a.s.},$$
(4.33)

where  $d_t = |R^*(s_{t-1}) - \zeta_{t-1}(s^{t-1})|.$ 

We use  $\bar{\kappa}$  to denote the strictly positive random variable that equals the limit in (4.33) a.s., and set  $\kappa = \bar{\kappa}/2$ . We claim that

$$\liminf \frac{1}{T} \# \{ t \in \{1, ..., T\} : d_t \ge \kappa \} > 0 \quad \text{a.s.},$$
(4.34)

which is equivalent to (C2). Indeed, assume the contrary (for the sake of contradiction). In that case, there is a sequence  $(T_k)_k$  (which depends on  $(s_t)$ ), such that

$$\frac{1}{T_k} \# \{ t \in \{1, ..., T_k\} : d_t \ge \kappa \} \to 0$$
(4.35)

with positive probability. For each k, denote by  $M_k$  (resp.  $N_k$ ) the set of those  $t \in \{1, ..., T_k\}$  for which  $d_t \ge \kappa$  (resp.  $d_t < \kappa$ ). Then, it yields, for all events where (4.35) holds,

$$\frac{1}{T_k} \sum_{t=1}^{T_k} d_t = \frac{1}{T_k} \sum_{t \in M_k} d_t + \frac{1}{T_k} \sum_{t \in N_k} d_t \le 2 \cdot \frac{1}{T_k} \#(M_k) + \kappa,$$
(4.36)

since  $d_t \leq 2$ . According to (4.35), we have  $(T_k)^{-1} \cdot \#(M_k) \to 0$ . Therefore, for all events where (4.35) holds,

$$\liminf_{T \to \infty} \frac{1}{T} \sum_{t=1}^{T} d_t \le \liminf_{k \to \infty} \frac{1}{T_k} \sum_{t=1}^{T_k} d_t \le \kappa < \bar{\kappa},$$

which contradicts the definition of  $\bar{\kappa}$ .

Proof of Theorem 4.3. Consider the nonnegative random variables  $v_t = \delta_{\rho}(|R^*(s_{t-1}) - \zeta_{t-1}(s^{t-1})|)$ . By using (4.32), we have  $Ev_t \leq E[E(D_t|s^{t-1})] = ED_t$ , which gives, by virtue of (4.25),

$$\sum_{t=1}^{T} Ev_t \le E \ln \frac{r_T^1}{1 - r_T^1} + C \le -E \ln(1 - r_T^1) + C,$$

where C is some constant. From the hypothesis (D2), the expectations  $-E \ln(1-r_T^1)$  do not converge to  $+\infty$ . Thus the non-negative sums  $Ev_1 + \ldots + Ev_T$  are bounded by some constant  $C_1$ . Consequently, in light of the Fatou lemma,

$$E \lim_{T \to \infty} \sum_{t=0}^{T} v_t = E \liminf_{T \to \infty} \sum_{t=0}^{T} v_t \le \liminf_{T \to \infty} \sum_{t=0}^{T} E v_t \le C_1.$$

Therefore, we get  $\sum_{t=0}^{\infty} v_t < \infty$  a.s., thus  $v_t \to 0$  a.s., and so  $|R^*(s_{t-1}) - \zeta_{t-1}(s^{t-1})| \to 0$  a.s.

# Chapter 5

# Model with Risk-free Asset

As shown in Chapters 2–4, the majority of studies in this field up to now have been concentrated on the case that all the assets are purely risky, which means that asset prices are random and endogenous, deternimed by the strategy profiles of all the investors as a result of the asset market dynamics in short-run equilibrium. In this chapter, we examine a model that incorporates a risk-free asset, in the sense that its price is exogenous (which can be normalized to 1 and serves as a numeraire). A key feature of the model is that asset payoffs are assumed to be scalable, depending linearly on the total cash in all participants' bank accounts (no currency outside banks), which makes the numeraire effectively invariant under redenominations. If the asset payoffs and prices grow in scale, the total cash required to function the system has to increase accordingly. The primary aim of the study is also to identify whether there exists an (asymptotically unique) investment portfolio rule that allows an investor to "survive" in the long-run market selection process.<sup>1</sup>

### 5.1 The Model

A Summary on Changes in Model Settings. Based on the general model framework stated in Section 2.1, a summary of the changes that occur in this model is as follows:

• Consider a market where  $K \geq 2$  short-lived risky assets (securities) k = 1, 2, ..., K and a risk-free asset k = 0 are traded by  $N \geq 2$  investors (traders). We use (K+1)-dimensional vectors  $x_t^i := (x_{t,0}^i, x_{t,1}^i, ..., x_{t,K}^i) \in \mathbb{R}^{K+1}_+$  (where  $x_{t,k}^i \geq 0, \ k = 0, 1, ..., K$ , i.e. short sales are ruled out) and  $p_t = (p_{t,0}, ..., p_{t,K}) \in \mathbb{R}^{K+1}_+$  to denote the portfolio of investor i and asset prices at date  $t \geq 0$ , respectively. Accordingly, the market value of investor i's portfolio at date t is given by  $\langle p_t, x_t^i \rangle := \sum_{k=0}^{K} p_{t,k} x_{t,k}^i$ . These

<sup>&</sup>lt;sup>1</sup>This chapter discusses the study conducted by Belkov, Evstigneev, and Hens [13].

vectors  $x_t^i$  and  $p_t$  depend on  $s^t$ :

$$p_t = p_t(s^t), \ x_t^i = x_t^i(s^t)$$

and all functions of  $s^t$  are assumed to be measurable. Note that the prices  $p_{t,1}, ..., p_{t,K}$  of the risky assets k = 1, 2, ..., K are determined endogenously through asset market dynamics by short-run equilibrium over each time period [t-1,t). The price  $p_{t,0}$  of the risk-free asset k = 0, which can be interpreted as cash, is exogenous and normalized to one:  $p_{t,0} = 1$ . That explains why asset k = 0 is considered risk-free. However, the interest rate for cash, which we introduce below, might be random and time dependent.

• Suppose that we have  $V_{0,k} > 0$  and  $V_{t,k}(s^t) > 0$  total units of each risky asset k = 1, 2, ..., K available in the market at date 0 and each subsequent period t = 1, 2, ..., respectively. However, there are no exogenously set limits for the total amount of the risk-free asset k = 0 (cash), which will be used as a numeraire to express the market values of all the assets. Risk-free holdings  $x_{t,0}^i$  in the portfolios  $x_t^i = (x_{t,0}^i, x_{t,1}^i, ..., x_{t,K}^i)$  of investors i = 1, 2, ..., N can be regarded as balances in their bank accounts, with some given interest rate  $\beta_t = \beta_t(s^t) \ge 0$ . It is assumed that all the money in the system is deposited in the bank accounts of the market participants, so that

$$\bar{w}_t := \sum_{i=1}^N x_{t,0}^i \tag{5.1}$$

represents the aggregate amount of money at date t.

• The payoffs  $A_{t+1,k}$  of the risky assets k = 1, 2, ..., K depend linearly on the ("narrow") money supply  $\bar{w}_t$  at date t and on the history of states of the world  $s^{t+1}$ :

$$A_{t+1,k} = a_{t+1,k}(s^{t+1})\bar{w}_t . (5.2)$$

The linear dependence of  $A_{t+1,k}$  on  $\bar{w}_t$  postulated in equation (5.2) reflects the idea of scalability of the asset payoffs and the relative nature of the numeraire value. The functions  $a_{t,k}(s^t)$ , k = 1, 2, ..., K, satisfy

$$\sum_{k=1}^{K} a_{t,k}(s^{t}) > 0 \text{ for all } t, s^{t},$$
(5.3)

which means that in each random scenario, there is at least one of the assets k = 1, 2, ..., K yields a strictly positive payoff (as long as  $\bar{w}_t > 0$ ).

• Investors have initial endowments  $w_0^i > 0$  (i = 1, 2, ..., N) as their budgets at date 0. For date  $t \ge 1$ , investor *i*'s budget is

$$w_t^i := \langle A_t, x_{t-1}^i \rangle = \sum_{k=0}^K A_{t,k} x_{t-1,k}^i , \qquad (5.4)$$

where

$$A_t := (A_{t,0}, \dots, A_{t,K}) \text{ and } A_{t,0} := 1 + \beta_t.$$
 (5.5)

The budget (5.4) is formed by the payoffs

$$A_{t,k}x_{t-1,k}^i = a_{t,k}\bar{w}_{t-1}x_{t-1,k}^i \tag{5.6}$$

of the risky assets k = 1, ..., K held in investor *i*'s previous portfolio  $x_{t-1}^i$ , along with the amount of money  $[1 + \beta_t(s^t)]x_{t-1,0}^i$  in this investor's bank account at the beginning of period [t, t + 1). The budget  $\langle A_t, x_{t-1}^i \rangle$  is re-invested in the assets available at time *t*, which subsequently give the payoffs  $A_{t+1,k}$ , k = 1, ..., K, and interest with the rate  $\beta_{t+1}$  at time t + 1.

• At each  $t \ge 0$ , each investor i = 1, 2, ..., N selects a vector of investment proportions  $\lambda_t^i := (\lambda_{t,0}^i, ..., \lambda_{t,K}^i)$ , which belongs to the unit simplex

$$\Delta^{K+1} := \{ (b_0, ..., b_K) \in \mathbb{R}^{K+1}_+ : b_0 + ... + b_K = 1 \}.$$

A Description for the Asset Market Dynamics. The asset market dynamics of this model, analogous to those presented in Section 2.1, Eqs. (2.2)–(2.5), are outlined below.

1. For date t = 0:

The equilibrium prices  $p_{0,k}$  of the risky assets k = 1, 2, ..., K are determined from

$$p_{0,k}V_{0,k} = \sum_{i=1}^{N} \lambda_{0,k}^{i} w_{0}^{i}, \ k = 1, 2, ..., K.$$
(5.7)

The investors' portfolios  $x_0^i=(x_{0,1}^i,...,x_{0,K}^i),\ i=1,2,...,N$  are determined from

$$x_{0,k}^{i} = \frac{\lambda_{0,k}^{i} w_{0}^{i}}{p_{0,k}}, \ k = 1, 2, ..., K, \ i = 1, ..., N.$$
(5.8)

The 0th position  $x_{0,0}^i$  of portfolio  $x_0^i$  (the amount of money deposited into the bank account by investor *i*), is given by

.

$$x_{0,0}^{i} = \frac{\lambda_{0,0}^{i} w_{0}^{i}}{p_{0,0}} = \lambda_{0,0}^{i} w_{0}^{i}, \qquad (5.9)$$

where  $p_{0,0} = 1$ .

2. For date  $t \ge 1$ :

The equilibrium prices  $p_{t,k}$  of the risky assets k = 1, 2, ..., K are determined from

$$p_{t,k}V_{t,k} = \sum_{i=1}^{N} \lambda_{t,k}^{i} w_{t}^{i}, \ k = 1, ..., K.$$
(5.10)

The investors' portfolios  $x_t^i = (x_{t,1}^i, ..., x_{t,K}^i), i = 1, 2, ..., N$  are determined from

$$x_{t,k}^{i} = \frac{\lambda_{t,k}^{i} w_{t}^{i}}{p_{t,k}}, \ k = 1, ..., K,$$
(5.11)

with the 0th position  $x_{t,0}^i$  of portfolio  $x_t^i$ :

$$x_{t,0}^{i} = \frac{\lambda_{t,0}^{i} w_{t}^{i}}{p_{t,0}} = \lambda_{t,0}^{i} w_{t}^{i}.$$
(5.12)

**Remark 5.1** A comment regarding the admissibility of strategy profiles in this model is given. Portfolio positions  $x_{t,k}^i$  are well-defined by (5.11) only if for each  $t \ge 0$ , the prices  $p_{t,k}$ , k = 1, ..., K are strictly positive, or equivalently, if the total demand for each risky asset k = 1, ..., K is strictly positive:

$$\sum_{i=1}^{N} \lambda_{t,k}^{i} w_{t}^{i} > 0.$$
(5.13)

It is crucial that

$$\bar{w}_t = \sum_{i=1}^N x_{t,0}^i = \sum_{i=1}^N \lambda_{t,0}^i w_t^i > 0$$
(5.14)

for each  $t \ge 0$ , i.e., the total demand  $\sum_{i=1}^{N} \lambda_{t,0}^{i} w_{t}^{i}$  for the risk-free asset k = 0 is strictly positive as well. In fact, if  $\bar{w}_{t} = \sum_{i=1}^{N} x_{t,0}^{i} = 0$ , then it gives  $x_{t,0}^{i} = 0$  for every i = 1, ..., N and  $A_{t+1,k} = a_{t+1,k} \bar{w}_{t} = 0$  for all k = 1, ..., K. Consequently,

$$w_{t+1}^i = \langle A_{t+1}, x_t^i \rangle = \sum_{k=0}^K A_{t+1,k} x_{t,k}^i = (1+\beta_{t+1}) x_{t,0}^i + \sum_{k=1}^K a_{t+1,k} \bar{w}_t x_{t,k}^i = 0$$

due to  $x_{t,0}^i = \bar{w}_t = 0$ . Therefore, if  $\bar{w}_t = 0$  for some date t, then at date t+1 the wealth  $w_{t+1}^i$  of each trader i = 1, ..., N vanishes, and so the market collapses.

Those strategy profiles for which conditions (5.13) and (5.14) hold for all k = 0, 1, ..., K and  $t \ge 0$ , will be called *admissible*. In what follows, we will consider only strategy profiles that meet this assumption, as it guarantees the random dynamical system under consideration is well-defined with  $\bar{w}_t > 0$  and  $p_{t,k} > 0$  for all  $t \ge 0$  and k = 1, ..., K.

A sufficient condition for a strategy profile to be admissible, following from hypothesis (5.20) below, is provided, which will hold for all the strategy profiles we consider. Suppose that one of the traders, e.g., trader 1, follows a fully diversified portfolio rule, which prescribes  $\lambda_{t,k}^1 > 0$  for all assets k = 0, 1, ..., K and all  $t \ge 0$ . By induction, we get

$$w_t^1 = \sum_{k=0}^{K} A_{t,k} x_{t-1,k}^1 \ge A_{t,0} x_{t-1,0}^1 = (1+\beta_t) \lambda_{t-1,0}^1 w_{t-1}^1 > 0,$$

which, in turn, gives

$$\bar{w}_t = \sum_{i=1}^N \lambda_{t,0}^i w_t^i \ge \lambda_{t,0}^1 w_t^1 > 0$$

and

$$\sum_{i=1}^{N} \lambda_{t,k}^{i} w_{t}^{i} \ge \lambda_{t,k}^{1} w_{t}^{1} > 0 \text{ for } k = 1, ..., K.$$

This implies that any strategy profile containing a fully diversified portfolio rule is admissible.

Assuming the admissibility of strategy profiles, and summing up Eqs. (5.11) over i = 1, ..., N, we get

$$\sum_{i=1}^{N} x_{t,k}^{i} = \frac{\sum_{i=1}^{N} \lambda_{t,k}^{i} w_{t}^{i}}{p_{t,k}} = \frac{p_{t,k} V_{t,k}}{p_{t,k}} = V_{t,k}, \ k = 1, ..., K,$$
(5.15)

indicating that the market clears for each risky asset k = 1, ..., K and each date  $t \ge 0$ .

## 5.2 The Main Results: Existence and Asymptotic Uniqueness of Survival Portfolio Rules

Consider an admissible strategy profile  $(\Lambda^1, ..., \Lambda^N)$  of the investors and the corresponding path of the random dynamical system generated by it. We are still primarily interested in the long-run behavior of the relative wealth (or the market shares),  $r_t^i := w_t^i/W_t$  of the investors, i.e., in the asymptotic properties of the series of vector  $r_t = (r_t^1, ..., r_t^N)$  as  $t \to \infty$ , where  $w_t^i = \langle A_t, x_{t-1}^i \rangle$ ,  $W_t = \sum_{i=1}^N w_t^i$ . Note that  $r_t^i$ , i = 1, 2, ..., N are well-defined since  $W_t \ge \bar{w}_t > 0$ , as long as the strategy profile is admissible. Recall that the strategy  $\Lambda^i$  (or investor *i* using it) survives with probability one if  $\inf_{t\ge 0} r_t^i > 0$  a.s., and this portfolio rule is called a survival strategy if investor *i* using it survives with portfolio rules  $\Lambda^j$ ,  $j \ne i$ , are used by the other investors.

**Explicit Expression for the Survival Strategy**  $\Lambda^*$ . To formulate the main results, we define

$$R_{t+1,k} = \frac{a_{t+1,k}V_{t,k}}{\sum_{l=0}^{K} a_{t+1,l}V_{t,l}}, \ k = 0, ..., K,$$
(5.16)

where

$$a_{t+1,0} := 1 + \beta_{t+1}, \ V_{t,0} := 1.$$
 (5.17)

Clearly,

$$\sum_{k=0}^{K} R_{t+1,k} = 1.$$
 (5.18)

Consider the investment strategy  $\Lambda^* = (\lambda_t^*)$  for which the vector  $\lambda_t^* = (\lambda_{t,0}^*, ..., \lambda_{t,K}^*)$  is given by

$$\lambda_{t,k}^*(s^t) := E[R_{t+1,k}(s^{t+1})|s^t].$$
(5.19)

The strategy  $\Lambda^*$  shows the idea to distribute wealth by the proportions of the conditional expectations of the relative asset payoffs.

Assume that for each k = 0, 1, 2, ..., K,

$$E\ln E_t R_{t+1,k}(s^{t+1}) > -\infty.$$
(5.20)

This assumption implies that the conditional expectation  $E_t R_{t+1,k} = E(R_{t+1,k}|s^t)$  is strictly positive a.s., and so we can choose a version of this conditional expectation that is strictly positive for all  $s^t$ . This version will be used in the definition (5.19) of the portfolio rule  $\Lambda^* = (\lambda_t^*)$ .

# Two Theorems for the Existence and Asymptotic Uniqueness of the Survival Strategy $\Lambda^*$ .

#### **Theorem 5.1** The portfolio rule $\Lambda^*$ is a survival strategy.

Note that the portfolio rule  $\Lambda^*$  belongs to the class of basic portfolio rules.

**Theorem 5.2** If  $\Lambda = (\lambda_t)$  is a basic survival strategy, then

$$\sum_{t=0}^{\infty} ||\lambda_t^* - \lambda_t||^2 < \infty \ (a.s.).$$

$$(5.21)$$

Here, " $||\cdot||$ " represents any norm in a finite-dimensional linear space. This theorem shows that in the class of basic portfolio rules, the survival strategy  $\Lambda^* = (\lambda_t^*)$  is essentially unique: any other basic survival strategy  $\Lambda$  is asymptotically similar to  $\Lambda^*$ . Recall that according to Section 2.4.2, survival strategies  $\Lambda$  can be characterized by the property that the wealth  $w_t^j$  of any other investors j cannot grow asymptotically faster, with strictly positive probability, than the wealth of investor i using  $\Lambda$ .

In fact, Theorems 5.1 and 5.2 are direct consequences of Theorems 2.1 and 2.2 proved in Section 2.3. The proof procedures of Theorems 5.1 and 5.2 are routine and analogous to those in Section 2.3; therefore, the detailed proofs of the main results of this model are relegated to Appendix B.

## Chapter 6

# Model with Short Selling

Up to now, all the models we have reviewed are established on the assumptions of excluding short selling and exogenous asset supply, which are the foundations upon which the comprehensive theory of evolutionary dynamics has been developed. In the model examined in this chapter, market participants are allowed to construct their portfolios not only with long, but also with short positions. To create short positions, investors issue "replicas" of the original assets, which share the same equilibrium prices and payoffs. These newly issued assets are sold at the equilibrium prices, generating the short-selling income for the issuer, which in turn increases the investment budget available for purchasing other assets (i.e., creating long portfolio positions). On the other hand, selling each unit of an asset short obligates the seller to pay the buyer the same payoff as the original asset. Moreover, short selling results in an increase of the exogenously given total quantity  $V_{t,k}$  of each asset k in the market, thereby influencing the equilibrium prices.

As with all the Evolutionary Finance models considered above, our primary focus is on the fundamental questions of the existence and (asymptotic) uniqueness of a survival strategy. First, we ask whether the strategy  $\Lambda^*$  (without short selling) guarantees survival in a market where the rivals of the  $\Lambda^*$ -investor can sell short, and if this strategy is (at least asymptotically) unique. The findings shown in Section 6.2 confirm that the strategy  $\Lambda^*$  (which rules out short selling) indeed guarantees survival in a market where short sales are allowed. However, are there strategies including short selling that also guarantee survival? If so, are they asymptotically distinct from  $\Lambda^*$ ? The answers are negative. <sup>1</sup>

### 6.1 The Model

A Summary on Changes in Model Settings. A starting point for this model remains the basic model presented in Section 2.1, and the changes that occur in this model are summarized as follows:

<sup>&</sup>lt;sup>1</sup>This chapter reviews the work by Amir, Belkov, Evstigneev, and Hens [1].

• Consider a market where  $K \geq 2$  risky assets (securities) are traded by  $N \geq 2$  investors (traders) at dates t = 0, 1, ..., and the market is influenced by random factors captured by an exogenous stochastic process  $s_1, s_2, ...$ (states of the world), where  $s_t$  is a random element in a measurable space  $S_t$ . At each date t, the total number of units of asset k available is  $V_{t,k} = V_{t,k}(s^t) > 0$  (with the constant  $V_{0,k} > 0$ ), and each investor i = 1, ..., N holds some wealth  $w_t^i = w_t^i(s^t)$  (with the initial endowments  $w_0^i > 0$ ). The payoff functions  $A_{t,k} = A_{t,k}(s^t) \geq 0$  for each asset k = 1, ..., K at date t = 1, 2, ..., are measurable and satisfy

$$\sum_{k=1}^{K} A_{t,k}(s^{t}) > 0.$$
(6.1)

- At each date  $t \ge 0$ , investor *i* purchases  $x_{t,k}^i \ge 0$  units of asset k = 1, ..., Kand/or sells short (at this date)  $y_{t,k}^i \ge 0$  units of this asset *k*. At time period [t, t+1], the payoff that investor *i* receives at date t+1 from  $x_{t,k}^i$ units of asset *k* will be  $A_{t+1,k}x_{t,k}^i = A_{t+1,k}(s^{t+1})x_{t,k}^i$ ; if, investor *i* sells short  $y_{t,k}^i$  units of asset *k* at date *t*, then this investor has an obligation to pay  $A_{t+1,k}(s^{t+1})y_{t,k}^i$  at date t+1.
- Denote the vectors of asset prices by  $p_t = p_{t,k}(s^t) = (p_{t,1}, ..., p_{t,K})$ , which are determined *endogenously* by the *market equilibrium*, reached when total supply of each asset k is equal to its total demand (i.e., the market clears):

$$V_{t,k} + \sum_{i=1}^{N} y_{t,k}^{i} = \sum_{i=1}^{N} x_{t,k}^{i}, \ k = 1, ..., K.$$
(6.2)

Meanwhile, the amount  $y_{t,k}^i$  of asset k that the investor has sold short provides the investor with extra budget for further investments, with this amount being  $p_{t,k}y_{t,k}^i$  (the short-selling income defined below). The total investment budget of trader i at date t, represented as  $w_t^i + v_t^i$ , consists of this trader's wealth  $w_t^i$  and the short selling income

$$v_t^i := \sum_{k=1}^K p_{t,k} y_{t,k}^i.$$
(6.3)

At each date t, each trader spends the entire available budget  $w_t^i + v_t^i$  to purchase assets:

$$w_t^i + v_t^i = \sum_{k=1}^K p_{t,k} x_{t,k}^i.$$

The wealth  $w_{t+1}^i$  of investor *i* at the end of the time period [t, t+1] can be calculated by

$$w_{t+1}^{i} := \sum_{k=1}^{K} A_{t+1,k} x_{t,k}^{i} - \sum_{k=1}^{K} A_{t+1,k} y_{t,k}^{i} = \sum_{k=1}^{K} A_{t+1,k} \left( x_{t,k}^{i} - y_{t,k}^{i} \right).$$
(6.4)

The quantities  $x_{t,k}^i \ge 0$ , k = 1, ..., K of the assets purchased by investor i form the vector  $x_t^i := (x_{t,1}^i, ..., x_{t,K}^i)$ , and the volumes  $y_{t,k}^i \ge 0$ , k = 1, ..., K of the assets sold short by investor i form the vector  $y_t^i := (y_{t,1}^i, ..., y_{t,K}^i)$ . The portfolio of investor i is given by the pair of vectors  $(x_t^i, -y_t^i)$ .

• For each  $t \ge 0$ , each investor i = 1, 2, ..., N selects a vector of investment proportions  $\gamma_t^i := (\gamma_{t,1}^i, ..., \gamma_{t,K}^i)$ , which belongs to the unit simplex

$$\gamma_t^i \in \Delta^K := \{ (a_1, ..., a_K) \in \mathbb{R}^K : a_1 + ... + a_K = 1, \, a_k \ge 0, \, k = 1, ..., K \}.$$
(6.5)

When selling short  $y_{t,k}^i$  units of asset k = 1, ..., K at date t, investor i issues and sells  $y_{t,k}^i$  "replicas" of asset k, which have the same price and yield the same payoff for the buyer at the next date t + 1 as the original asset k. Though this operation increases investor i's investment budget by  $p_{t,k}y_{t,k}^i$  at date t, it leads to an obligation to pay  $A_{t+1,k}y_{t,k}^i$  at date t + 1. Formally, in this model investor i's decision (or action) at date t is specified by a pair of vectors  $\xi_t^i = (\gamma_t^i, y_t^i)$ , where  $\gamma_t^i$  is the vector of investment proportions and  $y_t^i$  is the vector whose coordinates define the short positions. Note that long positions of a portfolio are specified in terms of *investment proportions*, while its short positions are defined in terms of units of assets!

• In the following analysis, we will consider only those decisions that do not permit to open simultaneously a long and a short position for the same asset, i.e.,

$$\gamma_{t,k}^{i} y_{t,k}^{i} = 0, \ k = 1, \dots, K, \ t \ge 0.$$
(6.6)

This property will be included into the definition of investors' decisions.

• Similar to the most general setting, the investment decisions  $\xi_t^i$  at each date  $t \ge 0$  are selected by N investors simultaneously and independently (as in a simultaneous-move N-person dynamic game). For  $t \ge 1$ , this decision making usually depends on  $s^t$  and the history of the game (or the history of the market)  $\xi^{t-1} := \{\xi_l^i, i = 1, ..., N, l = 0, ..., t-1\}$ . A pair of vectors  $\Xi_0^i = (\Gamma_0^i, Y_0^i) \in \Delta^K \times \mathbb{R}_+^K$  and a sequence of measurable functions  $\Xi_t^i(s^t, \xi^{t-1}) = (\Gamma_t^i(s^t, \xi^{t-1}), Y_t^i(s^t, \xi^{t-1}))$ , t = 1, 2, ..., taking values in  $\Delta^K \times \mathbb{R}_+^K$  form a portfolio rule  $\Xi^i$  of investor i, according to which investor i makes the decision

$$\xi_t^i = \Xi_t^i \left( s^t, \xi^{t-1} \right) \tag{6.7}$$

at each date  $t \ge 0$ . This provides a game-theoretic framework for defining general portfolio rules. Within this framework, we will distinguish those for which  $\Xi_t^i$  depends only on  $s^t$  and not on the game history  $\xi^{t-1}$ , and such portfolio rules  $\Xi_t^i(s^t)$  are termed basic.

A Description for the Asset Market Dynamics. Given a decision  $\xi_t^i = (\gamma_t^i, y_t^i)$  of investor *i*, the long positions  $x_{t,k}^i$  of *i*'s portfolio  $(x_t^i, -y_t^i)$  can be

computed by following formulas:

$$x_{t,k}^{i} = \frac{\gamma_{t,k}^{i}(w_{t}^{i} + v_{t}^{i})}{p_{t,k}} = \frac{1}{p_{t,k}}\gamma_{t,k}^{i}\left(w_{t}^{i} + \sum_{m=1}^{K} p_{t,m}y_{t,m}^{i}\right), \ k = 1, ..., K,$$
(6.8)

and the short positions of this portfolio are specified by the vector  $-y_t^i$ , where  $y_t^i = (y_{t,1}^i, ..., y_{t,K}^i)$ .

In the system of equations (6.2),  $x_{t,k}^i$  and  $v_t^i$  can be expressed by applying formulas (6.8) and (6.3), respectively. This leads to the following system of equations from which we can determine the vector  $p_t = (p_{t,1}, ..., p_{t,K})$  of equilibrium asset prices:

$$\sum_{i=1}^{N} \gamma_{t,k}^{i} \left( w_{t}^{i} + \sum_{m=1}^{K} p_{t,m} y_{t,m}^{i} \right) = p_{t,k} \left( V_{t,k} + \sum_{i=1}^{N} y_{t,k}^{i} \right), \ k = 1, \dots, K.$$
(6.9)

**Proposition 6.1** Let the following conditions hold:

$$w_t^i > 0, \ i = 1, ..., N; \ \sum_{i=1}^N \gamma_{t,k}^i w_t^i > 0, \ k = 1, ..., K.$$
 (6.10)

Then the system of equations (6.9) has a unique strictly positive solution  $p_t = (p_{t,1}, ..., p_{t,K}), p_{t,k} > 0$ , for each k.

According to this proposition, if at date t the wealth  $w_t^i$  of each investor i = 1, 2, ..., N is strictly positive and for each asset k = 1, ..., K at least one of the investors selects a strictly positive investment proportion  $\gamma_{t,k}^i > 0$ , then the asset market has a unique equilibrium with strictly positive prices. Note that if the first inequality in (6.10) is satisfied and at least one of the investors has a strictly positive vector of investment proportions  $\gamma_t^i = (\gamma_{t,1}^i, ..., \gamma_{t,K}^i)$ , then the second inequality in (6.10) holds as well.

We conclude this section with remarks on the design of the model at hand.

**Remark 6.1** Some comments on modeling long and short portfolio positions in our context. The approach to short selling that involves "replicas" of assets with the same exogenous payoffs is widely employed in mathematical models within Financial Economics (Magill and Quinzii [111]) and Mathematical Finance (Pliska [128], Ross [130], Föllmer and Schied [69], and Zierhut [170]). However, quite often it is not explicitly spelled out, as a deeper analysis of this question is usually unnecessary. Here, we wish to discuss this method in more detail, particularly due to a certain asymmetry in our model-long portfolio positions are specified in terms of investment proportions, while short ones are in "physical" units of assets-an asymmetry that is conceptually important and has a clear meaning, as it reflects the substantial difference between the operations of creating long and short portfolio positions in our context. The former is concerned with purchasing available assets by distributing wealth across them according to the given investment strategy. The latter operation, understood as creating new one-period assets that replicate the initial ones, is nothing but endogenous asset supply. In the purely financial context, endogenous asset supply may be regarded as the analogue of production in models with real assets. (This analogy becomes especially transparent if we examine the creation of derivative securities, rather then identical replicas of the basic assets.) It is worth noting that the liabilities arising from the creation of new securities-copying the basic ones-can be precisely estimated only if the number of units issued is known: for each unit of asset k sold short at time t, the seller must pay the buyer the amount (denoted in our model by)  $A_{t+1,k}(s^{t+1})$  at time t+1. Furthermore, since the exogenous component  $V_{t,k}$  in the equilibrium pricing equations (6.9) is expressed in terms of units of assets, the total asset supply must also be represented in the same way, which justifies the approach used in this paper for specifying short positions in investors' portfolios. As regards the long ones, theoretically one can describe them either in terms of units of assets or in terms of their monetary values and investment proportions. The latter approach is traditional in classical capital growth theory (see, e.g., Evstigneev et al. [63], Chapter 17,18), and since EF may be viewed as an extension of this theory to the case of endogenous asset prices, it is natural to design the model in a way similar to the classical one in order to use, whenever possible, similar machinery, notation, etc.<sup>2</sup>

## 6.2 The Main Results: Existence and Asymptotic Uniqueness of Survival Portfolio Rules

We focus on the analysis of the stochastic dynamics of investors' wealth depending on their strategies. Let  $w_t = (w_t^1, ..., w_t^N)$  be the vector of investors' wealth, and let  $\Xi = (\Xi^1, ..., \Xi^N)$  be the strategy profile of N investors. The dynamics of  $w_t$  will be defined recursively, step by step from t to t + 1, given that the initial state (at t = 0) is the vector  $w_0 = (w_0^1, ..., w_0^N)$ , which stands for the initial endowment of each investor. Suppose  $w_0, w_1, ..., w_t$  are defined for some  $t \ge 0$ . Assume that the following condition holds:

(A). The vector  $w_t = (w_t^1, ..., w_t^N)$  and the investment proportions  $\gamma_{t,k}^i$  (generated the strategy profile  $\Xi$ ) satisfy (6.10).

Then according to Proposition 6.1, there exists a unique strictly positive vector  $p_t = (p_{t,1}, ..., p_{t,K})$  of equilibrium asset prices, in terms of which we can

<sup>&</sup>lt;sup>2</sup>For further reading on other models in capital growth theory and EF involving short selling and endogenous asset supply, see Bucher and Woehrmann [38], Horváth and Urbán [92], Schenk-Hoppé and Sokko [138].

express the wealth  $w_{t+1}^i$  of each investor *i*:

$$w_{t+1}^{i} = \sum_{k=1}^{K} A_{t+1,k} x_{t,k}^{i} - \sum_{k=1}^{K} A_{t+1,k} y_{t,k}^{i}$$
$$= \sum_{k=1}^{K} A_{t+1,k} \frac{1}{p_{t,k}} \gamma_{t,k}^{i} \left( w_{t}^{i} + \sum_{m=1}^{K} p_{t,m} y_{t,m}^{i} \right) - \sum_{k=1}^{K} A_{t+1,k} y_{t,k}^{i}$$
$$= \sum_{k=1}^{K} A_{t+1,k} \left( \frac{1}{p_{t,k}} \gamma_{t,k}^{i} \left( w_{t}^{i} + \sum_{m=1}^{K} p_{t,m} y_{t,m}^{i} \right) - y_{t,k}^{i} \right).$$
(6.11)

**Definition 6.1** If in the course of this dynamical process, condition (A) happens to hold almost surely  $(a.s.)^3$  for all  $t \in [0,T)$   $(T \leq \infty)$ , we say that the given strategy profile  $\Xi$  is admissible for the time interval [0,T).

**Remark 6.2** Assume that one of the investors, e.g., investor 1, adopts a *fully* diversified portfolio rule that prescribes investing into all assets in strictly positive proportions  $\gamma_{t,k}^1 > 0$  for all k = 1, ..., K and all  $t \ge 0$ . Then, given that no investor goes bankrupt during the time interval [0, T), the strategy profile is admissible for the time interval [0, T).

By summing up (6.11) over i = 1, ..., N and taking into account the pricing equations (6.9), we get the following formula for the total market wealth  $W_{t+1}$ :

$$W_{t+1} := \sum_{i=1}^{N} w_{t+1}^{i} = \sum_{i=1}^{N} \sum_{k=1}^{K} A_{t+1,k} \left( \frac{1}{p_{t,k}} \gamma_{t,k}^{i} \left( w_{t}^{i} + \sum_{m=1}^{K} p_{t,m} y_{t,m}^{i} \right) - y_{t,k}^{i} \right)$$
$$= \sum_{k=1}^{K} A_{t+1,k} \left( \frac{1}{p_{t,k}} \sum_{i=1}^{N} \left[ \gamma_{t,k}^{i} \left( w_{t}^{i} + \sum_{m=1}^{K} p_{t,m} y_{t,m}^{i} \right) \right] - \sum_{i=1}^{N} y_{t,k}^{i} \right)$$
$$= \sum_{k=1}^{K} A_{t+1,k} \left( V_{t,k} + \sum_{i=1}^{N} y_{t,k}^{i} - \sum_{i=1}^{N} y_{t,k}^{i} \right) = \sum_{k=1}^{K} A_{t+1,k} V_{t,k}.$$
(6.12)

As before, denote by  $r_{t+1}^i := \frac{w_{t+1}^i}{W_{t+1}}$  the relative wealth (market share) of investor *i* and put  $r_{t+1} := (r_{t+1}^1, ..., r_{t+1}^N)$ .

**Definition 6.2** We say that a strategy  $\Xi^i$  employed by investor *i* can be driven out of the market at a (finite) time  $T < \infty$  if there exists a strategy profile  $(\Xi^1, ..., \Xi^N)$  including the strategy  $\Xi^i$  and admissible for  $t \in [0, T)$  such that  $P\{w_T^i \leq 0\} > 0.$ 

<sup>&</sup>lt;sup>3</sup>In this chapter, we will identify random variables coinciding almost surely, and we might often omit "a.s." if this does not lead to ambiguity. In fact the random variables under consideration may be defined not everywhere, but only almost everywhere: with probability one. If not otherwise stated, all relations between them (equalities, inequalities, etc.) will be supposed to hold almost surely.

**Definition 6.3** We say that a strategy  $\Xi^i$  employed by investor *i* can be driven out of the market in an infinite time if there exists a strategy profile  $(\Xi^1, ..., \Xi^N)$ including the strategy  $\Xi^i$  and admissible for  $t \in [0, \infty)$  such that  $P\left\{\inf_{t\geq 0} r_t^i = 0\right\} > 0$ .

**Definition 6.4** We call a strategy  $\Xi$  a survival strategy if for any number  $N \ge 2$  of agents acting in the market an investor using  $\Xi$  cannot be driven out of the market in any (finite or infinite) time.

**Explicit Expression for the Survival Strategy**  $\Xi^*$ . Define the *relative payoffs* by

$$R_{t+1,k} := \frac{A_{t+1,k}V_{t,k}}{\sum_{l=1}^{K} A_{t+1,l}V_{t,l}}$$
(6.13)

and put  $R_{t+1} := (R_{t+1,1}, ..., R_{t+1,K})$ . Consider the investment strategy  $\Xi^* = (\xi_t^*)_{t=0}^{\infty} = (\gamma_t^*, y_t^*)_{t=0}^{\infty}$  for which  $y_t^* = 0$  and  $\gamma_t^*(s^t) := (\gamma_{t,1}^*(s^t), ..., \gamma_{t,K}^*(s^t))$ , where

$$\gamma_{t,k}^*(s^t) := E_t R_{t+1,k}(s^{t+1}). \tag{6.14}$$

Throughout this chapter, we will assume that for each k = 1, ..., K,

$$E\ln E_t R_{t+1,k}(s^{t+1}) > -\infty.$$
(6.15)

This assumption implies that the conditional expectations  $E_t R_{t+1,k} = E(R_{t+1,k}|s^t)$ (k = 1, ..., K) are strictly positive (a.s.). Consequently, we can choose their versions  $\gamma_{t,k}^*(s^t)$  which are strictly positive for all  $s^t$ . In following analysis, the notation  $\gamma_{t,k}^*(s^t)$  will refer to such versions of the above conditional expectations.

# Three Theorems for the Existence and Asymptotic Uniqueness of the Survival Strategy $\Xi^*$ .

#### **Theorem 6.1** The portfolio rule $\Xi^*$ is a survival strategy.

It is important to note that the portfolio rule  $\Xi^*$  is basic and it does not include short selling, but it survives in competition with *all*, not necessarily basic, strategies with short selling. The results we formulate below show that in the class of basic strategies, the survival portfolio rule  $\Xi^*$  is (at least asymptotically) unique.

**Theorem 6.2** If a basic strategy prescribes to sell short at least one asset at some moment of time with strictly positive probability, then it can be driven out of the market in a finite time.

Hence, basic survival portfolio rules can exist only in the class of basic strategies that do not involve short selling (a.s.). It should be emphasized that a short seller can be driven out of the market in a finite time by a basic strategy profile of the rivals, which will be shown in the course of the proof of Theorem 6.2. **Theorem 6.3** If  $\Xi$  is a basic survival strategy defined by a sequence of decisions  $(\gamma_t(s^t), y_t(s^t)), t = 0, 1, 2..., with <math>y_t(s^t) = 0$  (a.s.), then

$$\sum_{t=0}^{\infty} ||\gamma_t^* - \gamma_t||^2 < \infty \ (a.s.).$$
(6.16)

This theorem (pertaining to a version of the present model without short selling) follows easily from Theorem 2.2, Section 2.2. Notice that vectors of investment proportions  $\gamma_t^*$  coincide with vectors  $\lambda_t^*$  generated by the strategy  $\Lambda^*$  considered in Section 2.2. In that model (where no short-selling is allowed),  $\Lambda^*$  is also an asymptotically unique basic survival strategy.

Proofs of Proposition 6.1, and Theorems 6.1, 6.2 and 6.3 are given in the next section.

### 6.3 Proofs of the Main Results

### 6.3.1 Short-run Equilibrium

Proof of Proposition 6.1. Fix t and omit it in the notation. For all k = 1, ..., K, define

$$u_{k} := p_{k} \left( V_{k} + \sum_{i=1}^{N} y_{k}^{i} \right), \ \sigma_{k} := \frac{\sum_{i=1}^{N} y_{k}^{i}}{V_{k} + \sum_{i=1}^{N} y_{k}^{i}},$$
$$\theta_{k}^{i} := \begin{cases} \frac{y_{k}^{i}}{\sum_{j=1}^{N} y_{k}^{j}}, & \text{if } \sum_{j=1}^{N} y_{k}^{j} > 0, \\ N^{-1}, & \text{otherwise.} \end{cases}$$

Note that  $0 \leq \sigma_k < 1$  because  $V_k > 0$ , and we have  $\sum_{i=1}^N \theta_k^i = 1$  and  $\theta_k^i \geq 0$ . If  $\sum_{j=1}^N y_k^j > 0$ , then for each m = 1, ..., K, the following identity holds:

$$y_{m}^{i} = \left(V_{m} + \sum_{j=1}^{N} y_{m}^{j}\right) \frac{(\sum_{j=1}^{N} y_{m}^{j})y_{m}^{i}}{(V_{m} + \sum_{j=1}^{N} y_{m}^{j}) \cdot \sum_{j=1}^{N} y_{m}^{j}} = \left(V_{m} + \sum_{j=1}^{N} y_{m}^{j}\right)\sigma_{m}\theta_{m}^{i},$$

which gives

$$p_m^i y_m^i = u_m \sigma_m \theta_m^i, \ m = 1, ..., K.$$

If  $\sum_{j=1}^{N} y_k^j = 0$ , the above equality is valid as well, since in that case  $y_m^i = \sigma_m = 0$ . Consequently, the system of equations (6.9) can be written as

$$\sum_{i=1}^{N} \gamma_k^i w^i + \sum_{i=1}^{N} \gamma_k^i \sum_{m=1}^{K} \theta_m^i \sigma_m u_m = u_k, \ k = 1, ..., K.$$
(6.17)

A vector  $u = (u_1, ..., u_K)$  solves (6.17) if and only if u is a fixed point of the operator

$$F(u) := \left(\sum_{i=1}^{N} \gamma_k^i w^i + \sum_{i=1}^{N} \gamma_k^i \sum_{m=1}^{K} \theta_m^i \sigma_m u_m\right)_{k=1}^{K}$$

which is determined by the left-hand side of (6.17). This operator transforms the cone

$$\mathbb{R}^{K}_{+} = \{ u \, | \, u_k \ge 0, \, k = 1, ..., K \}$$

of non-negative K-dimensional vectors into itself. And to establish that it has a unique fixed point in  $\mathbb{R}_{+}^{K}$ , it suffices to show that it is contracting in the norm  $||u|| = \sum_{k=1}^{K} |u_k|$ . This follows from the chain of relations:

$$\|F(u) - F(u')\| = \sum_{k=1}^{K} |\sum_{i=1}^{N} \gamma_{k}^{i} \sum_{m=1}^{K} \theta_{m}^{i} \sigma_{m} (u_{m} - u'_{m})|$$
  
$$\leq (\max_{m} \sigma_{m}) \sum_{i=1}^{N} (\sum_{k=1}^{K} \gamma_{k}^{i}) \sum_{m=1}^{K} \theta_{m}^{i} |u_{m} - u'_{m}|$$
  
$$= (\max_{m} \sigma_{m}) \sum_{m=1}^{K} |u_{m} - u'_{m}| \sum_{i=1}^{N} \theta_{m}^{i} = (\max_{m} \sigma_{m}) ||u - u'||$$

where the equalities hold because  $\sum_{k=1}^{K} \gamma_k^i = 1$  and  $\sum_{i=1}^{N} \theta_k^i = 1$ . Here,  $\max_m \sigma_m < 1$  since  $\sigma_m < 1$  for all m, and thus the operator F is contracting. Therefore, the system (6.17) has a unique solution  $u = (u_1, ..., u_K) \ge 0$ . Moreover, if condition (6.10) is satisfied, then

$$u_k = \sum_{i=1}^N \gamma_k^i w^i + \sum_{i=1}^N \gamma_k^i \sum_{m=1}^K \theta_m^i \sigma_m u_m \ge \sum_{i=1}^N \gamma_k^i w^i > 0$$

and the system (6.9) also has the unique strictly positive solution

$$p = (p_1, ..., p_K), \ p_k = \frac{u_k}{V_k + \sum_{i=1}^N y_k^i}$$

and  $p_k > 0$  for each k.

#### 6.3.2 Survival Portfolio Rules

Proof of Theorem 6.1. Consider a market with  $N \geq 2$  investors. Suppose that investor 1 employs the strategy  $(\gamma_t^*, 0) = (E_t R_{t+1}(s^{t+1}), 0)$ . Agent 1, using the strategy  $\Xi^*$ , cannot be driven out of the market at any finite time  $0 < T < \infty$  because assumption (6.15) guarantees that investor 1's portfolio is fully diversified (a.s.), thus,  $w_T^1 > 0$  (a.s.). We shall show that  $\xi_t^*$  cannot be driven out of the market for the time interval  $[0, \infty)$ .

Consider a strategy profile  $\Xi = (\Xi^1, ..., \Xi^N)$  that is admissible over  $[0, \infty)$ , for which  $w_t^i > 0$  a.s. for all  $t \ge 0$  and i = 1, ..., N. Let  $(\xi_t^1(s^t), ..., \xi_t^N(s^t))_{t=0}^{\infty}$  be the set of investors' decisions generated by  $\Xi$ . For all  $t \ge 0$ , k = 1, ..., K, i = 1, ..., N, define the numbers

$$\lambda_{t,k}^{i} = \lambda_{t,k}^{i} \left( s^{t} \right) = \frac{w_{t}^{i} + \sum_{m=1}^{K} p_{t,m} y_{t,m}^{i}}{w_{t}^{i}} \gamma_{t,k}^{i} - \frac{p_{t,k} y_{t,k}^{i}}{w_{t}^{i}}, \qquad (6.18)$$

and the vectors  $\lambda_{t,k} = \left(\lambda_{t,k}^1, ..., \lambda_{t,k}^N\right)$  and  $\lambda_t^i = \left(\lambda_{t,1}^i, ..., \lambda_{t,K}^i\right)$ . Note that  $\lambda_t^1 = \gamma_t^*$  since  $\xi_t^1 = (\gamma_t^*, 0)$ .

 $\gamma_t^*$  since  $\xi_t^1 = (\gamma_t^*, 0)$ . We shall formulate a system of equations that describes the dynamics of the market shares  $r_t^i$  in terms of the sequence of vectors  $(\lambda_t^1, ..., \lambda_t^N)$ . We have

$$p_{t,k}V_{t,k} = \langle \lambda_{t,k}, w_t \rangle, \ k = 1, ..., K,$$

where  $\langle \lambda_{t,k}, w_t \rangle := \sum_{i=1}^N \lambda^i_{t,k} w^i_t$ . Indeed,

$$\sum_{i=1}^{N} \lambda_{t,k}^{i} w_{t}^{i} = \sum_{i=1}^{N} (w_{t}^{i} + \sum_{m=1}^{K} p_{t,m} y_{t,m}^{i}) \gamma_{t,k}^{i} - \sum_{i=1}^{N} p_{t,k} y_{t,k}^{i} = p_{t,k} V_{t,k}$$

in view of (6.18) and (6.9). Proposition 6.1 and the definition of an admissible strategy profile over time interval  $[0, \infty)$  guarantee that

$$p_{t,k} = \frac{\langle \lambda_{t,k}, w_t \rangle}{V_{t,k}} > 0 \text{ (a.s.)}, \ k = 1, ..., K.$$
(6.19)

By virtue of (6.18) and (6.8), we obtain

$$\frac{\lambda_{t,k}^{i}w_{t}^{i}V_{t,k}}{\langle\lambda_{t,k},w_{t}\rangle} = \frac{\lambda_{t,k}^{i}w_{t}^{i}}{p_{t,k}} = \frac{w_{t}^{i} + \sum_{m=1}^{K} p_{t,m}y_{t,m}^{i}}{p_{t,k}}\gamma_{t,k}^{i} - \frac{p_{t,k}y_{t,k}^{i}}{p_{t,k}} = x_{t,k}^{i} - y_{t,k}^{i} \quad (6.20)$$

(k = 1, ..., K). Hence, the wealth  $w_{t+1}^i$  of investor *i* given by (6.4) can be expressed as follows:

$$w_{t+1}^{i} = \sum_{k=1}^{K} A_{t+1,k} \left( x_{t,k}^{i} - y_{t,k}^{i} \right) = \sum_{k=1}^{K} A_{t+1,k} V_{t,k} \frac{\lambda_{t,k}^{i} w_{t}^{i}}{\langle \lambda_{t,k}, w_{t} \rangle}.$$
 (6.21)

By summing up the equations in (6.21) over i = 1, ..., N, we get

$$W_{t+1} = \sum_{k=1}^{K} A_{t+1,k} V_{t,k} \frac{\sum_{i=1}^{N} \lambda_{t,k}^{i} w_{t}^{i}}{\langle \lambda_{t,k}, w_{t} \rangle} = \sum_{k=1}^{K} A_{t+1,k} V_{t,k}.$$
 (6.22)

Dividing the left-hand side of (6.21) by  $W_{t+1}$ , and the right-hand side of (6.21) by

$$\sum_{m=1}^{K} A_{t+1,m} V_{t,m},$$

and applying (6.13), we arrive at the system of equations

$$r_{t+1}^{i} = \sum_{k=1}^{K} R_{t+1,k} \frac{\lambda_{t,k}^{i} r_{t}^{i}}{\langle \lambda_{t,k}, r_{t} \rangle}, \ i = 1, ..., N.$$
(6.23)

Since we consider a strategy profile  $(\Xi^1, ..., \Xi^N)$  that is admissible for the time interval  $[0, \infty)$ , all the market shares are strictly positive:

$$r_t^i > 0, \ i = 1, ..., N, \ t \ge 0.$$
 (6.24)

Observe that

$$\sum_{k=1}^{K} \lambda_{t,k}^{i} = \frac{w_{t}^{i} + \sum_{m=1}^{K} p_{t,m} y_{t,m}^{i}}{w_{t}^{i}} \sum_{k=1}^{K} \gamma_{t,k}^{i} - \frac{\sum_{k=1}^{K} p_{t,k} y_{t,k}^{i}}{w_{t}^{i}} = 1,$$

and thus

$$\sum_{k=1}^{K} \lambda_{t,k}^{i} = 1, \ t \ge 0.$$
(6.25)

From (6.19) we conclude that

$$\langle \lambda_{t,k}, r_t \rangle > 0, \ k = 1, ..., K$$
 (6.26)

for each  $t \geq 0$ .

To complete the proof, we shall use Proposition 6.2 from which we conclude that  $\inf_{t\geq 0} r_t^1 > 0$  (a.s.) and investor 1 cannot be driven out of the market at moment  $T = \infty$ . Therefore,  $\Xi^*$  is a survival strategy.

**Proposition 6.2** If  $\lambda_t^1(s^t) = E_t R_{t+1}(s^{t+1})$  and dynamical system (6.23) satisfies the conditions (6.24)-(6.26) then

$$\inf_{t \ge 0} r_t^1 > 0 \ (a.s.).$$

The proof for Proposition 6.2 is exactly the same as the  $2^{nd}$  and  $3^{rd}$  steps of the proof for Theorem 2.1, presented in Section 2.3.

#### 6.3.3 Asymptotic Uniqueness of Survival Strategies

Proof of Theorem 6.2. Put  $\overline{K} = \{1, ..., K\}$  and consider a market with N = 2 investors. Assume that investor 2 adopts a strategy  $\Xi^2$  prescribing to open a short position with strictly positive probability for some asset at some moment of time, and let  $T \ge 0$  be the smallest among such moments of time. Then, there exists a non-random M-element subset  $\overline{M} \subset \overline{K}$  such that  $M \ge 1$  and the event

$$\bar{S}^T := \{s^T : y_{T,k}^2\left(s^T\right) > 0 \text{ for } k \in \bar{M} \text{ and } y_{T,k}^2\left(s^T\right) = 0 \text{ for } k \in \bar{K} \setminus \bar{M}\}$$

has a strictly positive probability. (Note that  $k \in \overline{K} \setminus \overline{M} \neq \emptyset$  because investor 2 cannot sell short *all* the assets.)

Without loss of generality, we can assume that investor 2 cannot be driven out of the market at any time  $t \leq T$ ; otherwise, the theorem is proved. Under this condition, we construct a *spiteful strategy*  $\Xi^1$  of investor 1, i.e. a strategy driving investor 2 out of the market at time T + 1. With the strategy  $\Xi^1$ , the strategy profile  $(\Xi^1, \Xi^2)$  will be admissible for the time interval [0, T+1) and the wealth  $w_{T+1}^2$  of investor 2 will be negative with strictly positive probability. The negativity of  $w_{T+1}^2$  with strictly positive probability will be established under an additional assumption that the initial wealth  $w_0^2$  of investor 2 is sufficiently small comparative to the initial wealth  $w_0^1$  of his/her rival.

Fix some  $\mu \in \overline{M}$ . For any  $\delta > 0$  and  $\delta' > 0$ , denote by  $\overline{S}^{T+1}(\delta, \delta')$  the following event:

$$\bar{S}^{T+1}\left(\delta,\delta'\right) = \{s^{T+1} : s^T \in \bar{S}^T, \ V_{T,\mu}^{-1}R_{T+1,\mu}(s^{T+1}) > \delta', \ y_{T,\mu}^2\left(s^T\right) > \delta\}.$$
(6.2)

Since  $P\{s^T \in \bar{S}^T\} > 0$  and  $P\{V_{T,\mu}^{-1}R_{T+1,\mu} > 0 \mid s^T\} > 0$  a.s. (in view of assumption (6.15)), there exist  $\delta > 0$  and  $\delta' > 0$  such that  $P\{s^{T+1} \in \bar{S}^{T+1}(\delta, \delta')\} > 0$ 0.

Fix a positive number  $\varepsilon < \min(\delta'\delta/(M \cdot K), 1/K)$  and define the strategy  $\Xi^1$  as follows. For each  $t \in [0, T+1)$ , put  $y_t^1 = 0$  (no short selling) and define:

$$e = (1, 1, ..., 1),$$
  

$$\gamma_t^1 = \frac{1}{2}\gamma_t^2 + \frac{1}{2K}e,$$
(6.28)

for t < T and

$$\gamma_{T,k}^{1} = \begin{cases} \varepsilon & \text{if } k \in \bar{M}, \\ \frac{1-M\varepsilon}{K-M} & \text{if } k \in \bar{K} \setminus \bar{M}. \end{cases}$$
(6.29)

Since investor 2 cannot be driven out of the market at any time  $t \leq T$ , we conclude that  $w_t^2 > 0$  (a.s.) for  $t \leq T$ . Further, observe that  $\gamma_{t,k}^1 > 0$ , which implies  $w_t^1 > 0$  and  $\sum_{i=1}^N \gamma_{t,k}^i w_t^i \ge \gamma_{t,k}^1 w_t^1 > 0$  for each  $k \in \overline{K}$ . Therefore, the strategy profile  $(\Xi^1, \Xi^2)$  is admissible for the time interval [0, T+1). Note that  $y_t^1 = y_t^2 = 0$  and  $\gamma_t^1 \ge \frac{1}{2}\gamma_t^2$  (see (6.28)) for each t < T. Conse-

quently, by virtue of (6.11) and (6.8), we get

$$\frac{w_{t+1}^2}{w_{t+1}^1} = \frac{\sum_{k \in \bar{K}} A_{t+1,k} x_{t,k}^2}{\sum_{k \in \bar{K}} A_{t+1,k} x_{t,k}^1} = \frac{\sum_{k \in \bar{K}} A_{t+1,k} \frac{\gamma_{t,k}^2 w_t^2}{p_{t,k}}}{\sum_{k \in \bar{K}} A_{t+1,k} \frac{\gamma_{t,k}^1 w_t^1}{p_{t,k}}} \le 2\frac{w_t^2}{w_t^1}, \quad t < T.$$
(6.30)

Now assume that the initial wealth  $w_0^2$  of investor 2 is small enough compared to the initial wealth  $w_0^1$  of investor 1, specifically,

$$\frac{w_0^2}{w_0^1} \le \frac{1}{2^T} \left( \frac{\delta'\delta}{K} - M\varepsilon \right). \tag{6.31}$$

Then, the inequality in (6.30) yields

$$\frac{w_T^2}{w_T^1} \le 2^T \frac{w_0^2}{w_0^1} \le \frac{\delta'\delta}{K} - M\varepsilon.$$
(6.32)

For  $s^{T+1} \in \bar{S}^{T+1}\left(\delta, \delta^{'}\right)$ , we obtain

$$\gamma_{T,k}^2 = 0 \text{ for } k \in \bar{M} \tag{6.33}$$

because  $y_{T,k}^2 > 0$  for  $k \in \overline{M}$ . In view of (6.9), (6.33) and (6.29), the prices  $p_{T,k}$  satisfy the following conditions:

$$p_{T,k} = \frac{w_T^1 \gamma_{T,k}^1 + \left(w_T^2 + v_T^2\right) \gamma_{T,k}^2}{V_{T,k} + y_{T,k}^2} = \frac{w_T^1 \varepsilon}{V_{T,k} + y_{T,k}^2}, \text{ if } k \in \bar{M},$$
(6.34)

$$p_{T,k} \ge \frac{w_T^1 \cdot \gamma_{T,k}^1}{V_{T,k}} \ge \frac{w_T^1}{V_{T,k}} \frac{1}{K}, \text{ if } k \in \bar{K} \setminus \bar{M}.$$
(6.35)

By applying relations (6.8), (6.34), (6.35) and (6.32) to estimate  $x_{T,k}^2$ ,  $k \in \bar{K} \setminus \bar{M}$ , we find that

$$x_{T,k}^{2} \leq \frac{w_{T}^{2} + \sum_{m \in \bar{M}} p_{T,m} y_{T,m}^{2}}{p_{T,k}} \leq \left(w_{T}^{2} + \sum_{m \in \bar{M}} \frac{w_{T}^{1} \varepsilon \cdot y_{T,m}^{2}}{V_{T,m} + y_{T,m}^{2}}\right) \cdot \left(\frac{w_{T}^{1}}{V_{T,k}} \frac{1}{K}\right)^{-1} \leq \left(\frac{w_{T}^{2}}{w_{T}^{1}} + M\varepsilon\right) V_{T,k} K \leq \delta' \delta V_{T,k}.$$
(6.36)

Let  $s^{T+1} \in \overline{S}^{T+1}(\delta, \delta')$ . By using (6.27), (6.36), and (6.13), we arrive at the wealth of investor 2 expressed by the following sequence of inequalities:

$$w_{T+1}^{2} \left( s^{T+1} \right) = -\sum_{k \in \bar{M}} A_{T+1,k} y_{T,k}^{2} + \sum_{k \in \bar{K} \setminus \bar{M}} A_{T+1,k} x_{T,k}^{2}$$
  
$$\leq -A_{T+1,\mu} \delta + \delta' \delta \sum_{k \in \bar{K} \setminus \bar{M}} A_{T+1,k} V_{T,k} + \delta' \delta \sum_{k \in \bar{M}} A_{T+1,k} V_{T,k}$$
  
$$= -A_{T+1,\mu} \delta + \delta' \delta \sum_{k \in \bar{K}} A_{T+1,k} V_{T,k} = \delta \cdot \sum_{k \in \bar{K}} A_{T+1,k} V_{T,k} \cdot \left( \delta' - \frac{R_{T+1,\mu}}{V_{T,\mu}} \right) \leq 0$$

Therefore, if condition (6.31) holds and investor 1 employs the strategy defined by (6.29) and (6.28), then

$$P\left\{w_{T+1}^2 \le 0\right\} \ge P\left\{s^{T+1} \in \bar{S}^{T+1}\left(\delta, \delta'\right)\right\} > 0,$$

which means that the strategy  $\Xi^2$  does not survive.

Proof of Theorem 6.3. Consider the basic strategy  $\Xi'$  defined by the sequence of decisions  $(\gamma_t(s^t), 0)$  (t = 0, 1, ...). Denote by  $w_t, t = 0, 1, ...$  the wealth of investor using  $\Xi$  and by  $w'_t, t = 0, 1, ...$  the wealth of investor using  $\Xi'$ . Then  $w'_t = w_t$  (a.s.), and  $\Xi'$  is a survival strategy. Strategies  $\Xi^*$  and  $\Xi'$  do not allow for short selling (not just a.s., but everywhere). From Theorem 2.2 given in Section 2.2, we obtain (6.16).

# Chapter 7

# Conclusion

This paper reviews some key models with short-lived assets in the field of Evolutionary Finance (EF). EF focuses on "survival and extinction" questions of investment portfolio rules in the market selection process. In contrast with the conventional theory of equilibrium and dynamics of asset markets that based on the Walrasian equilibrium paradigm, the EF theory we examined depicts a radically different world: 1) we consider a notion of behavioral equilibrium defined in consecutive short-run terms with a whole variety of strategy patterns, rather than maximizing small investors' individual utilities of consumption subject to budget constraints  $^1$ : 2) we eliminate the need for the "perfect foresight" assumption to establish equilibrium; 3) we get rid of the reliance on knowledge of unobservable individual utilities and beliefs, which is hard to capture in investment practice; 4) a world of large, even super large (primarily institutional) investors who may operate on a global level and pursue objectives of an evolutionary nature: domination, fastest growth and survival (especially in crisis environments). Indeed, fastest growth is often related, and as shown in Section 2.4.2 is equivalent, to survival. Within this framework, each market actor's investment decisions may considerably influence equilibrium prices, as opposed to many classical market models where individual influence is generally negligible. Moreover, from game-theoretic aspects, the models we considered in this paper involve elements of both stochastic dynamic games (strategic frameworks) and evolutionary game theory (solution concepts).

The primary goal of the studies is to identify investment strategies that guarantee "long-run survival", i.e., keeping a strictly positive, bounded away from zero, share of market wealth over an infinite time horizon, irrespective of what strategies used by others. It turns out that in all the models we considered, there always exists a portfolio rule ensuring unconditional long-run survival, which is given by Kelly's well-known portfolio rule of "betting one's beliefs" and can be expressed with explicit formulas. Additionally, we demonstrate the

<sup>&</sup>lt;sup>1</sup>In Walrasian equilibrium paradigm, market equilibrium can be understood as a situation in which the objectives (interests) of such economic agents are equilibrated by the market clearing prices (see, e.g., Flåm [68]).

asymptotic uniqueness (within a specific class of basic strategies) of such a rule.

We started with the basic modeling framework under substantially more general assumptions in Chapter 2. The results regarding the existence of a survival strategy and its asymptotic uniqueness within the class of basic portfolio rules are proved. However, as demonstrated by the counterexample shown in Section 2.4.1, this property of asymptotic uniqueness does not naturally extend to the entire class of general portfolio rules. Moreover, we also discussed that under such model settings, a portfolio rule guarantess survival if and only if it is unbeatable – in the sense that "in order to survive you have to win".

Chapter 3 presented an approach to formulating an evolutionary solution concept in terms of a non-conventional (stronger) version of Nash equilibrium that holds almost surely, where a strategy strictly dominates the market at an exponential rate. This can be achieved by introducing the Lyapunov exponent of the relative growth of wealth of a market participant-"relative" referring to the ratio of the player's wealth to the total wealth of the group of his/her rivals—as the objective function. It is a stronger version in the sense that, according to this definition of an equilibrium strategy, any unilateral deviation from the unique symmetric Nash equilibrium leads to a decrease in the random payoff with probability one, not just in the expected payoff. Hence, by replacing the standard Nash equilibrium conditions (formulated using expected utilities) by holding almost surely, we obtained a solution concept equivalent to the exponential domination of the market. As a consequence, those investors who adopt the Kelly portfolio rule achieve the highest growth rate of wealth in any population of fixed-mix strategies, ultimately becoming the single survivor who accumulates the entire market wealth in the long run.

Chapter 4 described a model with a Markovian nature, which removed two strong assumptions from Chapter 3, and instead, used a homogeneous discretetime Markov process to describe the states of the world. The central result indicated that an investor who distributes wealth across available assets according to their relative conditional expected payoffs (a direct analogue of "betting one's beliefs" rule above) is a single survivor, provided this strategy is asymptotically distinct from the CAPM rule.

In Chapter 5 and Chapter 6, we examined models extended to the market involving a risk-free asset (typically represented as cash) and short selling, respectively. Our primary interest was also in the fundamental questions of the existence and (asymptotic) uniqueness of a survival strategy, and the answers to both questions are affirmative in both models.

EF is a promising yet still emerging field of research, with many questions remaining unanswered. In this paper, our focus is on the class of short-lived assets, which is more amenable to mathematical analysis and enable the development of a more complete and transparent theory. Because of this, it often serves as a "proving ground" for testing new conjectures in this field. For instance, in Chapter 5 and 6, we examined models involving a risk-free asset (cash) and short-selling, respectively. These two problems have been addressed in models of short-lived assets, but for long-lived assets they remain open.<sup>2</sup> In particular, the classical theory of derivative securities pricing, such as the Black-Scholes formula, is grounded in hypotheses which can be rigorously formulated only in the asset market models where short selling is allowed. This makes further investigation of this direction both theoretically and practically significant in quantitative financial applications.

 $<sup>^{2}</sup>$  For a different evolutionary finance model, dealing with long-lived assets (see Evstigneev et al. [59]). And note that the counterpart of the global evolutionary stability that we stated in Chapter 3, was obtained for models with long-lived assets in Bahsoun et al. [12], thereby this gap has been filled.

# Appendix A

# Some Elementary Inequalities

In this paper, there are some elementary inequalities: (i) the inequality given in Evstigneev et al. [58], Lemma 1, (ii) Lemma 3.1, and (iii) Lemma 4.1, that play a key role in proving some main results: Theorem 3.4, Theorem 3.3, and Theorem 4.1, respectively. In fact, the last two inequalities, (ii) and (iii), are both based on the result of (i). To underpin and simplify the derivations, we begin with inequality (i) below, then (ii) and (iii).

### Proof of Evstigneev et al. [58], Lemma 1

**Lemma A.1 (Evstigneev et al. [58], Lemma 1)** For any  $\mu = (\mu_k) \in \Delta^K$ with  $\mu > 0$  and any  $\kappa \in [0, 1]$ , we have

$$E \ln \sum_{k=1}^{K} R_k(s) \frac{\lambda_k^*}{\lambda_k^* \kappa + \mu_k (1-\kappa)} - E \ln \sum_{k=1}^{K} R_k(s) \frac{\mu_k}{\lambda_k^* \kappa + \mu_k (1-\kappa)} \ge 0.$$
 (A.1)

Furthermore, if  $\lambda^* \neq \mu$ , then the difference on the left-hand side of (A.1) is strictly positive.

*Proof.* Clearly, if  $\mu = \lambda^*$ , inequality (A.1) turns into an equality. We shall show that the expression on the left-hand side of (A.1), which is denoted by  $\Upsilon(\mu, \kappa)$ , is strictly positive for all  $\kappa \in [0, 1]$  and  $\mu \neq \lambda^*$ . By applying Jensen's inequality, we find

$$E\ln\sum_{k=1}^{K}R_{k}(s)\frac{\lambda_{k}^{*}}{\lambda_{k}^{*}\kappa+(1-\kappa)\mu_{k}} \geq \sum_{k=1}^{K}\lambda_{k}^{*}\ln\frac{\lambda_{k}^{*}}{\lambda_{k}^{*}\kappa+(1-\kappa)\mu_{k}}$$
(A.2)  
$$E\ln\sum_{k=1}^{K}R_{k}(s)\frac{\mu_{k}}{\lambda_{k}^{*}\kappa+(1-\kappa)\mu_{k}} \leq \ln E\sum_{k=1}^{K}R_{k}(s)\frac{\mu_{k}}{\lambda_{k}^{*}\kappa+(1-\kappa)\mu_{k}}$$
(A.3)

and so

$$\Upsilon(\kappa,\mu) \geq \sum_{k=1}^{K} a_k \ln \frac{a_k}{a_k \kappa + (1-\kappa)\mu_k} - \ln \sum_{k=1}^{K} a_k \frac{\mu_k}{a_k \kappa + (1-\kappa)\mu_k}$$
(A.4)

where  $a_k = \lambda_k^*$ .

Let  $\kappa = 0$ . Then the right-hand side of inequality (A.4) reduces to

$$\sum a_k \ln a_k - \sum a_k \ln \mu_k.$$

In view of Lemma 2.2, this difference is strictly positive, since  $(a_k) \neq (\mu_k)$ .

If  $\kappa \in (0, 1]$ , then we have a strict inequality in (A.3). To prove this, it is sufficient to show that the function

$$\phi(s) = \sum_{k=1}^{K} R_k(s) \mu_k [\lambda_k^* \kappa + (1-\kappa)\mu_k]^{-1}, \ s \in S,$$

is not a constant. Suppose  $\phi(s)$  is constant, i.e.,  $\phi(s) \equiv \gamma$ . Then

$$\sum_{k=1}^{K} R_k(s) \left( \mu_k \left[ \lambda_k^* \kappa + (1-\kappa) \, \mu_k \right]^{-1} - \gamma \right) = 0, \ s \in S,$$

which implies  $\mu_k = \gamma \left(\lambda_k^* \kappa + (1-\kappa) \mu_k\right)$ , since the functions  $R_k(\cdot)$ , k = 1, 2, ..., K, are linearly independent. We can see that  $\gamma = 1$ , and so  $\kappa \left(\lambda_k^* - \mu_k\right) = 0$ . Since  $\kappa > 0$ , this implies  $\lambda_k^* = \mu_k$ , k = 1, 2, ..., K, which, however, is ruled out by our assumptions.

It remains to show that the expression on the right-hand side of (A.4) is non-negative. It is equal to zero if  $\kappa = 1$ . If  $\kappa < 1$ , we can write it in the form

$$g(u) = \sum_{k=1}^{K} a_k \ln \frac{a_k}{a_k u + \mu_k} - \ln \sum_{k=1}^{K} a_k \frac{\mu_k}{a_k u + \mu_k}$$
(A.5)

where  $u = \kappa (1 - \kappa)^{-1}$ . We can see that  $g(u) \to 0$  as  $u \to \infty$ . Thus it remains to prove the inequality  $g'(u) \leq 0$  for all u > 0. We have

$$g'(u) = -\sum_{k=1}^{K} a_k^2 \left(a_k \, u + \mu_k\right)^{-1} + \frac{\sum_{k=1}^{K} a_k^2 \, \mu_k \left(a_k \, u + \mu_k\right)^{-2}}{\sum_{k=1}^{K} a_k \, \mu_k \left(a_k \, u + \mu_k\right)^{-1}}$$

The sign of g'(u) is the same as the sign of the expression

$$J := -\left[\sum_{k=1}^{K} a_k^2 \left(a_k \, u + \mu_k\right)^{-1}\right] \sum_{k=1}^{K} a_k \, \mu_k \left(a_k \, u + \mu_k\right)^{-1} + \sum_{k=1}^{K} a_k^2 \, \mu_k \left(a_k \, u + \mu_k\right)^{-2}.$$

By setting  $w_k = a_k u + \mu_k$ , we find  $\mu_k = w_k - a_k u$  and

$$J = -\left[\sum_{k=1}^{K} a_k^2 w_k^{-1}\right] \sum_{k=1}^{K} a_k (w_k - a_k u) w_k^{-1} + \sum_{k=1}^{K} a_k^2 (w_k - a_k u) w_k^{-2}$$
  
$$= -\left[\sum_{k=1}^{K} a_k^2 w_k^{-1}\right] \left[1 - \sum_{k=1}^{K} a_k^2 u w_k^{-1}\right] + \sum_{k=1}^{K} a_k^2 w_k^{-1} - \sum_{k=1}^{K} a_k^3 u w_k^{-2}$$
  
$$= u \left[\left(\sum_{k=1}^{K} a_k v_k\right)^2 - \sum_{k=1}^{K} a_k v_k^2\right],$$

where  $v_k = a_k w_k^{-1}$ . The last expression is non-positive in view of the Schwartz inequality. The lemma is proved.

### Proof of Lemma 3.1

**Lemma A.2 (Lemma 3.1)** For any  $\lambda \in \Delta^K$  distinct from  $\lambda^*$ , there exist numbers H > 0 and  $\delta > 0$  such that

$$E\min\{H, \ln F(\lambda, \kappa, s)\} \ge \delta$$

for all  $\kappa \in (0, 1]$ , where

$$F(\lambda,\kappa,s) := \frac{A}{B} = \frac{\sum_{k=1}^{K} R_k(s) \frac{\lambda_k^*}{\lambda_k^* \kappa + \lambda_k(1-\kappa)}}{\sum_{k=1}^{K} R_k(s) \frac{\lambda_k}{\lambda_k^* \kappa + \lambda_k(1-\kappa)}}.$$

*Proof.* Let us first assume that at least one of the coordinates of  $\lambda$  is zero, i.e.  $\mathbf{K} := \{k : \lambda_k = 0\} \neq \emptyset$ . Denote by S' the set of those s for which  $\sum_{k=1}^{K} R_k(s)\lambda_k = 0$ . If  $s \in S'$ , then  $R_k(s)\lambda_k = 0$  for all k. Thus,

$$B = \sum_{k=1}^{K} R_k(s) \frac{\lambda_k}{\lambda_k^* \kappa + \lambda_k (1-\kappa)} = 0, \ s \in S',$$

and so  $F(\lambda, \kappa; s) = +\infty$ ,  $s \in S'$ . Hence for any H > 1, we have

$$E\min[H, \ln F(\lambda, \kappa; s)] = E\min[H, \ln F(\lambda, \kappa; s)]\mathbf{1}_{S'} + E\min[H, \ln F(\lambda, \kappa; s)]\mathbf{1}_{S\setminus S'} \geq HP(S') + 2(\ln c)(1 - P(S')),$$
(A.6)

where  $\mathbf{1}_{\Gamma}$  is the indicator function of the set  $\Gamma$ .

Suppose that P(S') > 0. Then for all *H* large enough, the expression in (A.6) is greater than 1. The assertion of Lemma 3.1 will be true for any *H* for which (A.6) is greater than 1 and  $\delta = 1$ .

Now assume that P(S') = 0, i.e.  $\sum_{k=1}^{K} R_k(s)\lambda_k > 0$  (a.s.). In this case, we will proceed with the proof of the lemma in three steps.

Step 1: Let us show that

$$E \ln F(\lambda, \kappa; s) > 0 \text{ for all } \kappa \in (0, 1].$$
(A.7)

By applying Jensen's inequality, we obtain

$$E\ln\sum_{k=1}^{K} R_k(s) \frac{\lambda_k^*}{\lambda_k^* \kappa + \lambda_k(1-\kappa)} \ge E\sum_{k=1}^{K} R_k(s)\ln\frac{\lambda_k^*}{\lambda_k^* \kappa + \lambda_k(1-\kappa)}$$
(A.8)

$$=\sum_{k=1}^{K}\lambda_{k}^{*}\ln\frac{\lambda_{k}^{*}}{\lambda_{k}^{*}\kappa+\lambda_{k}(1-\kappa)}$$
(A.9)

and

$$E\ln\sum_{k=1}^{K}R_{k}(s)\frac{\lambda_{k}}{\lambda_{k}^{*}\kappa+\lambda_{k}(1-\kappa)} < \ln E\sum_{k=1}^{K}R_{k}(s)\frac{\lambda_{k}}{\lambda_{k}^{*}\kappa+\lambda_{k}(1-\kappa)}$$
(A.10)

$$= \ln \sum_{k=1}^{K} \lambda_k^* \frac{\lambda_k}{\lambda_k^* \kappa + \lambda_k (1-\kappa)}.$$
 (A.11)

The inequality in (A.10) is strict because its right-hand side is finite (all  $\lambda_k^*$  and some  $\lambda_k$  are strictly positive) and there is no constant  $\gamma$  such that

$$\sum_{k=1}^{K} R_k(s) \frac{\lambda_k}{\lambda_k^* \kappa + \lambda_k (1-\kappa)} = \gamma \text{ (a.s.)}.$$
 (A.12)

Indeed, if (A.12) holds, then

$$\sum_{k=1}^{K} R_k(s)\nu_k = 0 \text{ (a.s.)}, \tag{A.13}$$

where  $\nu_k := \lambda_k [\lambda_k^* \kappa + \lambda_k (1 - \kappa)]^{-1} - \gamma$ . Observe that at least one of the numbers  $\nu_k$  is not equal to zero. Otherwise,  $\lambda_k = [\lambda_k^* \kappa + \lambda_k (1 - \kappa)]\gamma$  for all k, and by summing up these equations over k, we obtain  $\gamma = 1$ , which yields  $\lambda_k = [\lambda_k^* \kappa + \lambda_k (1 - \kappa)]$ . Hence  $\lambda_k = \lambda_k^*$  for all k (recall that  $\kappa \neq 0$ ). This is a contradiction because  $\lambda \neq \lambda^*$ . From this we conclude that at least one of the numbers  $\nu_k$  is not equal to zero, which indicates that (A.13) cannot hold since the functions  $R_k(s)$  are linearly independent with respect to the given distribution on S.

By combining (A.8)-(A.11), we have

$$E\ln F(\lambda,\kappa;s) > \Phi_{\kappa}(\lambda), \tag{A.14}$$

where

$$\Phi_{\kappa}(\lambda) := \sum_{k=1}^{K} \lambda_k^* \ln \frac{\lambda_k^*}{\lambda_k^* \kappa + \lambda_k (1-\kappa)} - \ln \sum_{k=1}^{K} \lambda_k^* \frac{\lambda_k}{\lambda_k^* \kappa + \lambda_k (1-\kappa)}.$$

Following from Evstigneev et al. [58], Lemma 1 (the first lemma that we proved in this appendix), we have that  $\Phi_{\kappa}(\lambda) > 0$  for all  $\lambda = (\lambda_1, ..., \lambda_K) > 0$ .

Thus  $\Phi_{\kappa}(\lambda(1-\varepsilon)+\varepsilon\lambda^*) > 0$  for each  $\varepsilon > 0$ . The function  $\Phi_{\kappa}(\lambda)$  is finite and continuous on  $\Delta^{K}$ . Consequently,  $\Phi_{\kappa}(\lambda) = \lim_{\varepsilon \downarrow 0} \Phi_{\kappa}(\lambda(1-\varepsilon)+\varepsilon\lambda^*)) \ge 0$ . This inequality combined with (A.14) yields (A.7).

Step 2: Let us show that for all  $\kappa \in (0, 1]$ ,

$$F(\lambda, \kappa, s) \ge F(\lambda, 1, s) = \left[\sum_{k=1}^{K} R_k(s) \frac{\lambda_k}{\lambda_k^*}\right]^{-1}$$
(A.15)

 $(0^{-1} := +\infty)$ . Fix any  $s \in S$ . If  $R_k(s)\lambda_k = 0$  for all k, then  $F(\lambda, \kappa, s) = F(\lambda, 1, s) = +\infty$ , and so inequality (A.15) is valid. Let us suppose that  $R_k(s)\lambda_k \neq 0$  for some k. Then the denominator in (3.9) is strictly positive, and so  $c^2 \leq F(\lambda, \kappa, s) < \infty$ . For  $\kappa \in (0, 1)$ , we get  $F(\lambda, \kappa; s) = G(\lambda, u, s)$ , where

$$G(\lambda, u, s) := \frac{\sum_{k=1}^{K} R_k(s) \frac{\lambda_k^*}{\lambda_k^* u + \lambda_k}}{\sum_{k=1}^{K} R_k(s) \frac{\lambda_k}{\lambda_k^* u + \lambda_k}} = \frac{\sum_{k=1}^{K} R_k(s) \frac{\lambda_k^*}{\lambda_k^* + \lambda_k/u}}{\sum_{k=1}^{K} R_k(s) \frac{\lambda_k}{\lambda_k^* + \lambda_k/u}},$$

and  $u := \kappa(1-\kappa)^{-1}$ . When the variable  $\kappa$  ranges through (0,1), the variable u ranges through  $(0,+\infty)$ , and  $G(\lambda, u, s) \to [\sum_{k=1}^{K} R_k(s)\lambda_k/\lambda_k^*]^{-1}$  as  $u \to \infty$ . Hence, to verify (A.15), it suffices to show that the derivative  $G(\lambda, u; s)'$  of the function  $G(\lambda, u; s)$  with respect to u is non-positive for all u > 0. Assuming that the parameter s is fixed, we omit "s" in the notation and write

$$=\frac{(\sum R_k \frac{\lambda_k^*}{\lambda_k^* u + \lambda_k})'(\sum R_k \frac{\lambda_k}{\lambda_k^* u + \lambda_k}) - (\sum R_k \frac{\lambda_k^*}{\lambda_k^* u + \lambda_k})(\sum R_k \frac{\lambda_k}{\lambda_k^* u + \lambda_k})'}{(\sum R_k \frac{\lambda_k}{\lambda_k^* u + \lambda_k})^2},$$

where  $\sum = \sum_{k=1}^{K}$ . The sign of the above fraction is the same as the sign of its nominator

$$J := -\sum R_k \frac{(\lambda_k^*)^2}{(\lambda_k^* u + \lambda_k)^2} \sum R_k \frac{\lambda_k}{\lambda_k^* u + \lambda_k} + \sum R_k \frac{\lambda_k^*}{\lambda_k^* u + \lambda_k} \sum R_k \frac{\lambda_k \lambda_k^*}{(\lambda_k^* u + \lambda_k)^2}$$
By setting  $w_k := \lambda_k^* u + \lambda_k$  and  $v_k := \lambda_k^* w_k^{-1}$ , we obtain  $\lambda_k = w_k - \lambda_k^* u$  and

$$\begin{split} & J = -\sum R_k (\lambda_k^*)^2 w_k^{-2} \sum R_k \lambda_k w_k^{-1} + \sum R_k \lambda_k^* w_k^{-1} \sum R_k \lambda_k \lambda_k^* w_k^{-2} \\ &= -\sum R_k (\lambda_k^*)^2 w_k^{-2} \sum R_k (w_k - \lambda_k^* u) w_k^{-1} \\ &+ \sum R_k \lambda_k^* w_k^{-1} \sum R_k (w_k - \lambda_k^* u) \lambda_k^* w_k^{-2} \\ &= -\sum R_k v_k^2 \sum (R_k - R_k u v_k) + \sum R_k v_k \sum R_k (v_k - v_k^2 u) \\ &- \sum R_k v_k^2 (1 - \sum R_k u v_k) + \sum R_k v_k (\sum R_k v_k - \sum R_k v_k^2 u) \\ &= -\sum R_k v_k^2 + (\sum R_k v_k^2) \sum R_k v_k u + (\sum R_k v_k)^2 - (\sum R_k v_k) \sum R_k v_k^2 u \\ &= (\sum R_k v_k)^2 - \sum R_k v_k^2 \le 0. \end{split}$$

The last inequality is non-positive by virtue of the Schwartz inequality (we apply here the fact that  $R_k \ge 0$  and  $\sum R_k = 1$ ). This completes the proof of inequality (A.15).

Step 3. Let us demonstrate that there exists a natural number m, such that  $E\min\{m, \ln F(\lambda, 1, s)\} > 0$ . The series of random variables

$$\phi_m := \min\{m, \ln F(\lambda, 1, s)\}, m = 1, 2, ...,$$

is bounded below by  $2 \ln c$  (see (3.10)), is nondecreasing and tends to  $\ln F(\lambda, 1, s)$ for each s. Therefore, we get  $E\phi_m \to E \ln F(\lambda, 1, s) > 0$  (see (A.7)), and so  $E\phi_m > 0$  for some  $m = m_0$ . By setting  $H := m_0, \delta := E \min\{H, \ln F(\lambda, 1; s)\} > 0$  and applying (A.15), we find

$$E\min\{H, \ln F(\lambda, \kappa; s)\} \ge E\min\{H, \ln F(\lambda, 1; s)\} = \delta,$$

which proves Lemma 3.1 in the case when at least one of the coordinates of  $\lambda$  is zero.

Now assume that  $\lambda$  has no zero coordinates:  $\lambda_k > 0$  for each k. Then the function  $\ln F(\lambda, \kappa; s), \kappa \in [0, 1]$ , is uniformly bounded:

$$2\ln c \le \ln F(\lambda,\kappa;s) \le \ln(\min_k \lambda_k)^{-2},$$

and so  $E \ln F(\lambda, \kappa; s)$  is continuous in  $\kappa \in [0, 1]$ . It is sufficient to show that the infimum of  $E \ln F(\lambda, \kappa; s)$  with  $\kappa \in [0, 1]$  is strictly positive (then  $\delta$  can be defined as this infimum and H as  $2|\ln c| + 2|\ln \min \lambda_k|$ ). By virtue of the continuity of  $E \ln F(\lambda, \kappa; s)$ , this will be proved if we establish the inequality  $E \ln F(\lambda, \kappa; s) > 0$  for each  $\kappa \in [0, 1]$ . For  $\kappa \in (0, 1]$  this was proved above (see (A.7)) under the condition that  $\sum_{k=1}^{K} R_k(s)\lambda_k > 0$  (a.s.), which holds if  $\lambda > 0$ . In the case of  $\kappa = 0$ , we have

$$E\ln F(\lambda, 0, s) = E\ln\sum_{k=1}^{K} R_k(s) \frac{\lambda_k^*}{\lambda_k} \ge \ln\sum_{k=1}^{K} \lambda_k^* \frac{\lambda_k^*}{\lambda_k} \ge \sum_{k=1}^{K} \lambda_k^* \ln \frac{\lambda_k^*}{\lambda_k} > 0$$

as long as  $\lambda^*, \lambda > 0$  and  $\lambda \neq \lambda^*$ .

## Proof of Lemma 4.1

Let S be a finite set, and, for each  $s, \sigma \in S$ , let  $p(\sigma|s)$  be a probability distribution on S:

$$p(\sigma|s) \ge 0, \ \sum_{\sigma} p(\sigma|s) = 1.$$

For each  $\sigma \in S$ , let  $R(\sigma, s) = (R_1(\sigma, s), ..., R_k(\sigma, s))$  be a vector in unit simplex  $\Delta^K$  satisfying (A1) and (A2) for all  $s \in S$ .

Let  $\rho > 0$  be a number, such that  $R_k^*(s) > \rho$ ,  $s \in S$  (see (4.11)). Denote by  $\Delta_{\rho}^K$  the set of those vectors  $(b_1, ..., b_K)$  in  $\Delta^K$  that satisfy  $b_k \ge \rho$ , k = 1, ..., K. Consider the function

$$\Psi(s,\kappa,\mu) = \sum_{\sigma\in S} p(\sigma|s) \ln \sum_{k=1}^{K} R_k(\sigma,s) \frac{R_k^*(s)}{R_k^*(s)\kappa + (1-\kappa)\mu_k} - \sum_{\sigma\in S} p(\sigma|s) \ln \sum_{k=1}^{K} R_k(\sigma,s) \frac{\mu_k}{R_k^*(s)\kappa + (1-\kappa)\mu_k}$$

of  $s \in S$ ,  $\kappa \in [0, 1]$  and  $\mu = (\mu_k) \in \Delta_{\rho}^K$ .

**Lemma A.3 (Lemma 4.1)** There exists a constant  $L_{\rho}$  and a function  $\delta_{\rho}(\gamma) \geq 0$  of  $\gamma \in [0, \infty)$  satisfying the following conditions:

- 1. The function  $\delta(\cdot)$  is non-decreasing, and  $\delta_{\rho}(\gamma) > 0$  for all  $\gamma > 0$ .
- 2. For any  $s \in S$ ,  $\kappa \in [0,1]$  and  $\mu = (\mu_k) \in \Delta_{\rho}^K$ , we have

$$L_{\rho}|R^{*}(s) - \mu| \ge \Phi(s, \kappa, \mu) \ge \delta_{\rho}(|R^{*}(s) - \mu|).$$
 (A.16)

*Proof.* It follows from Evstigneev et al. [58], Lemma 1 (the first lemma that we proved in this appendix) that, for all  $s \in S$ ,  $\kappa \in [0, 1]$  and any  $\mu \in \Delta_+^K$ ,  $\mu \neq R^*(s)$ , the value of  $\Psi(s, \kappa, \mu)$  is strictly positive.

Fix some  $\gamma_0 > 0$  for which the set  $W(s, \gamma) = \{\mu \in \Delta_{\rho}^K : |R^*(s) - \mu| \ge \gamma\}$  is non-empty for all  $s \in S, \gamma \in [0, \gamma_0]$  and define

$$\delta_{\rho}(s,\gamma) = \inf\{\Psi(s,\kappa,\mu): \kappa \in [0,1], \ \mu \in W(s,\gamma)\}.$$

if  $\gamma \in [0, \gamma_0]$  and  $\delta_{\rho}(s, \gamma) = \delta_{\rho}(s, \gamma_0)$  if  $\gamma > \gamma_0$ . Since  $\Psi(s, \kappa, \mu)$  is continuous and strictly positive on the compact set  $[0, 1] \times W(s, \gamma)$  ( $\gamma > 0$ ), the function  $\delta_{\rho}(s, \gamma)$  takes on strictly positive values for  $\gamma > 0$ . Clearly this function is nondecreasing in  $\gamma$ . Fix some s, consider any  $\mu \in \Delta_{\rho}^{K}$  and define  $\gamma = |R^*(s) - \mu|$ . Then we have  $\mu \in W(s, \gamma)$ , and so

$$\Psi(s,\kappa,\mu) \ge \delta_{\rho}(s,\gamma) = \delta_{\rho}(s,|R^*(s)-\mu|).$$

From this we can see that the sought-for function  $\delta_{\rho}(\gamma)$  can be defined as

$$\delta_{\rho}(\gamma) = \min_{s \in S} \delta_{\rho}(s, \gamma).$$

We can write  $\Psi(s, \kappa, \mu) = \Psi(s, \kappa, \mu) - \Psi(s, \kappa, R_k^*(s))$  since the latter term is zero. The function  $\Psi(s, \kappa, \mu)$  is differentiable in  $\mu \in \Delta_+^K$  and its gradient  $\Psi'_{\mu}(s, \cdot, \cdot)$  is continuous, and hence bounded, on the compact set  $[0, 1] \times \Delta_{\rho}^K$ . This implies the existence of the Lipschitz constant  $L_{\rho}$  in (A.16).

## Appendix B

## Proofs of the Main Results in Section 5.2

To prove Theorems 5.1 and 5.2, we derive a random dynamical system describing the dynamics of the market shares  $r_t^i$  of investors i = 1, ..., N.

**Theorem B.1** The following equations hold:

$$r_{t+1}^{i} = \sum_{k=0}^{K} R_{t+1,k} \frac{\lambda_{t,k}^{i} r_{t}^{i}}{\langle \lambda_{t,k}, r_{t} \rangle}, \ t = 0, 1, 2, ..., \ i = 1, 2, ..., N.$$
(B.1)

Proof. Put

$$\bar{V}_{t,0} := \bar{w}_t \ \left[ = \sum_{i=1}^N x_{t,0}^i = \sum_{i=1}^N \lambda_{t,0}^i w_t^i \right], \ \bar{V}_{t,k} = V_{t,k}, \ k = 1, 2, ..., K.$$
(B.2)

Using the notation  $A_{t+1,0} = 1 + \beta_{t+1}$ , we write

$$W_{t+1} = \sum_{i=1}^{N} w_{t+1}^{i} = \sum_{i=1}^{N} \langle A_{t+1}, x_{t}^{i} \rangle = \sum_{i=1}^{N} \sum_{k=0}^{K} A_{t+1,k} x_{t,k}^{i}$$
$$= \sum_{k=0}^{K} \left( A_{t+1,k} \sum_{i=1}^{N} x_{t,k}^{i} \right) = (1 + \beta_{t+1}) \bar{w}_{t} + \sum_{k=1}^{K} A_{t+1,k} V_{t,k} = \sum_{k=0}^{K} A_{t+1,k} \bar{V}_{t,k}$$
(B.3)

Define  $\lambda_{t,k} := (\lambda_{t,k}^1, ..., \lambda_{t,k}^N)$  and  $w_t := (w_t^1, ..., w_t^N)$ . From Eqs. (5.7) and (5.10) we obtain

$$p_{t,k}V_{t,k} = \langle \lambda_{t,k}, w_t \rangle, \ k = 1, ..., K, \ t \ge 0,$$

which implies

$$x_{t,k}^{i} = \frac{\lambda_{t,k}^{i} w_{t}^{i}}{p_{t,k}} = \frac{\lambda_{t,k}^{i} w_{t}^{i} V_{t,k}}{\langle \lambda_{t,k}, w_{t} \rangle} = \frac{\lambda_{t,k}^{i} w_{t}^{i} \overline{V}_{t,k}}{\langle \lambda_{t,k}, w_{t} \rangle}, \ k = 1, \dots, K, \ t \ge 0$$
(B.4)

(see Eq. (5.11)). Further, we find

$$p_{t,0}\bar{V}_{t,0} = \bar{V}_{t,0} = \bar{w}_t = \langle \lambda_{t,0}, w_t \rangle,$$
 (B.5)

where the first equality holds because  $p_{t,0} = 1$ , the second follows from (B.2), and the third is a consequence of (5.1) and (5.12). Applying (5.12) and (B.5), we get

$$x_{t,0}^{i} = \frac{\lambda_{t,0}^{i} w_{t}^{i}}{p_{t,0}} = \frac{\lambda_{t,0}^{i} w_{t}^{i} \bar{V}_{t,0}}{\langle \lambda_{t,0}, w_{t} \rangle}.$$
 (B.6)

From (B.6) and (B.4), we obtain

$$w_{t+1}^{i} = \sum_{k=0}^{K} A_{t+1,k} x_{t,k}^{i} = A_{t+1,0} \bar{V}_{t,0} \frac{\lambda_{t,0}^{i} w_{t}^{i}}{\langle \lambda_{t,0}, w_{t} \rangle} + \sum_{k=1}^{K} A_{t+1,k} \bar{V}_{t,k} \frac{\lambda_{t,k}^{i} w_{t}^{i}}{\langle \lambda_{t,k}, w_{t} \rangle}$$
$$= \sum_{k=0}^{K} A_{t+1,k} \bar{V}_{t,k} \frac{\lambda_{t,k}^{i} w_{t}^{i}}{\langle \lambda_{t,k}, w_{t} \rangle}.$$

Dividing the first and the last terms in this chain of equalities by  $W_{t+1}$  and using (B.3), we have

$$r_{t+1}^{i} = \sum_{k=0}^{K} \frac{A_{t+1,k} \bar{V}_{t,k}}{\sum_{l=0}^{K} A_{t+1,l} \bar{V}_{t,l}} \frac{\lambda_{t,k}^{i} w_{t}^{i}}{\langle \lambda_{t,k}, w_{t} \rangle}.$$
 (B.7)

Observe that

$$\frac{\lambda_{t,k}^{i}w_{t}^{i}}{\langle\lambda_{t,k},w_{t}\rangle} = \frac{\lambda_{t,k}^{i}w_{t}^{i}/W_{t}}{\langle\lambda_{t,k},w_{t}/W_{t}\rangle} = \frac{\lambda_{t,k}^{i}r_{t}^{i}}{\langle\lambda_{t,k},r_{t}\rangle}.$$
(B.8)

Further, we obtain

$$\frac{A_{t+1,k}\bar{V}_{t,k}}{\sum_{l=0}^{K}A_{t+1,l}\bar{V}_{t,l}} = \frac{a_{t+1,k}\bar{w}_t\bar{V}_{t,k}}{\left(1+\beta_{t+1}\right)\bar{w}_t+\sum_{l=1}^{K}a_{t+1,l}\bar{w}_t\bar{V}_{t,l}} \\
= \frac{a_{t+1,k}V_{t,k}}{\sum_{l=0}^{K}a_{t+1,l}V_{t,l}} = R_{t+1,k} \tag{B.9}$$

(k = 1, ..., K) and

$$\frac{A_{t+1,0}\bar{V}_{t,0}}{\sum_{l=0}^{K}A_{t+1,l}\bar{V}_{t,l}} = \frac{(1+\beta_{t+1})\bar{w}_{t}}{(1+\beta_{t+1})\bar{w}_{t}+\sum_{l=1}^{K}a_{t+1,l}\bar{w}_{t}V_{t,l}}$$
$$= \frac{a_{t+1,0}V_{t,0}}{\sum_{l=0}^{K}a_{t+1,l}V_{t,l}} = R_{t+1,0}$$
(B.10)

(see (5.16) and (5.17)). By substituting Eqs. (B.8)-(B.10) into (B.7), we arrive at (B.1). The proof is complete.  $\blacksquare$ 

We have shown that the dynamics of  $r_t^i$  of the investors i = 1, 2, ..., N, in this model is governed by the random dynamical system (B.1). A detailed analysis

of this system was carried out in Section 2.3. As we mentioned in Section 5.2, Theorems 5.1 and 5.2 are direct consequences of Theorems B.2 and B.3 (correspondingly, Theorem 2.1 and Theorem 2.2 proved in Section 2.3), which we formulate below.

Let  $R_{t,k}(s^t) \geq 0$  (t = 0, 1, 2, ...; k = 0, ..., K) be measurable real-valued functions satisfying (5.18) and (5.20). Consider sequences of measurable vector functions  $\Lambda = (\lambda_t(s^t))_{t=0}^{\infty}$ , where  $\lambda_t(s^t) = (\lambda_{t,0}(s^t), ..., \lambda_{t,K}(s^t)) \in \Delta^{K+1}$ . Denote by  $\mathcal{L}$  the set of N-tuples  $(\Lambda^1, ..., \Lambda^N)$  of such sequences for which the random dynamical system (B.1) generates well-defined vectors  $r_t^i = (r_{t,0}^i, ..., r_{t,K}^i) \in \Delta^{K+1}$ , i = 1, ..., N, t = 0, 1, ..., i.e., the validity of the inequality

$$\sum_{i=1}^{N} \lambda_{t,k}^{i} r_{t}^{i} > 0, \ k = 0, 1, ..., K,$$

is guaranteed for each  $t \ge 0$ . Define  $\Lambda^* = (\lambda_t^*(s^t))_{t=0}^{\infty}$  by (5.19).

**Theorem B.2** For each  $(\Lambda^1, ..., \Lambda^N) \in \mathcal{L}$  with  $\Lambda^1 = \Lambda^*$ , we have  $\inf_{t \ge 0} r_t^1 > 0$  (a.s.).

**Theorem B.3** Let  $\Lambda = (\lambda_t(s^t))_{t=0}^{\infty}$  be a sequence of measurable vector functions with values in  $\Delta^{K+1}$ . If  $\inf_{t\geq 0} r_t^1 > 0$  for any  $(\Lambda^1, ..., \Lambda^N) \in \mathcal{L}$  with  $\Lambda^1 = \Lambda$ , then (5.21) holds.

For proofs of Theorems B.2 and B.3, see Section 2.3.

## Bibliography

- Amir, R., Belkov, S., Evstigneev, I. V., and Hens, T. (2022). An evolutionary finance model with short selling and endogenous asset supply. *Economic Theory* 73(2), 655-677.
- [2] Amir, R., Evstigneev, I.V., Hens, T., and Schenk-Hoppé, K.R. (2005). Market selection and survival of investment strategies. *Journal of Mathematical Economics* 41 (Special Issue on Evolutionary Finance), 105–122.
- [3] Amir, R., Evstigneev, I.V., Hens, T., and Xu, L. (2011). Evolutionary finance and dynamic games. *Mathematics and Financial Economics* 5, 161–184.
- [4] Amir, R., Evstigneev, I.V., and Schenk-Hoppé, K.R. (2013). Asset market games of survival: a synthesis of evolutionary and dynamic games. *Annals* of Finance 9(2), 121–144.
- [5] Alchian, A. (1950). Uncertainty, evolution and economic theory. Journal of Political Economy 58, 211–221.
- [6] Algoet, P.H., and Cover, T.M. (1988). Asymptotic optimality and asymptotic equipartition properties of log-optimum investment. Annals of Probability 16(2), 876-898.
- [7] Amir, R., Sahi, S., Shubik, M., and Yao, S. (1990). A strategic market game with complete markets. *Journal of Economic Theory* 51, 126–143.
- [8] Arkin, V.I., and Evstigneev I.V. (1987). Stochastic Models of Control and Economic Dynamics. London: Academic Press.
- [9] Arnott, R.D., Hsu, J.C., and West, J.M. (2008). The Fundamental Index: A Better Way to Invest. Wiley.
- [10] Aurell, E., Baviera, R., Hammarlid, O., Serva, M., and Vulpiani, A. (2000). Growth optimal investment and pricing of derivatives. *Physica A: Statistical Mechanics and its Applications* 280(3-4), 505-521.
- [11] Bachmann, K.K., De Giorgi, E.G., and Hens, T. (2018). Behavioral Finance for Private Banking: From the Art of Advice to the Science of Advice, 2nd Edition, Wiley Finance.

- [12] Bahsoun, W., Evstigneev, I.V., and Xu, L. (2011). Almost sure Nash equilibrium strategies in evolutionary models of asset markets. *Mathematical Methods of Operations Research* 73, 235–250.
- [13] Belkov, S., Evstigneev, I. V., and Hens, T. (2020). An evolutionary finance model with a risk-free asset. Annals of Finance 16(4), 593-607.
- [14] Belkov, S., Evstigneev, I. V., Hens, T., and Xu, L. (2020). Nash equilibrium strategies and survival portfolio rules in evolutionary models of asset markets. *Mathematics and Financial Economics* 14(2), 249-262.
- [15] Bell, R.M., and Cover, T.M. (1980). Competitive optimality of logarithmic investment. *Mathematics of Operations Research* 5(2), 161–166.
- [16] Bell, R.M., and Cover, T.M. (1988). Game-theoretic optimal portfolios. Management Science 34(6), 724–733.
- [17] Benninga, S., Björk, T., and Wiener, Z. (2002). On the use of numeraires in option pricing. *The Journal of Derivatives* 10, 43–58.
- [18] Bertsekas, D.P., and Shreve, S.E. (1978). Stochastic Optimal Control: The Discrete Time Case. New York: Academic Press.
- [19] Birge, J.R., and Louveaux, F. (2011). Introduction to Stochastic Programming. New York: Springer.
- [20] Björk, T. (1999). Arbitrage Theory in Continuous Time. Oxford: Oxford University Press.
- [21] Blume, L., and Easley, D. (1992). Evolution and market behavior, *Journal of Economic Theory* 58, 9–40.
- [22] Blume, L., and Easley, D. (2006). If you are so smart, why aren't you rich? Belief selection in complete and incomplete markets. *Econometrica* 74, 929-966.
- [23] Blume, L., and Easley, D. (2009). Market selection and asset pricing. Handbook of Financial Markets: Dynamics and Evolution (Hens, T., Schenk-Hoppé, K.R. eds.), Chapter 7, Amsterdam: North-Holland, 403-437.
- [24] Borch, K. (1966). A utility function derived from a survival game. Management Science 12, 287–295.
- [25] Borel, E. (1921). La théorie du jeu et les équations intégrales à noyau symétrique. CR Hebd Acad Sci 173, 1304-1308. English translation: Borel, E. (1953). The theory of play and integral equations with skew symmetric kernels. Econometrica 21, 97-100.
- [26] Borovkov, A.A. (1998). Mathematical Statistics. Amsterdam: Gordon and Breach.

- [27] Bottazzi, G., and Dindo, P. (2013). Evolution and market behavior in economics and finance: Introduction to the special issue. *Journal of Evolutionary Economics* 23, 507–512.
- [28] Bottazzi, G., and Dindo, P. (2013). Selection in asset markets: the good, the bad, and the unknown. *Journal of Evolutionary Economics* 23, 641– 661.
- [29] Bottazzi, G., Dindo, P., and Giachini, D. (2018). Long-run heterogeneity in an exchange economy with fixed-mix traders. *Economic Theory* 66, 407–447.
- [30] Bottazzi, G., Dindo, P., and Giachini, D. (2019). Momentum and reversal in financial markets with persistent heterogeneity. *Annals of Finance* 15, 455-487.
- [31] Bottazzi, G., Dosi, G., and Rebesco, I. (2005). Institutional architectures and behavioral ecologies in the dynamics of financial markets. *Journal of Mathematical Economics* 41, 197–228.
- [32] Boulding, K.E. (1981). Evolutionary Economics. London: Sage.
- [33] Bouton, C.L. (1901-2). Nim, a game with a complete mathematical theory. Ann Math 3, 35-39.
- [34] Breiman, L. (1961). Optimal gambling systems for favorable games. Proceedings of the Fourth Berkeley Symposium on Mathematical Statistics and Probability 1, 65–78.
- [35] Brock, W.A., and Hommes, C.H. (1998). Heterogeneous beliefs and routes to chaos in a simple asset pricing model. *Journal of Economic Dynamics* and Control 22, 1235–1274.
- [36] Brock, A.W., Hommes, C.H., and Wagener, F.O.O. (2005). Evolutionary dynamics in markets with many trader types. *Journal of Mathematical Economics* 41 (Special Issue on Evolutionary Finance), 7–42.
- [37] Browne, S., and Whitt, W. (1996). Portfolio choice and the Bayesian Kelly criterion. Advances in Applied Probability 28(4), 1145-1176.
- [38] Bucher, M., and Woehrmann, P. (2006). Survival of noise traders and the implications for asset prices. Working Paper, 31 March 2006, Institute for Empirical Research in Economics, University of Zurich.
- [39] Cabrales, A. (2000). Stochastic replicator dynamics. International Economic Review 41, 451–481.
- [40] Coury, T., and Sciubba, E. (2012). Belief heterogeneity and survival in incomplete markets. *Economic Theory* 49, 37–58.
- [41] Cover, T.M. (1991). Universal portfolios. *Mathematical finance* 1(1), 1-29.

- [42] Cover, T.M. (1998). Shannon and investment. IEEE Information Theory Society Newsletter, Summer, Special Golden Jubilee Issue, 10–11.
- [43] Cover, T.M., and Thomas, J.A. (2006). Elements of Information Theory, 2nd Edition, Wiley.
- [44] De Giorgi, E., and Hens, T. (2006). Making prospect theory fit for finance. Financial Markets and Portfolio Management 20, 339–360.
- [45] De Giorgi, E., Hens, T., and Rieger, M.O. (2010). Financial market equilibria with cumulative prospect theory. *Journal of Mathematical Economics* 46 (Mathematical Economics: Special Issue in honour of Andreu Mas-Colell, Part 1), 633–651.
- [46] De Giorgi, E., Levy, H. and Hens, T. (2012). Two paradigms and Nobel Prizes in economics: A contradiction or coexistence? *European Financial Management* 18(2), 163–182.
- [47] Dempster, M.A.H., Evstigneev, I.V., and Schenk-Hoppé, K.R. (2007). Volatility-induced financial growth. *Quantitative Finance* 7, 151–160.
- [48] Denneberg, D. (1994). Non-additive measure and Integral. Kluwer Academic Publishers.
- [49] Dorfman, R., Samuelson, P.A., and Solow, R.M. (1958). Linear Programming and Economic Analysis. New York: McGraw-Hill.
- [50] Dynkin, E.B. (1969). Game variant of a problem on optimal stopping. Soviet Mathematics Doklady 10, 270–274.
- [51] Dynkin, E.B. (1972). Stochastic concave dynamic programming. Math USSR Sbor 16, 501-515.
- [52] Dynkin, E.B., and Yushkevich, A.A. (1979). Controlled Markov Processes and Their Applications. New York: Springer.
- [53] Dubins, L., and Savage, L.M. (1965). How to Gamble if You Must. New York: McGraw-Hill.
- [54] Enke, S. (1951). On maximizing profits: a distinction between Chamberlin and Robinson. *The American Economic Review* 41, 566–578.
- [55] Epstein, L.G., and Zin, S.E. (1989). Substitution, risk aversion, and the temporal behavior of consumption and asset returns: a theoretical framework. *Econometrica* 57, 937–969.
- [56] Epstein, L.G., and Zin, S.E. (1991). Substitution, risk aversion, and the temporal behavior of consumption and asset returns: an empirical analysis. *Journal of Political Economy* 99, 263–286.

- [57] Erickson, R., and Pakes, A. (1995). Markov-perfect industry dynamics: A framework for empirical work. *The Review of Economic Studies* 62, 53-82.
- [58] Evstigneev, I.V., Hens, T., and Schenk-Hoppé, K.R. (2002). Market selection of financial trading strategies: Global stability. *Mathematical Finance* 12, 329–339.
- [59] Evstigneev, I.V., Hens, T., and Schenk-Hoppé, K.R. (2006). Evolutionary stable stock markets. *Economic Theory* 27(2), 449–468.
- [60] Evstigneev, I.V., Hens, T., and Schenk-Hoppé, K.R. (2008). Globally evolutionarily stable portfolio rules. *Journal of Economic Theory* 140, 197-228.
- [61] Evstigneev, I.V., Hens, T., and Schenk-Hoppé, K.R. (2009). Evolutionary finance. Handbook of Financial Markets: Dynamics and Evolution (Hens T., and Schenk-Hoppé, K.R., eds.), Chapter 9, Amsterdam: Elsevier, 507-566.
- [62] Evstigneev, I.V., Hens, T., and Schenk-Hoppé, K.R. (2011). Local stability analysis of a stochastic evolutionary financial market model with a riskfree asset. *Mathematics and Financial Economics* 5, 185-202.
- [63] Evstigneev, I.V., Hens, T., and Schenk-Hoppé, K.R. (2015). Mathematical Financial Economics: A Basic Introduction. Springer.
- [64] Evstigneev, I.V., Hens, T., and Schenk-Hoppé, K.R. (2016). Evolutionary behavioural finance. *Handbook of Post Crisis Financial Modelling* (E. Haven et al., eds.), Palgrave MacMillan, 214–234.
- [65] Farmer, J.D. (2002). Market force, ecology and evolution. Industrial and Corporate Change 11, 895–953.
- [66] Farmer, J.D., and Lo, A.W. (1999). Frontiers of finance: Evolution and efficient markets, *Proceedings of the National Academy of Sciences of the* USA 96, 9991-9992.
- [67] Flåm, S.D. (2010). Portfolio management without probabilities or statistics. Annals of Finance 6, 357–368.
- [68] Flåm, S.D. (2019). Emergence of price-taking behavior. Economic Theory, online publication: https://doi.org/10.1007/s00199-019-01232-5.
- [69] Föllmer, H., and Schied, A. (2002). *Stochastic Finance: An Introduction in Discrete Time*. Walter de Gruyter.
- [70] Foster, D., and Young P. (1990). Stochastic evolutionary game dynamics. *Theoretical Population Biology* 38, 219–232.
- [71] Fudenberg, D., and Harris, C. (1992). Evolutionary dynamics with aggregate shocks. *Journal of Economic Theory* 57, 420–441.

- [72] Gale, D. (1967). On optimal development in a multi-sector economy. *Review of Economic Studies* 34, 1–18.
- [73] Gale, D., and Stewart, F.M. (1953). Infinite games with perfect information. Contributions to the Theory of Games II, Annals of Mathematical Studies (Kuhn, H.W., and Tucker, A.W., eds.), Princeton: Princeton University Press, 28, 245-266.
- [74] Geman, H., El Karoui, N., and Rochet, J.C. (1995). Change of Numeraire, Changes of Probability Measures and Pricing of Options. *Journal of Applied Probability* 32, 443-458.
- [75] Germano, F. (2007). Stochastic evolution of rules for playing finite normal form games. *Theory and Decision* 62(4), 311–333.
- [76] Gintis, H. (2009). Game Theory Evolving: A Problem-Centered Introduction to Modeling Strategic Interaction, Second Edition, Princeton University Press.
- [77] Giraud, G., and Stahn, H. (2008). On Shapley-Shubik equilibria with financial markets. *Economic Theory* 35, 469-496.
- [78] Grandmont, J.-M. (1977). Temporary general equilibrium theory. *Econo*metrica 45, 535-572.
- [79] Grandmont, J.-M., ed. (1988). Temporary Equilibrium. San Diego: Academic Press.
- [80] Grandmont, J.-M., and Hildenbrand, W. (1974). Stochastic processes of temporary equilibria. *Journal of Mathematical Economics* 1, 247–277.
- [81] Hakansson, N.H. (1975). Optimal investment and consumption strategies under risk for a class of utility functions. *Econometrica* 38, 587-607.
- [82] Hakansson, N.H., and Ziemba, W.T. (1995). Capital growth theory. Handbooks in Operations Research and Management Science (Jarrow R.A. et al., eds.), Amsterdam: Elsevier, 65–86.
- [83] Hall, P., and Heyde, C.C. (1980). Martingale limit theory and its application. London: Academic Press.
- [84] Hamilton, W.D. (1967). Extraordinary sex ratios. Science 156, 477-488.
- [85] Harrison, J., and Kreps, J. (1981). Martingales and arbitrage in multiperiod securities markets. *Journal of Economic Theory* 11, 418-443.
- [86] Haurie, A., Zaccour, G., and Smeers, Y. (1990). Stochastic equilibrium programming for dynamic oligopolistic markets. *Journal of Optimization Theory and Applications* 66, 243–253.

- [87] Hens, T., and Schenk-Hoppé, K.R. (2005). Evolutionary stability of portfolio rules in incomplete markets. *Journal of Mathematical Economics* 41, 43–66.
- [88] Hicks, J.R. (1946). Value and Capital, 2nd Edition. Oxford: Clarendon Press.
- [89] Hildenbrand, W. (1974). Core and Equilibria of a Large Economy. Princeton, NJ: Princeton University Press.
- [90] Hodgeson, G.M. (1993). Economics and Evolution: Bringing Life Back into Economics. Ann Arbor: University of Michigan Press.
- [91] Hofbauer, J., and Sigmund, K. (1998). Evolutionary Games and Population Dynamics. Cambridge, UK: Cambridge University Press.
- [92] Horváth, M., and Urbán, A. (2012). Growth optimal portfolio selection with short selling and leverage. *Machine Learning for Financial Engineering* (Györfi, L., Ottucsák, G., and Walk, H., eds.), London: Imperial College Press, 151-176.
- [93] Karatzas, I., and Shreve, S. (1998). Methods of Mathematical Finance. New York: Springer-Verlag.
- [94] Karni, E., and Schmeidler, D. (1986). Self-preservation as a foundation of rational behavior under risk. *Journal of Economic Behavior and Organi*zation 7, 71–81.
- [95] Kelly, J.L. (1956). A new interpretation of information rate. Bell System Technical Journal 35(4), 917–926.
- [96] Kifer, Y. (2000). Game options. Finance and Stochastics 4, 443–463.
- [97] Kojima, F. (2006). Stability and instability of the unbeatable strategy in dynamic processes. *International Journal of Economic Theory* 2, 41–53.
- [98] Kuhn, D., and Luenberger, D.G. (2010). Analysis of the rebalancing frequency in log-optimal portfolio selection. *Quantitative Finance* 10, 221-234.
- [99] Latané, H. (1959). Criteria for choice among risky ventures. Journal of Political Economy 67, 144–155.
- [100] Lindahl, E. (1939). Theory of Money and Capital. London: Allen and Unwin.
- [101] Lintner, J. (1965). The valuation of risk assets and the selection of risky investments in stock portfolios and capital budgets. *Review of Economics* and Statistics 47(1), 13–37.

- [102] Lo, A.W. (2004). The Adaptive Markets Hypothesis: Market efficiency from an evolutionary perspective. The Journal of Portfolio Management 30, 15–29.
- [103] Lo, A.W. (2005). Reconciling Efficient Markets with Behavioral Finance: The Adaptive Market Hypothesis. *Journal of Investment Consulting* 7, 21–44.
- [104] Lo, A.W. (2012). Adaptive Markets and the New World Order. Financial Analyst Journal 68, 18–29.
- [105] Lo, A.W. (2017). Adaptive Markets: Financial Evolution at the Speed of Thought. Princeton University Press.
- [106] Lo, A.W., Orr, H.A., and Zhang, R. (2018). The growth of relative wealth and the Kelly criterion. *Journal of Bioeconomics* 20, 49–67.
- [107] Luce, R.D., and Raiffa, H. (1989). Games and Decisions, 2nd Edition. New York: Dover.
- [108] Luenberger, D. (1998). Investment Science. New York: Oxford University Press.
- [109] MacLean, L.C., Thorp, E.O., and Ziemba, W.T., eds. (2010). The Kelly Capital Growth Investment Criterion: Theory and Practice. Singapore: World Scientific.
- [110] MacLean, L.C., and Ziemba, W.T., eds. (2013). Handbook of the Fundamentals of Financial Decision Making. Singapore: World Scientific.
- [111] Magill, M., and Quinzii, M. (1996). Theory of Incomplete Markets. Cambridge MA: MIT Press.
- [112] Magill, M., and Quinzii, M. (2003). Incentives and the stock market in general equilibrium. *General Equilibrium: Problems, Prospects and Alternatives* (Hahn, F., and Petri, F. eds.), New York: Routledge, 283–313.
- [113] Maitra, A.P., and Sudderth, W.D. (1996). Discrete Gambling and Stochastic Games. New York: Springer.
- [114] Markowitz, H.M. (1952). Portfolio Selection. Journal of Finance 7(1), 77– 91.
- [115] Marshall, A. (1949). Principles of Economics, 8th Edition. London: Macmillan.
- [116] Martin, D.A. (1975). Borel determinacy. Ann Math 102, 363-371.
- [117] Maynard Smith, J. (1982). Evolution and the Theory of Games. Cambridge: Cambridge University Press.

- [118] Maynard Smith, J., and Price, G. (1973). The logic of animal conflicts. *Nature* 246, 15-18.
- [119] McKenzie, L.W. (1986). Optimal economic growth, turnpike theorems and comparative dynamics. *Handbook of Mathematical Economics III* (Arrow, K.J., and Intrilligator, M.D., eds.), Amsterdam: North-Holland, 1281– 1355.
- Milnor, J., and Shapley, L.S. (1957). On games of survival. Contributions to the Theory of Games III, Annals of Mathematical Studies 39 (Dresher, M., et al., eds.). Princeton, NJ: Princeton University Press, 15–45.
- [121] Mossin, J. (1966). Equilibrium in a capital asset market. Econometrica 34(4), 768–783.
- [122] Myerson, R.B. (1991). Game Theory: Analysis of Conflict. Cambridge, MA: Harvard University Press.
- [123] Nelson, R.R., and Winter, S.G. (1982). An Evolutionary Theory of Economic Change. Cambridge, MA: Harvard University Press.
- [124] Neveu, J. (1965). Mathematical Foundations of the Calculus of Probability Theory. San Francisco: Holden Day.
- [125] Neyman, A., and Sorin, S., eds. (2003). Stochastic Games and Applications, NATO ASI series. Dordrecht: Kluwer Academic Publishers.
- [126] Nikaido, H. (1968). Convex Structures and Economic Theory. New York: Academic Press.
- [127] Penrose, E.T. (1952). Biological analogies in the theory of the firm. The American Economic Review 42, 804–819.
- [128] Pliska, S.R. (1997). Introduction to Mathematical Finance: Discrete Time Models. Blackwell Publ.
- [129] Rockafellar, R.T., and Wets, R. (1976). Stochastic convex programming: Basic duality. *Pacific Journal of Mathematics* 62, 173–195.
- [130] Ross, S. (1999). An Introduction to Mathematical Finance. Cambridge: Cambridge University Press.
- [131] Saks, S. (1964). The Theory of Integral. New York: Dover.
- [132] Samuelson, L. (1997). Evolutionary Games and Equilibrium Selection. Cambridge, MA: MIT Press.
- [133] Samuelson, P.A. (1947). Foundations of Economic Analysis. Cambridge, MA: Harvard University Press.
- [134] Sandholm, W.H. (2010). Population Games and Evolutionary Dynamics. Cambridge, MA: MIT Press.

- [135] Sandroni, A. (2000). Do markets favor agents able to make accurate predictions? *Econometrica* 68, 1303-1341.
- [136] Schaffer, M. (1988). Evolutionarily stable strategies for a finite population and a variable contest size. *Journal of Theoretical Biology* 132, 469-478.
- [137] Schaffer, M. (1989). Are profit-maximizers the best survivors? Journal of Economic Behavior and Organization 12, 29-45.
- [138] Schenk-Hoppé, K.R., and Sokko, A. (2017). Margin Requirements and Evolutionary Asset Pricing. Swiss Finance Institute Research Paper 17-20.
- [139] Schumpeter, J.A. (1911). Theorie der wirtschaftlichen Entwicklung. Leipzig: Duncker & Humblot.
- [140] Schwalbe, U., and Walker, P. (2001). Zermelo and the early history of game theory. *Games and Economic Behavior* 34, 123–137.
- [141] Sciubba, E. (2005). Asymmetric information and survival in financial markets. *Economic Theory* 25, 353–379.
- [142] Sciubba, E. (2006). The evolution of portfolio rules and the capital asset pricing model. *Economic Theory* 29, 123–150.
- [143] Secchi, P., and Sudderth, W.D. (2001). Stay-in-a-set games. International Journal of Game Theory 30, 479-490.
- [144] Shannon, C. (1956). Lectures on investment problems. MIT (unpublished).<sup>1</sup>
- [145] Shapley, L.S. (1953). Stochastic games. Proceedings of the National Academy of Sciences of the USA 39, 1095–1100.
- [146] Shapley, L.S., and Shubik, M. (1977). Trade using one commodity as a means of payment. *Journal of Political Economy* 85, 937–968.
- [147] Sharpe, W.F. (1964). Capital asset prices: A theory of market equilibrium under conditions of risk. *Journal of Finance* 19(3), 425–442.
- [148] Shiller, R.J. (2003). From efficient markets theory to behavioral finance. Journal of Economic Perspectives 17(1), 83–104.
- [149] Shleifer, A. (2000). Inefficient Markets: An Introduction to Behavioral Finance. Oxford, UK: Oxford University Press.
- [150] Shubik, M., and Thompson, G. (1959). Games of economic survival. Naval Research Logistics Quarterly 6, 111–123.

<sup>&</sup>lt;sup>1</sup>See Cover [42].

- [151] Shubik, M., and Whitt, W. (1973). Fiat money in an economy with one durable good and no credit. *Topics in Differential Games* (Blaquiere, A., ed.), Amsterdam: North-Holland, 401–448.
- [152] Tarnaud, R. (2019). Convergence within binary market scoring rules. Economic Theory 68, 1017–1050.
- [153] Telgársky, R. (1987). Topological games: On the 50th anniversary of the Banach-Mazur game. Rocky Mountain Journal of Mathematics 17, 227-276.
- [154] Thaler, R.H., ed. (2005). Advances in Behavioral Finance II. Princeton, NJ: Princeton University Press.
- [155] Thorp, E.O. (1971). Portfolio choice and the Kelly criterion. Stochastic optimization models in finance (Ziemba, W.T., and Vickson, R.G., eds.), New York: Academic Press, 599–619.
- [156] Thorp, E.O. (2006). The Kelly criterion in Blackjack sports betting and the stock market. *Handbook of Asset and Liability Management* (Zenios, S.A., Ziemba, W.T., eds.), Amsterdam: Elsevier, Volume 1, 387-428.
- [157] Tobin, J. (1958). Liquidity preference as behavior towards risk. Review of Economic Studies 25(2), 65–86.
- [158] Tversky, A., and Kahneman, D. (1991). Loss aversion in riskless choice: A reference-dependent model. *Quarterly Journal of Economics* 106, 1039– 1061.
- [159] Vega-Redondo, F. (1996). Evolution, Games, and Economic Behavior. Oxford, UK: Oxford University Press.
- [160] Vieille, N. (2000). Two-player stochastic games I: A reduction. Israel Journal of Mathematics 119(1), 55–91.
- [161] Vieille, N. (2000). Two-player stochastic games II: The case of recursive games. Israel Journal of Mathematics 119(1), 93–126.
- [162] Vieille, N. (2000). Small perturbations and stochastic games. Israel Journal of Mathematics 119(1), 127–142.
- [163] von Neumann, J. (1928). Zur Theorie der Gesellschaftsspiele. Math Ann 100, 295-320.
- [164] Wallace, S.W., Ziemba, W.T., eds. (2005). Applications of Stochastic Programming. SIAM - Mathematical Programming Society Series on Optimization, Philadelphia.
- [165] Weibull, J. (1995). Evolutionary Game Theory. Cambridge, MA: MIT Press.

- [166] Zermelo, E. (1913). Über eine Anwendung der Mengenlehre auf die Theorie des Schachspiels. Proceedings of the Fifth International Congress of Mathematicians (Cambridge 1912) (Hobson, E.W., and Love, A.E.H., eds.), Cambridge MA: Cambridge University Press, Volume 2, 501-504.
- [167] Zhang, R., Brennan, T.J., and Lo, A.W. (2014). Group Selection as Behavioral Adaptation to Systematic Risk. PLOS ONE 9, 1–9.
- [168] Zhou, X.Y. (2010). Mathematicalising behavioural finance. Proceedings of the International Congress of Mathematicians. Hyderabad, India.
- [169] Ziemba, W.T., and Vickson, R.G. (2011). Models of optimal capital accumulation and portfolio selection and the capital growth criterion. *The Kelly Capital Growth Investment Criterion: Theory and Practice* (MacLean, L.C., et al., eds), Singapore: World Scientific, 473–485.
- [170] Zierhut, M. (2020). Indeterminacy of Cournot–Walras equilibrium with incomplete markets. *Economic Theory*. Online publication: https://doi.org/10.1007/s00199-020-01248-2.